

https://lib.uliege.be



https://matheo.uliege.be

# Are francophone learners of English as a Foreign Language provided with the necessary tools to ensure viable pronunciation?

Auteur : Peters, Tom Promoteur(s) : Simons, Germain Faculté : Faculté de Philosophie et Lettres Diplôme : Master en langues et lettres modernes, orientation germaniques, à finalité didactique Année académique : 2022-2023 URI/URL : http://hdl.handle.net/2268.2/18955

Avertissement à l'attention des usagers :

Tous les documents placés en accès ouvert sur le site le site MatheO sont protégés par le droit d'auteur. Conformément aux principes énoncés par la "Budapest Open Access Initiative" (BOAI, 2002), l'utilisateur du site peut lire, télécharger, copier, transmettre, imprimer, chercher ou faire un lien vers le texte intégral de ces documents, les disséquer pour les indexer, s'en servir de données pour un logiciel, ou s'en servir à toute autre fin légale (ou prévue par la réglementation relative au droit d'auteur). Toute utilisation du document à des fins commerciales est strictement interdite.

Par ailleurs, l'utilisateur s'engage à respecter les droits moraux de l'auteur, principalement le droit à l'intégrité de l'oeuvre et le droit de paternité et ce dans toute utilisation que l'utilisateur entreprend. Ainsi, à titre d'exemple, lorsqu'il reproduira un document par extrait ou dans son intégralité, l'utilisateur citera de manière complète les sources telles que mentionnées ci-dessus. Toute utilisation non explicitement autorisée ci-avant (telle que par exemple, la modification du document ou son résumé) nécessite l'autorisation préalable et expresse des auteurs ou de leurs ayants droit.



Faculté de Philosophie et Lettres Département de Langues Modernes : Linguistique, littérature et traduction

# Are francophone learners of English as a Foreign Language provided with the necessary tools to ensure viable pronunciation?

Promoteur : Germain SIMONS

Co-promoteur : Laurent RASIER

Mémoire présenté par Tom PETERS en vue de l'obtention du grade de Master en langues et lettres modernes, orientation germaniques, à finalité didactique

Année académique 2022-2023

ASR	Automatic Speech Recognition				
CAPT	Computer Assisted Pronunciation Teaching				
EFL	English as a Foreign Language				
EIL	English as an International Language				
ELF	English as a Linga Franca				
ESL	English as a Second Language				
GA	General American				
$H_{13}$	Hypothesis number thirteen				
HVPT	High Variability Phonetic Training				
IPA	International Phonetic Alphabet				
L1	First Language				
L2	Second Language				
LFC	Lingua Franca Core				
MCQ	Multiple-Choice Question				
NNS	Non-Native Speaker				
NS	Native Speaker				
PCA	Phonetic Coding Ability				
$\mathbf{Q}_1$	Research question number one				
Q1	Question number one				
RP	Received Pronunciation				
SGAV	Structuro Global Audio Visual				
VTM	Verbo-Tonal Method				

#### Acknowledgments

This master's thesis could not have been possible without the help and assistance of many people. I would like to thank them.

I would like to thank both my supervisors, Mr Simons and Mr Rasier, for their gracious help and fruitful discussions. Each of their comments and remarks proved most useful in improving the quality, conciseness and precision of this dissertation.

I would like to thank Mrs Renson for her important advice on constructing a survey. She also kindly accepted to proofread my survey in its final stages.

I would like to thank all of the teachers who kindly accepted to test the survey before its final stage. Also, I send my thanks to the teachers who completed and shared the questionnaire.

I would like to thank both of my parents who proved, once again, to be most supportive. I thank my mother with whom, for close to three years, I shared university updates around a morning coffee. I thank my father for being the prime inspiration for this dissertation. I would like to thank my girlfriend, whose unfaltering support soothed me many times, and who put up with me being so much focused on work. I would like to thank my University classmates for their supportive spirits.

Eventually, I would like to thank everyone who showed interest for my research.

## **Table of Contents**

TABLE OF ILLUSTRATIONS	8
1. INTRODUCTION: MOTIVATION, AIMS, RESEARCH QUESTIONS, APPROACH	10
2. PRONUNCIATION: VIABILITY, SEGMENTS AND SUPRASEGMENTALS	13
2.1. Introduction	13
2.2. Pronunciation	13
2.3. Comprehensibility, Intelligibility and Accentedness	14
2.3.1. Comprehensibility	14
2.3.2. Intelligibility	14
2.3.3. Accentedness	15
2.4. The Core Elements of Pronunciation	18
2.4.1. Segmentals	18
2.4.2. Suprasegmentals	18
2.5. Conclusion	22
3. TEACHING: WHAT AND WHEN	23
3.1. Introduction	23
3.2. Goals and Norms	24
3.3. Contexts	27
3.3.1. Teaching Contexts	27
3.3.2. The Listener	
3.3.3. The Learner	
3.3.4. Differences Between English and French	31
3.3.5. Oral Teacher Input	31
3.4. Segmentals Versus Suprasegmentals	32
3.5. When to Teach	34
3.6. Conclusion	35
4. OVERVIEW OF PRONUNCIATION TEACHING METHODS	37
4.1. Introduction	37
4.2. Phonetic Notation	37
4.3. Technology	
4.4. Drama	43
4.4.1. Shadowing and Mirroring	45
4.5. Perception	45
4.6. Feedback	48
4.7. Physical Skills and the Silent Way	49

4.8. The Verbo-Tonal Method of Phonetic Correction	50
4.8.1. Origins and the Structuro-Global Audio-Visual Method	50
4.8.2. Perception, Intonation and the Body	51
4.8.3. Concrete Applications of the Verbo-Tonal Method	52
4.9. Conclusion	54
5. TEACHING: ACTUAL PRACTICES	56
5.1. Introduction	56
5.2. Teachers' Reported Practices and Beliefs	56
5.3. Teachers' Observed Practices	59
5.4. Learners' Points of View	61
4.5. Textbooks	63
4.6. Conclusion	64
6. SURVEY: STATED PRACTICES	66
6.1. Introduction	66
6.2. Hypotheses	66
6.3. Methodology	70
6.3.1. Constructing the Survey	70
6.3.2. Structuring the Survey	72
6.3.3. The Shortcomings of the Survey	77
6.3.4. Administrating the Survey	80
6.3.5. Processing the Survey	82
6.4. Analysing the Survey	82
6.4.1. Section 1: The Respondents' Profiles	83
6.4.2. Section 2: Goals and Norms	85
6.4.3. Section 3: Reasons, Beliefs, Opinions and Training in Pronunciation	86
6.4.3.1. General Reasons for Focusing on Pronunciation	88
6.4.3.2. Learners' Age	89
6.4.3.3. Training	89
6.4.3.4. Natives and Input	90
6.4.3.5. Activities During Teacher Training	90
6.4.4. Section 4: Contents	93
6.4.5. Section 5: Material and Activities	99
6.4.6. Section 6: Vocabulary Lists and Oral Tests	109
6.4.7. Section 7: Self-Assessment	112
6.5. Discussion and conclusion	113

APPENDICES	
References	
7.2. Prospects	
7.1. Synthesis	
7. CONCLUSION	
6.5.2. Research Questions	115
6.5.1. Explicit or Implicit Approaches	113

### **Table of Illustrations**

Figure 1 – Share of Respondents in Different Levels and Stage
Figure 2 – Share of Respondents in the Secondary Education Subject Fields
Figure 3 – Years of Experience Among the Respondents
Figure 4 – Linguistic Pronunciation Activities Done During Teacher Training
Figure 5 – Pronunciation Activities Done in Language Teaching Methodology
Figure 6 – Pronunciation Element Focused on in EFL Classes
Figure 7 – Method Used to Teach Pronunciation
Figure 8 – Activities Used to Introduce New Pronunciation Elements
Figure 9 – Resources Used to Teach Pronunciation 103
Figure 10 – Moments of Feedback During Spontaneous Speech Acts 104
Figure 11 – Moments of Feedback During Aloud Readings 104
Figure 12 – Mistakes Corrected During Spontaneous Speech Acts 105
Figure 13 – Mistakes Corrected During Aloud Readings 105
Figure 14 – Pronunciation Information in Vocabulary Lists 110
Figure 15 – Focus of Pronunciation Information in Vocabulary Lists 111
Figure 16 – Weight of Pronunciation Criterion in Oral Tests
Figure 17 – Exclusion Pronunciation Criterion in Oral Tests
Figure 18 – Self-Assessment of Own Pronunciation, Mastery of Theoretical Concepts, Ability to Teach Pronunciation

Table 1 – Research Questions 10
Table 2 – Research Questions and Their Hypotheses for the Survey
Table 3 – Percentages of Agreements With Statements About Pronunciation Teaching 86
Table 4 – Importance of Pronunciation Elements in EFL Classes  94
Table 5 – Level of Problems Pronunciation Elements May Pose  95
Table 6 – Importance of Criteria to Select Pronunciation Elements to Teach
Table 7 – Percentages of Agreements With Statements on IPA
Table 8 – Frequent Pronunciation Activities in EFL Classes  100
Table 9 – Explicit and Implicit Pronunciation Teaching
Table 10 – Frequency of Feedback Types  106
Table 11 – Focus of Feedback 106

Table 12 – Frequency of Accents Used in Listening Comprehensions	. 107
Table 13 – Frequency of Integration of Pronunciation Modules Into a Teaching Sequence	108
Table 14 – Frequency of Pronunciation Combined With Grammar, Lexis, Syntax and Lang	uage
1 411010115	. 107

#### 1. Introduction: Motivation, Aims, Research Questions, Approach

My father works at an international company as a technical service manager. He works with people from India, Turkey, Germany, or from the United States, meaning that they need a common language to communicate: English. During the 2020 lockdown, he had to work from home and he decided to settle his office next to my bedroom, so many times a day, I heard him talk over the phone to some of his colleagues. Because I am studying English teaching methodology, I marvelled at my father's and his colleagues' strong foreign accents. I do not believe my father has ever received any pronunciation instruction, and yet, there he was, efficiently communicating with others, who, like him, were accented. The question then arose: Is pronunciation that important in language learning?

This is what sparked my interest in pronunciation and this dissertation ensued. Mixed with my interest in teaching, the research question became: Are francophone learners of English as a Foreign Language provided with the necessary tools to ensure viable pronunciation?

Many other questions derive from that central research question. Because I decided to conduct a survey, I can answer similar questions in both a practical (thanks to the questionnaire) and a theoretical way (thanks to the literature review). A comparison between what is advised by research and what is said to be done in class is then possible. The following table presents the 23 research questions ( $Q_1$ ) stemming from the central one; a double arrow indicates correspondence between a question discussed in theory and in practice.

#### Table 1

Research Questions Discussed Theoretically		Research Questions Discussed Practically		
$\mathbf{Q}_1$	What goals should teachers pursue	⇔	<b>Q</b> <sub>13</sub>	What goals do teachers pursue in
<b>Q</b> <sub>2</sub>	What are effective techniques to teach pronunciation?	⇔	<b>Q</b> <sub>14</sub>	What techniques do teacher use to teach pronunciation?
<b>Q</b> <sub>3</sub>	Are textbooks suitable ways to teach pronunciation?	⇔	Q15	Do teachers use textbooks to teach pronunciation?
<b>Q</b> <sub>4</sub>	How does feedback impact pronunciation teaching?	⇔	Q16	How do teachers use feedback regarding pronunciation?
<b>Q</b> 5	Is the use of a metalanguage effective to teach pronunciation?	⇔	Q17	Do teachers use any kind of metalanguage to teach pronunciation?
<b>Q</b> <sub>6</sub>	Do non-native accents have a place in class?	⇔	Q18	Do teachers call upon non-native accents in class?

#### **Research Questions**

Research Questions Discussed Theoretically		/	Research Questions Discussed Practically	
<b>Q</b> 7	What is good pronunciation?	î	Q19	What norms do teachers refer to when teaching/evaluating pronunciation?
<b>Q</b> <sub>8</sub>	What are the features involved in pronunciation?		Q <sub>20</sub>	What features do teachers teach/focus on regarding pronunciation?
Q9	How do these features impact pronunciation?		Q <sub>21</sub>	Why do teachers (not) teach pronunciation (much)?
Q <sub>10</sub>	Do teaching contexts impact pronunciation goals?		Q <sub>22</sub>	What kind of training in (teaching) pronunciation did/do teachers get?
Q11	What is the pronunciation teaching situation in some countries?		Q23	How is pronunciation integrated into the classroom?
Q <sub>12</sub>	How does teacher input influence the pupils' pronunciation?			

The research questions in the left column will be answered in the chapters 2 through 5, while the questions in the right column will be answered in Chapter 6. The reason why some theoretical questions do not get a practical is because their answers will serve to posit hypotheses and practical questions. Said differently, the impact of features involved in pronunciation ( $Q_8$ ), the contexts ( $Q_{10}$ ), and the effects of teacher input ( $Q_{12}$ ) will allow to posit some hypotheses when it comes to understanding why teachers do or do not teach pronunciation ( $Q_{21}$ ), or what features they rather focus on ( $Q_{20}$ ). An overview of the pronunciation teaching situation in some countries ( $Q_{11}$ ) will allow to hypothesise answers to every question in the right column of Table 1. The countries that will be analysed are the ones surveyed in various articles from scientific literature and these include: Finland, France, Germany, Poland, Spain, Switzerland, New Zealand, the United States, Australia, Canada and Uruguay (see Chapter 5).

At the beginning of every chapter, the questions that are dealt with are stated; hypotheses are given only for  $Q_{13}$  to  $Q_{23}$  included (i.e., the ones from Chapter 6) because the other research questions require formal knowledge. Chapter 2 deals with  $Q_7$ ,  $Q_8$ ,  $Q_9$ . Chapter 3 provides an answer to  $Q_1$ ,  $Q_6$ ,  $Q_{10}$ ,  $Q_{12}$ . Chapter 4 copes with  $Q_2$ ,  $Q_4$ ,  $Q_5$ . Chapter 5 sheds light on  $Q_3$ ,  $Q_{11}$ .

The approach taken in this dissertation is the following. Literature is reviewed in the next four chapters to better comprehend pronunciation as a whole. Chapter 2 gives a definition of pronunciation, advances three concepts to understand what good pronunciation may mean and focuses on pronunciation features. Chapter 3 zooms in on teaching contents and contexts: the goals and norms, the impact of the context, the importance of pronunciation features and the moments to teach pronunciation are reviewed. Chapter 4 dives into the different techniques

available to put into practice what is detailed in the two previous chapters: an array of pronunciation methods and exercises are presented. Chapter 5 looks at teachers' actual practices through examining surveys, questionnaires, and interviews conducted with them, through observing what instructors actually do in class, through asking learners what they think of pronunciation, and through discussing the place of pronunciation in textbooks. Chapter 6 discusses the results obtained after sending the questionnaires to English teachers working in Belgium. Chapter 7 concludes this dissertation by comparing the two types of research questions.

Some extra information relating to the dissertation overall must be provided before diving into the different research questions. Throughout every chapter, and throughout the introduction, the American Psychological Association guidelines (seventh edition) are followed and applied. Whenever a phonetic transcription is given for a word, or whenever a word is given to illustrate a sound, the *Cambridge English pronouncing dictionary* (Jones, 2011a) is used. As recommended by supervisor Mr Simons and for the sake of easiness, the appendices are provided to the supervisors and to the readers on USB sticks, in both Word and PDF format.

#### 2. Pronunciation: Viability, Segments and Suprasegmentals

#### **2.1. Introduction**

This chapter looks into what pronunciation is, what it entails and into three concepts that make the notion of good or viable pronunciation clearer; these are comprehensibility, intelligibility and accentedness. Each theme is discussed in one separate subsection. The research questions dealt with are: "What is good pronunciation  $(Q_7)$ ?", "What are the features involved in pronunciation  $(Q_8)$ ?", "How do these features impact pronunciation  $(Q_9)$ ?"

#### 2.2. Pronunciation

The very first element that must be defined to speak of pronunciation, is pronunciation itself. Yet, very few authors seem to explicitly define it, so an adequate starting point could be a dictionary; *pronunciation* is defined as "the way in which a word is pronounced" ('Pronunciation', 2012, p. 576) and *to pronounce* is defined as "to make the sound of a word or part of a word" ('Pronounce', 2012, p. 576). Here, the focus lies on the production of sounds and very much resembles what authors call *pronouncing*, that is, the physical elements, or the "motor skill component of pronunciation" (Messum & Young, 2021, p. 169): the placing of the tongue, the position of the lips, the vibrations of the vocal folds, etc. Still, there is more to it than production. Researchers (Setter & Jenkins, 2005, p. 1) emphasise the perception involved in pronunciation: one perceives sounds (i.e., segmentals, see 2.4.1.) that are produced and that are combined with larger speech features (i.e., suprasegmentals, see 2.4.2.).

Additionally, Pennington (1988) defines pronunciation as:

Conveying many different types of messages to a hearer related to the information structure of a discourse, the speaker's attitude and mood, and other social and psychological features of the speaker or of the relationship between the speaker and the hearer. (p. 204)

She puts emphasis on a speaker (i.e., production), a hearer (i.e., perception) and on the relationship between the two; pronunciation then possesses a social component. A speaker may adapt the way they pronounce a word to better fit the communication situation or needs, such as register needs (e.g., formal or informal language) (Pennington, 2019, pp. 373–375).

Detey et al. (2016, p. 19) provide a comprehensive definition: pronunciation entails acquiring a phonological and phonetic system (and its social representations), at different levels, that is, production/perception and segmentals/suprasegmentals. Phonology deals with the sound system, aims at finding patterns and structures (Crystal, 2008, pp. 365–366; van Oostendorp, 2020, pp. 3–6), while phonetics studies the "physical" (van Oostendorp, 2020, p. 3), "articulatory", "acoustic" features of sounds (Crystal, 2008, pp. 364–365). What is at stake with pronunciation is then much more than mere sounds (Detey et al., 2016, p. 19).

#### 2.3. Comprehensibility, Intelligibility and Accentedness

#### 2.3.1. Comprehensibility

*Comprehensibility* is how easy or difficult it is for someone to understand a specific piece of speech (Derwing et al., 1998, p. 396; Derwing & Munro, 2009, p. 478; Derwing & Munro, 2014, p. 40; Galante & Piccardo, 2022, p. 376; Gordon & Darcy, 2016, p. 57; Munro & Derwing, 2001, p. 454; Yenkimaleki & van Heuven, 2021, p. 9). Thus, comprehensibility bears no relation to what one understands but it is related to how much "effort" (Derwing, 2010, p. 29), to "the amount of time" (Derwing & Munro, 2009, p. 478), or to "the amount of work it takes listeners to process speech" (Levis et al., 2016, p. 7), hence, some using "processability" as a synonym for comprehensibility (Thomson, 2018, p. 225). Therefore comprehensibility is a totally subjective judgment of speech (Derwing et al., 1998, p. 396) and is measured through the use of scales assessing difficulty of speech processing (Derwing et al., 1998, p. 396; Levis et al., 2016, p. 7). Because repeated exposure to particular speech entails less effort required for processability (Gass & Varonis, 1984, p. 77), teachers may not be best suited for judgment on their pupils' comprehensibility (Munro & Derwing, 2006, p. 521). Repeated exposure to familiar errors may eventually lead to new norms in the classroom (Intravaia, 2000, p. 238).

#### 2.3.2. Intelligibility

*Intelligibility* refers to the extent of actual understanding or to how much of an utterance was understood (Derwing et al., 1998, p. 396; Derwing & Munro, 2009, p. 479; Gordon & Darcy, 2016, p. 57; Gut, 2009, p. 254; Munro & Derwing, 1999, p. 289, 2001, p. 454), recognized (Field, 2005, p. 401; Yenkimaleki & van Heuven, 2021, p. 2) or decoded (Levis, 2007, p. 188; Levis et al., 2016, p. 6). Then, similarly to comprehensibility, central to the concept is both the listener and the speaker (Levis, 2005, p. 372; Pennington, 2021, p. 18); intelligibility is often implied with regard to native speakers (NSs), but it may be needed to extend it to other listeners, like non-native speakers (NNSs) and "automated speech recognition assistant-based personal

assistants such as Apple's Siri and Google's Alexa" (Rogerson-Revell, 2021, p. 201). With intelligibility, the focus lies on the listener and their perception (Field, 2005, p. 400). It is measured through the use of transcriptions which are then compared with the actual utterances pronounced; depending on the number of missing or misrecognised words, an intelligibility score may be assigned (Munro & Derwing, 1999, p. 289). On the basis of all these elements, intelligibility can then be argued to be the most crucial of the three concepts surrounding pronunciation, that is, comprehensibility, intelligibility, and accentedness (Derwing et al., 2013, pp. 23–24).

Several factors may impact actual understanding. One of them is the fact that a native listener applies "speech processing strategies that are appropriate to their native English phonology" (Zielinski, 2008, p. 70) to non-native speech, which was investigated in Rasier et al. (2011, p. 235), who demonstrated that native listeners find non-native speech less intelligible than native speech, while non-natives find it easier to decode non-native speech than native speech; that contrast points to different patterns being applied (see 3.3. for more information on the impact of language backgrounds).

Besides, Gass & Varonis (1984, p. 81) found that, among other elements, being accustomed to non-native speech positively influences intelligibility scores (see 3.3.). They asked two Japanese and two Arabic speakers to read out loud a fable and sentences related or unrelated to the story that 142 NSs of English listened to, who then transcribed the sentences and summarised the story. The factor that had the greatest influence on the transcription scores was the "familiarity with the topic of discourse" (Gass & Varonis, 1984, p. 81), followed by being accustomed to non-native speech and to accents. The last impactful element was familiarity with a specific speaker. Despite the flaw of having four speakers in total, the results shed light on intelligibility and its sources.

#### 2.3.3. Accentedness

*Accentedness* revolves around a difference between speakers' pronunciation (Murphy, 2014, p. 261) and may be defined as "divergences from the norms of [a] speech variety" (Flege, 1984, p. 692), which then entails that every speaker does necessarily have an accent since there exist different (equally good) English norms (Derwing & Munro, 2009, p. 476; Ioup, 2008, p. 54; Saito & Lyster, 2012, p. 597). Many authors do insist on the differential or deviational character of accent (Bent et al., 2007, p. 331; Derwing & Munro, 2009, p. 476; Gut, 2009, p. 253; Levis

et al., 2016, p. 7), one even stating that accent is synonymous with a "style of pronunciation" (Pennington, 2019, p. 377).

Factors influencing accentedness are numerous and range from physical properties to social ones (see below). It is of interest to note that there are some factors that do not influence accentedness like "education levels and intelligence" (Gordon & Darcy, 2016, p. 57).

About the impactful social properties first, an accent is a way to asses one's belonging to a group or simply to assess one's identity (Levis, 2005, pp. 374–375), whether it be "regional, social or ethnic" (Setter & Jenkins, 2005, p. 1). As a result, a speaker may adapt their speaking style in order to be approved of (Beebe & Giles, 1984, p. 8) or they may adopt a certain way of pronouncing for its social prestige, like the French accent, among other European ones, which may be seen as a sign of intelligence (Derwing & Munro, 2009, p. 484). It all boils down to projecting a specific wanted image (Setter & Jenkins, 2005, p. 5). Besides, the social aspect of accent goes both ways: due to high social pressures from their first language (L1) group, second language (L2) speakers of English may not desire attaining native pronunciation for they may fear being seen as lacking loyalty to their L1 group (Gatbonton et al., 2005, p. 504).

Furthermore, non-social elements such as "intonation patterns" (Jilka, 2007, p. 91), fluency (Gut, 2009, p. 300), pronunciation speed (Munro & Derwing, 2001, p. 464) do have an impact on accentedness. Gut (2009) lists language-independent factors exerting an influence on accentedness: "age at the beginning of the language learning, length of residence [in an English speaking country] and length of instruction" (p. 271). She also examined accentedness in two groups who had either taken a 6-month pronunciation course or had been abroad for 6 to 9 months and did not notice any significant improvement in either group (Gut, 2009, p. 278). In another study, Gut (2007, p. 165) examined vowel reduction in two groups who had also either followed a pronunciation course or had stayed abroad, but this time, they had both significantly improved.

Other researchers (Munro & Derwing 2001, p. 464) noticed that increasing a NNS's utterance by 10% had positive effects on accentedness and comprehensibility scores. This may be explained by the fact that the listeners must process speech faster and are therefore less likely to notice errors. Still, speeding up the utterances too much, or speaking too fast, does have the opposite effect.

Additionally, *transfer*, that is, learned languages impacting the subsequent language learning (Edwards & Zampini, 2008, p. 2) is argued to play a role in accentedness. While some

authors state that it does explain accent (Jilka, 2007, p. 83; Mehlhorn, 2007, p. 211; Rasier & Hiligsmann, 2009, p. 3), like transfer of "vowel duration ... voice onset time ... syllable structure production" (Ioup, 2008, p. 43), but remain careful when generalising, others (Bent et al., 2007, p. 331) consider transfer as the major cause of accents. Not only does transfer play a role in production, but in perception as well, especially in intonational meaning (e.g., surprise) (Chen, 2009, pp. 404–406) (see 2.4.2.).

When compared to both intelligibility and comprehensibility, accentedness is probably the most noticeable of the three (Derwing et al., 2002, p. 246; Derwing & Munro, 2009, p. 476). Flege's experiment (1984) is quite telling with regard to the saliency of accentedness. He tested French accent detection in samples of varying length, ranging from full phrases, isolated syllables, parts of syllables, to the very start of the word "two", that is, the "first 30 ms [...] or the release-burst of /t/" (Flege, 1984, p. 692); the listeners were successful in their identification no matter the length of the sample heard or their familiarity with the accent they were presented with. Similarly, Munro and Derwing (1999, p. 294) noticed that listeners are quite able to differentiate a non-native from a NS, even when a word is played backwards (Derwing & Munro, 2009, p. 477). *Voice quality* or "the characteristic resonance" (Crystal, 2008, p. 398) is argued to be the reason why listeners can determine nativeness based on backwards speech (Derwing & Munro, 2009, p. 477).

What is more, Munro & Derwing (1999, p. 303) looked at the connections between comprehensibility, intelligibility, and accentedness, and showed that speakers may be completely intelligible and comprehensible despite having strong accents. Unexpectedly, in their study, five utterances produced by non-natives received perfect intelligibility scores, and these were the only ones to get such possible scores. A conclusion is that accent, however strong it may be, does not necessarily negatively impact intelligibility; yet, the opposite situation seems to be always true: when a speaker is deemed unintelligible, their accent will be rated as high (Derwing & Munro, 2009, p. 479).

Moreover, accents are, in their very nature, tied to norms. Stress and rhythm do play a major part in defining English accent norms (Diana, 2010, p. 12). What further differentiates them is specific realisations of sounds or words; American pronunciation (or General American [GA]) is said to be rhotic, meaning /r/ is usually pronounced wherever it is in a word, while British accents (such as BBC Pronunciation) are typically non-rhotic (Jones, 2011b, p. vi). Another feature that is found in American English but not in British English is the "lax vowels [or] British short vowels [produced] with less oral tension" (p. ix).

Regarding names, *BBC English* or *BBC Pronunciation* (i.e., pronunciation features of newsreaders employed at the BBC) will be preferred to Received Pronunciation (or traditionally RP) for it has fallen out of use (Jones, 2011, p. xii); as far as the British standard is concerned, Sangster (2011, p. xxix) emphasises that BBC employees, while they could and still can be advised on how to keep an accurate pronunciation, were never and are not submitted to a clear standard, which explains why several accents can now be heard on the international British Broadcasting Corporation. Similarly to BBC Pronunciation, *GA* refers to how newsreaders from national American channels, who do not carry features of accents typical of New England of the Southern States, speak (Jones, 2011b, p. xii).

#### 2.4. The Core Elements of Pronunciation

When speaking of pronunciation, research distinguishes between two levels: the segmental and suprasegmental level (Couper, 2021, p. 137).

#### 2.4.1. Segmentals

*Segmentals* (or segments) are defined as the "individual sounds" (Foote et al., 2012, p. 3), the "phonemes" (Field, 2005, p. 402), the "speech sounds" (Hu et al., 2013, p. 366), or as the "isolated consonants and vowels" (Kissling, 2013, p. 722), including their "language specific combinations" (Gordon & Darcy, 2016, p. 58). In other words, segmentals may be called "simple sounds" (Chun, 2002, p. 3). They are said to be made of "features", which are body movements (van Oostendorp, 2020, pp. 21, 41). Instances of features are "labial" (i.e., relating to the lips), "voiced" (i.e., relating to the vocal folds) and "stop" (i.e., release of air in the larynx after it had been blocked) (pp. 26-31); combined, a labial voiceless plosive is the phoneme /p/ (as in "put") and the labial voiced stop is /b/ (as in "but"). Other examples of segmentals are /p/ (as in "hot") or /æ/ (as in "cat"). The correct production of some segments is important to ensure intelligibility: the correct production of vowels and of initial consonants are positively correlated with intelligibility, and any errors at the beginning of a word is more detrimental than an error at the end (Bent et al., 2007, pp. 341–344). Those findings were partly corroborated by Zielinski, (2008, p. 76) (see 2.4.2.).

#### 2.4.2. Suprasegmentals

*Suprasegmentals* include larger aspects of pronunciation than segments (Foote et al., 2012, p. 3), such as word stress, rhythm, intonation (Field, 2005, p. 402; Gordon & Darcy, 2016, p. 58; Hu et al., 2013, p. 366) and these are often collectively referred to as prosody (Baker, 2011, p.

263; Derwing & Munro, 2014, p. 40; Field, 2005, p. 402). While segmentals are phonemic units, suprasegmentals are "phonological units" and have scope over more than one vowel or consonant sound (Chun, 2002, p. 3). Central to suprasegmentals or prosody are the physical and pragmatic properties (Busà, 2007, pp. 167-169; Cauldwell, 2011, p. xxi; Chun, 2002, pp. xiv– xv): prosody does indeed distinguish questions from statements and allows speakers to convey emotions or politeness.

Gussenhoven (2002) differentiates between different suprasegmental-related codes which are interpreted on an affective (i.e., speaker's characteristics) and informational (i.e., characteristics of the message) level: the *frequency code* (i.e., a low or high pitch<sup>1</sup> due to properties of the speech-productive organs such as the size of the larynx), the *effort code* (i.e., the amount of energy needed to produce speech) and the *production code* (i.e., the links between breathing and energy) (p. 1). Affective and informational interpretations of the frequency code are respectively "masculinity, authoritativeness/assertiveness, and protectiveness (low pitch) and femininity, submissiveness/friendliness, and vulnerability (high pitch) ... 'certainness', leading to ... the difference between statements and questions" (p. 9). Affective and informational interpretations, relating to both ends of an utterance: a high pitch at the end indicates topic continuation, while a low pitch marks the end of a topic or, of a speech turn (p. 5).

Similarly, Chun (2002, pp. 77-78) identifies four functions of intonation: *grammatical* (i.e., telling a question from a statement), *attitudinal* (i.e., expressing feelings), *discourse* (i.e., signalling new information, contrast, illocutionary force of a speech act, finality or signalling the end of a turn), and *sociolinguistic* (i.e., establishing a link with a group). Some examples follow. In English, wh-questions typically end with a falling intonation (p. 209), falling intonation with a large amplitude may express anger (p. 222), "very high or very low pitch" may signal "emphasis" (p. 228), and "British female voices [are] in general relatively high-pitched" (p. 236). Chun (2002, p. 75) assigns intonation a sociolinguistic function: marking someone as having a foreign accent, thus belonging to NNSs (see 2.3.3. and 3.4.).

<sup>&</sup>lt;sup>1</sup> *Pitch* is defined as the perception of the vibration of the vocal folds; pitch refers to the height of the different sounds and is measured in Hertz (Chun, 2002, p. 5). A more comprehensive definition is given in the coming paragraph.

When attempting to describe the suprasegmental level, a distinction needs to be made between how prosody is expressed that is, frequency, duration, intensity and how they are perceived, that is, pitch for frequency, length for duration, and loudness for intensity (Chun, 2002, p. 4). *Pitch* refers to the perception of a fundamental frequency (F0), which is the vibration rate of the vocal cords; pitch thus represents the varying heights of individual sounds and is measured in Hertz; the role of pitch is to stress syllables (p. 5). *Length* refers to the span of the sounds, or their physical duration, or the amount of time a sound lasts (p. 6). Finally, *loudness* refers to how much energy a speaker puts into producing a sound; thus loudness is the "size of the vibrations of the vocal cords" and is then measured in decibels (p. 6).

Furthermore, suprasegmentals consist of three aspects of pronunciation as mentioned above: stress, rhythm, and intonation.

*Stress* is a polysemic term and may refer to two phenomena: word (or lexical) and sentence stress (Chun, 2002, p. 9; Frost, 2011, p. 68). Stress relates to the marking of prominence in a syllable in a word (lexical) or of a word in an utterance (sentence) (Cauldwell, 2011, p. xxi) for contrastive or emphatic purposes (Hahn, 2004, p. 202), or, in other words, "to mark the most newsworthy piece of information in an utterance" (Chun, 2002, p. 48). Pitch and loudness come into play when stressing a syllable (p. 6): an increased duration, intensity, and increased volume are typical of stressed syllables (Hahn, 2004, pp. 201–204; van Oostendorp, 2020, pp. 78, 125). Besides, duration is considered by some to be the major correlate of stress (Sluijter & van Heuven, 1996, p. 4); Frost (2011, p. 68) adds the formant structure to the list, but Sluijter & van Heuven (1996, p. 4) state that it has more to do with sentence stress than with lexical stress. Some authors (Rasier et al., 2011, p. 227; Sluijter & van Heuven, 1996) do distinguish between accent (sentence stress) and stress (lexical stress); accent may also be used to refer to both phenomena<sup>2</sup> (Chun, 2002, p. 7).

On the other hand, *rhythm* refers to a certain pattern, or to "strong beats falling on the stressed syllables of phrases and sentences" (Chun, 2002, p. 8). As opposed to French which is often called a syllable-timed language, English is often defined as a stress-timed language (Frost, 2011, p. 71; Setter & Jenkins, 2005, p. 8), because natives are said to take the same time to go from one stressed syllable to the next, independently of the number of unstressed syllables in between, resulting in phonetic reduction of the unstressed syllables (Chun, 2002, p. 175).

 $<sup>^{2}</sup>$  In this dissertation, "accent" will not be used to refer to either sentence stress or lexical stress; only "stress" will be used instead.

Nevertheless, these categories have been called into question (Chun, 2002, p. 9; Gut et al., 2007, pp. 11-12) for it has yet to be confirmed experimentally (Setter & Jenkins, 2005, p. 8).

Regarding intonation, Chun (2002, pp. 3-4) notices that it is often used synonymously with prosody and suprasegmentals, but she uses the definition given in Crystal (2008): "the distinctive use of patterns of pitch or melody" (p. 252).

Suprasegmentals are argued to play a bigger role in rendering an utterance intelligible and comprehensible than segmentals. Hahn (2004, p. 210) investigated the role of stress and found that wrongly placed stress has a great impact on listeners' perception. She tested 90 freshman students from a Midwestern university on their information retention; those were split into three groups and assigned an identical text to answer questions on and to listen to. What differentiated the text was the lexical stresses that were either correctly placed, misplaced or totally absent. The results showed that the students who had listened to the correctly stressed version remembered more than the two other groups, showing enhanced communication when stress is correctly placed and implying an impediment on communication when stress is not used appropriately (p. 215). The absence of stress also gave students the impression that the speaker spoke rapidly, as conveyed in their comments (p. 221).

Field (2005, p. 410) got similar results when he asked native and non-native listeners to transcribe disyllabic words where the stress pattern had been either switched or had not been switched, along with the vowel quality having changed; the results for both groups (natives and non-natives) showed that shifting stress negatively impacted intelligibility, whether the vowel quality had been changed or not, but the latter scenario had a lesser impact than the former case.

In addition, Zielinski (2008) focused on cues listeners use when confronted with an unintelligible utterance. The researcher asked three NSs to transcribe 50 utterances recorded during a discussion with three proficient L2 English Asian speakers. She found that the listeners always relied on stresses, and, less consistently on individual sounds. Indeed, the stress patterns were found unchanged in the transcriptions when utterances or words were difficult to understand, while segments were transcribed as is in fewer cases (p. 76). Word initial consonants and vowel sounds in stressed syllables were the listeners' focus (p. 76). In other words, stressed syllables were the most important cues when decoding the utterances, as they were preserved in many cases even if the patterns were non-standard. However, as mentioned by the author herself (p. 81), the major drawback of this study is the limited number of participants (i.e., three listeners and three speakers).

#### **2.5.** Conclusion

Going back to the three research questions structuring this chapter, good pronunciation may first be detailed thanks to three concepts (Derwing et al., 1998; Murphy, 2014): comprehensibility (i.e., easiness to understand), intelligibility (i.e. actual understanding), and accentedness (i.e., pronunciation differences between speakers and their norms) ( $Q_7$ , "What is good pronunciation?"). Because pronunciation involves both a speaker and a listener, these characteristics, which all involve at least two people, may form criteria to replace the vague notion of being good, but intelligibility remains the most important criterion to assess one's pronunciation (Derwing et al., 2013).

Going further, many factors influence comprehensibility (e.g., repeated exposure leading to easier processing) (Gass & Varonis, 1984), intelligibility (e.g. NNSs find NNSs more intelligible than NSs) (Rasier et al., 2011) and accentedness (e.g. conscious decision to sound foreign or native) (Gatbonton et al., 2005; Setter & Jenkins, 2005). Yet, the three factors may not be intercorrelated as some speakers may be perfectly intelligible and comprehensible, in spite of being strongly accented (Munro & Derwing, 1999); on the other hand, speakers who are unintelligible appear to always be considered accented (Derwing & Munro, 2009).

Finally, two levels are traditionally distinguished when speaking of pronunciation (Couper, 2021): the segmental level, which encompasses vowel and consonant sounds, and the suprasegmental level, which encompasses lexical stress, rhythm and intonation ( $Q_8$ , "What are the features involved in pronunciation?"); besides intonation carries meaning (Chun, 2002; Gussenhoven, 2002). Both are argued to be critical to intelligibility, but the latter level is considered to play a more critical role ( $Q_9$ , "How do these features impact pronunciation goals?"): correct production of vowels and of initial consonants is important (Bent et al., 2007; Zielinski, 2008) and so is the correct placement of stress (Hahn, 2004).

#### 3. Teaching: What and When

#### **3.1. Introduction**

The aim of this chapter is to give a review of what goals and norms teachers may pursue, how contexts (e.g., the difference between the L1 and L2) impact these, which level leads to a greater improvement in intelligibility and when to implement pronunciation teaching in the classroom setting. The research questions that are answered here are: "What goals should teachers pursue in pronunciation teaching  $(Q_1)$ ?", "Do non-native accents have a place in class  $(Q_6)$ ?", "Do teaching contexts impact pronunciation goals  $(Q_{10})$ ?", "How does teacher input influence the pupils' pronunciation  $(Q_{12})$ ?"

The first question one may ask is whether pronunciation deserves any attention in class. Levis (2005, p. 369) does believe pronunciation should be addressed in class in some shape or form, but should not become the main point focused on in class. Yet, many other questions arise when teaching pronunciation: Is pronunciation teaching effective at all? How should pronunciation be taught? What pronunciation goals should the teacher set? What pronunciation features should be taught? Before answering these broad questions, a few outlines are given as to make matters clearer.

First of all, much pronunciation research is actually done in controlled environments, and not in regular classes, as acknowledged by Foote et al. (2016, p. 184), thus rendering any conclusions on the efficiency of any method or simply on the best method to adopt not "straightforward" (Gut et al., 2007, p. 5). Additionally, teachers are the ones who ultimately decide what to teach and how to teach based on their pupils or students (Chun, 2002, p. 200), and embedding teaching material thoroughly in theoretical research is not expected from teaching professionals (Gut et al., 2007, p. 15), especially because mixing teaching and research is a daunting task (Chun, 2002, p. 43). What researchers then advocate is to arm teachers with the necessary research-based expertise to identify the learners' issues (see 3.2.), and with ways to help the learners improve (Derwing & Munro, 2005, pp. 387, 392).

As a result, many researchers (Chun, 2002, p. 144; Derwing, 2010, p. 26; Derwing & Munro, 2005, p. 387; Derwing & Munro, 2014, pp. 46-52; Jenkins, 2004, p. 109; Levis & Grant, 2003, p. 14; Pennington, 2019, p. 375) put forward several principles that should serve as guidelines on pronunciation teaching. A first principle is to teach both the segmental and suprasegmental level (Jenkins, 2004, p. 109), with a slight focus on the latter (Levis & Grant, 2003, p. 14). On that matter, Chun (2002, p. 144) is in favour of combining both a top-down

and a bottom-up approach: a *top-down approach* centres on communication and intelligibility (LaScotte et al., 2021, p. 145) and their links with sound, therefore focusing on the suprasegmental level (Derwing et al., 1998, p. 397; Pennington, 2019, p. 380), while a *bottom-up approach* centres on isolated or combined segments (Chun, 2002, p. xiv; LaScotte et al., 2021, p. 145; Pennington, 2019, p. 372). The question of the main teaching target is also central to other authors (Derwing, 2010, p. 26; Derwing & Munro, 2005, p. 387; Derwing & Munro, 2014, pp. 46-52; Pennington, 2019, p. 375) for teaching pronunciation should always focus on enhancing communication or speaking (Levis & Grant, 2003, p. 14), that is, making utterances more intelligible and more comprehensible. Eventually, better use of technology and of feedback is also advocated (Derwing, 2010, p. 26; Derwing & Munro, 2014, pp. 46-52).

#### 3.2. Goals and Norms

Historically speaking, pronunciation teaching has always had two trends, which both revolve around the principles of what good pronunciation is considered to be: "the nativeness principle and the intelligibility principle" (Levis, 2005, p. 370). Both principles hold different views on what goals to achieve in pronunciation teaching, namely a native accent, that is traditionally BBC Pronunciation and GA (Diana, 2010, p. 13; Marks, 2011, p. xxvi) and intelligible communication respectively (Levis, 2005, p. 370). Besides, if BBC Pronunciation and GA are conventionally considered as the norms to follow, they and other native norms do remain unclear (Taylor, 2006, p. 52) and are still debated (Low, 2021, p. 26), probably because of the lack of clear-cut rules on English pronunciation (Jones, 2011b, p. xii) and some "controversial" pronunciations<sup>3</sup> (p. xiv).

The nativeness principle used to be the dominant paradigm in the first half of the 20<sup>th</sup> century and is argued to still exert an influence on current teaching practices (Levis, 2005, p. 370), especially in teacher training (Detey et al., 2016, p. 18); a reason may be how noticeable accent is (see 2.3.3.) and teachers' belief that native accents are "an achievable ideal [while it actually is] an exception" (Levis, 2005, p. 370). Nowadays, authors plead for the intelligibility principle to be issued in class (Couper, 2021, p. 137; Diana, 2010, p. 13; Galante & Piccardo, 2022, p. 385; Mompean & Fouz-González, 2021, p. 161) for nativeness is a mere "ideal"

<sup>&</sup>lt;sup>3</sup> Only "kilometre" (/ki'lbm.i.tə<sup>r</sup>/, /ki'lbm.i.tə<sup>r</sup>/, /'kil.əo,mi:.tə<sup>r</sup>/, was regarded as inappropriate and American-inspired, but it has now become the dominant pronunciation" (Jones, 2011a, p. 277). Other pronunciations are said to be frowned upon or disapproved of. "Research" with the stress on the first syllable (/'ri:.s3:tʃ/) (Jones, 2011e), "liquorice" (/'lik.ər.1ʃ/) (Jones, 2011d) are two instances of disapproved pronunciation (Jones, 2011a, pp. 418, 291).

(Bowen, 1972, p. 84), "difficult to acquire ... and probably unnecessary for most learners" (Marks, 2011, p. xxvi), which is a viewpoint shared by others (Galante & Piccardo, 2022, p. 38; Galante & Thomson, 2017, p. 117). Yet, based on corpus studies,<sup>4</sup> Gut (2009) states that "near-native attainment of L2 phonology is possible and that it is worth trying to attain it" (p. 306).

This shift of paradigm has been taking place over the last decade (Detey et al., 2016, p. 18; Pennington, 2021, p. 4) and moving away from native norms does make sense in the 21<sup>st</sup> century because the number of people whose L2 is English is much bigger than the number of natives (Jenkins, 2011, p. xxii; Mompean & Fouz-González, 2021, p. 161): out of 1.452 billion English speakers, there are 373 million natives, which equals to a bit less than one fourth of the total number of speakers (Ethnologue, 2022; Lane, 2023).

If they do not explicitly make reference to intelligibility and its principle, many authors do however emphasise putting the focus on the right place (Derwing & Munro, 2009, p. 482). The ultimate goal should always remain the following: identifying and working on the learners' needs in the speaking situation, particularly the ones that may lead to a breakdown in communication (Bowen, 1972, p. 87; Derwing & Munro, 2014, p. 48; Galante & Thomson, 2017, p. 117; Gordon & Darcy, 2016, p. 58; Kissling, 2013, p. 735; Levis, 1999, p. 373; Levis & Grant, 2003, p. 14; Rasier, 2011, pp. 126-127; Taylor, 2006, p. 53; Thomson, 2012, p. 19). Precisions may be added as to this final objective: it must be "realistic" (Bowen, 1972, p. 87) and must cater to both the production and the perception level (see 4.5.) (Derwing & Munro, 2014, p. 48). Focusing on the learners' needs (and thus will) also implies that whatever goal model the learners choose, a "patronizing [sic] approach" should be avoided (Taylor, 2006, p. 51) and that teacher may have to emphasise "accent addition [instead of] accent reduction" (Setter & Jenkins, 2005, p. 12). In other words, in the classroom setting, foreign accents should not be considered as "inferior" (see 2.3.3.) (Gatbonton et al., 2005, p. 506).

In order to help teachers select features to work on, a diagnostic test may be an appropriate first step (Couper, 2021, p. 136), or the concept of "functional load" may prove useful to prioritise segmental errors based on how they may affect intelligibility (King, 1967; Brown, 1974, p. 54 as cited in Brown, 1988, p. 594; Catford, 1987, as cited in Derwing &

<sup>&</sup>lt;sup>4</sup> The corpus used was the LeaP corpus, which is a spoken language corpus of 73,941 words and more than 12 hours of recording. The speakers are learners of English or of German; there are 176 annotated recordings for L2 English and 183 annotated recordings for L2 German; there are a few native recordings for comparison (Gut, 2009, p. 65).

Munro, 2014, p. 48). The *functional load* is a scale ranking segmental contrasts or pairs (Munro & Derwing, 2006, p. 522) depending on their frequency and on the number of words they differentiate (Brown, 1988, pp. 594-597; King, 1967, p. 831). The higher the functional load, the more words they distinguish; the lower the functional load, the fewer words they distinguish. An example of a high functional load pair is "/n/ and /l/ [as in] no/low, night/light" (Derwing & Munro, 2014, p. 48), and an instance of a low functional load pair is "/u./ and /u/ [as in] pool/pull, fool/full" (Brown, 1988, p. 600). Brown (1988, p. 604) ranks phoneme pairs on a 10-point scale, the higher the number, the higher the functional load: /e/, /æ/ and /p/, /b/ are at level 10, / $\alpha$ /, / $\alpha$ / and / $\theta$ /, / $\delta$ / are at level 5, and / $\alpha$ /, / $\alpha$ I and /f/, / $\theta$ / are the lowest level, that is the first level; Catford (1987, as cited in Derwing & Munro, 2014, p. 49) uses percentages and makes a distinction between initial and final consonants and vowel sounds. The concept of functional load proved efficient in ranking segmental errors: the high functional load errors in sentences spoken by Cantonese people were more detrimental to comprehensibility and contributed to a higher ranking on accentedness than the low functional load mistakes (Munro & Derwing, 2006, p. 529).

Nevertheless, functional load may not be suitable when a non-native meets a monolingual native: in that context, it has been shown that non-bilingual speakers of English rely more on the sounds than on the context (Jenkins, 2000, p. 70). Put differently, the functional load sets priorities as to the most important sounds when it comes to intelligibility, but it seems that monolingual English speakers rely on every sound, which all, to NNSs, carry high functional loads.

The question of setting a pronunciation goal is undoubtedly linked to the models learners may be exposed to (see 3.3.2 for details on teacher input). Since teaching is gradually moving away from the native speaker norm, appropriate models to listen to or to imitate may include NNSs who are both intelligible and comprehensible. These specific models form "accessible" (Murphy, 2014, p. 259) and "appropriate [targets]" (Kissling, 2013, p. 737), accessible in the sense that few learners will ever attain native-like pronunciation proficiency and may then want to aspire to sound like proficient non-native models, and appropriate in the sense that it is statistically more probable that English learners will interact with non-natives than natives (see above). In addition, solely using native models<sup>5</sup> in class (e.g., through listening

<sup>&</sup>lt;sup>5</sup> Exclusive and constant use of native models may only occur in English as a Second Language settings, where the teacher is native.

comprehensions) implicitly conveys that learners should and will sound like natives (Murphy, 2014, p. 259).

As an example, Murphy (2014) provided 36 pronunciation specialists with a 31-item questionnaire on Javier Bardem's pronunciation, an Oscar-winning Spanish actor (Academy of Motion Picture Arts and Sciences, n.d.; Spiardi, n.d.), during a recorded interview. After analysis of the questionnaires, Murphy (2014, p. 263) concluded that Bardem could serve as a suitable model for he was considered both intelligible and comprehensible, despite some segmental errors and his accent. Murphy (2014) also stresses how rich of a model he may be: Bardem's "paralinguistic, linguistic and rhetorical characteristics" (p. 256) were considered as contributing to his intelligibility and comprehensibility. Bardem's interview could then serve as a way of emphasising the roles played by features other than the linguistic ones. Ultimately, what Murphy (2014, pp. 258, 266) advocates, is avoiding putting too much focus on native models, but instead raising the learners' awareness of non-native models and the positive features they may illustrate, points on which Low (2021, p. 28) and Jenkins (2000, p. 91) agree: exposure to non-traditional varieties of English, including non-native accents, is necessary, especially when the class is monolingual. Marks (2011, p. xxvi) justifies listening to different varieties by saying that pronunciation generally varies and changes from lone words to words in speech (see 4.5.). Derwing (2010) does not make any mention of NNSs but does insist on "[exposing students] to multiple voices from a range of ages and dialects" (p. 28).

#### **3.3.** Contexts

#### 3.3.1. Teaching Contexts

Teaching is impacted by the teaching contexts, that is, whether the students or pupils learn English as a second language (ESL), a foreign language (EFL), an international language (EIL) and as a lingua franca (ELF). For instance, it may be argued that in EFL contexts, where communication goals are to communicate with natives, native accents should be taken as the norms to attain (Jenkins, 2002, p. 85; Setter & Jenkins, 2005, p. 12). Also, it makes sense seeing the emphasis put on accuracy in EFL, and in such contexts, awareness should still be raised with regard to variation and its functions (Kramsch, 2002, p. 75).

On the one hand, *ELF* may be defined as English used when two speakers do not have a common mother tongue (Murphy, 2014, p. 259; Setter & Jenkins, 2005, p. 12), or, in other words, as "a means of intercultural communication among speakers who come from different first languages" (Jenkins, 2011, p. xxii). On the other hand, EIL does not particularly focus on one context (two L2 speakers of English in the case of ELF) to determine a central norm, but "acknowledges different varieties of English and how these are used for both international and intercultural communication" (Low, 2015, pp. 7-11, as cited in Low, 2021, p. 23): several norms are then taken into account because EIL contexts include interaction between non-natives among themselves, but also between a native and a non-native (Low, 2021, p. 29). However, some authors do not seem to agree on the exact difference between EIL and ELF: Jenkins (2004, p. 114, 2011, p. xxii) considers both terms synonyms,<sup>6</sup> while Low (2021, p. 23) clearly makes a distinction. In either case, what is at the core of both approaches is communication with non-natives.

Studies have shown that several elements are necessary intelligibility conditions in NNS communication and should therefore be the main focus of teaching in EIL and ELF contexts; these elements are commonly known as "The Lingua Franca Core" (LFC) (Jenkins, 2002, p. 96). Overall, "most consonant sounds, vowel quantity, initial and medial consonant clusters and tonic stress"<sup>7</sup> (Jenkins, 2004, p. 115; see Jenkins, 2002, p. 96 for further details) form the LFC and accuracy in all these levels should ensure mutual intelligibility in EIL or ELF contexts. The findings were based on analyses of the speech of Jenkins's international students. Outside the core are elements that should not be primarily focused on for they are unlikely to lead to a breakdown in communication; for instance, the consonant sounds / $\theta$ / (as in "nothing") and / $\partial$ / (as in "mother") fall outside the core (Jenkins, 2002, p. 96, 2011, p. xxiii). These non-core elements are also said to make up a speaker's foreign accent (Jenkins, 2002, p. 97). The LFC also means that the functional load principle is not applicable to EIL and ELF contexts (see 3.2.).

If the LFC offers an interesting approach in a context where there are more non-natives than natives, it has downsides such as the logical impossibility to plan with whom a speaker may interact more, meaning NSs or NNSs (Diana, 2010, p. 13) and another disadvantage is the focus the LFC puts on segmentals, while suprasegmentals are argued to be critical in intelligibility (see 2.4.2.) (Low, 2021, p. 25). Still, Jenkins (2011, p. xxiii) reminds critics that her LFC is not a substitute for other more traditional goals, meaning GA and BBC Pronunciation, but constitutes a suitable alternative when interaction takes place among non-

<sup>&</sup>lt;sup>6</sup> Both EIL and ELF will be considered synonyms in this dissertation.

<sup>&</sup>lt;sup>7</sup> *Tonic stress* is a syllable made prominent within a tone unit; a tone unit usually consists of a phrase or clause (Crystal, 2008, p. 487).

natives; on the emphasis put on segmentals, they have been shown to be critical to communication between two NNSs (Setter & Jenkins, 2005, p. 5).

Apart from learning contexts, the general teaching trends have also had and still have an impact on the teaching of pronunciation. With the advent of communicative language teaching between 1970 and 1980, which is said to still affect today's practices (Derwing, 2010, p. 25), pronunciation teaching was viewed as unimportant for mere input exposure would be enough to ensure correct pronunciation (Derwing & Munro, 2009, p. 481; Derwing & Munro, 2014, p. 38; Jenkins, 2004, p. 114) and experience was then the key to success (Thornbury, 2013, p. 211). If pronunciation was addressed at the time, it was still in terms of what natives did (Murphy & Baker, 2015, p. 17).

#### 3.3.2. The Listener

In pronunciation teaching, taking the listener into account is important to set targets and thus has implications on the perception training in class (see 4.5.); the listener should even be considered as important to successful interaction as the speaker themselves (Derwing, 2010, p. 33). The listener's abilities to decode an utterance are in many ways determined by their language background: they may be used to certain segments and intonations and then have difficulties when hearing other unfamiliar sounds or patterns, that is, "phonological 'deafnesses'" (Dupoux & Peperkamp, 2002, p. 2), a perception "filter" (Borrell & Salsignac, 2000, p. 166), or "a phonological barrier" (Borrell, 2000, p. 204). For example, the listeners may have trouble perceiving lexical stress (Dupoux et al., 1997, p. 418), the L1 influences how the L2 is perceived (Iverson & Evans, 2007, p. 2853), and non-native listeners rely more on what they actually hear than context, meaning clear pronunciation is important to them, whereas native listeners may rely more on the context to decode what they hear (Jenkins, 2002, p. 89). The notion of phonological barrier can be extended to many language units, some of which do not belong to phonology (see 4.8.2.): segments (Borrell, 2000, p. 204), suprasegmentals (Dupoux & Peperkamp, 2002, p. 2; Intravaia, 2000, p. 229), gestures, body position in interaction, style and argumentation (Intravaia, 2000, p. 229).

Additionally, non-native listeners find highly proficient NNSs more intelligible than NSs, if they have the same L1, and if they do not have a common L1, they may find the NNSs more or as intelligible as NSs; it works both ways: a native listener will find a native talker more intelligible than a NNS (Bent & Bradlow, 2003, p. 1607). In other words, a French English learner will find any non-native talker more or as intelligible as a native English speaker, the

condition being the high proficiency of the speaker. This is called "'[(mismatched)]<sup>8</sup> interlanguage speech intelligibility benefit' [or] the benefit afforded by a shared interlanguage between a non-native talker and a listener" (p. 1600). This phenomenon may be explained by the listener being "better equipped" (p. 1607) to perceive and decode specific pronunciations. It may even lead to the establishment of new non-native English norms among NNSs (p. 1608) (see 3.2.). Interlanguage speech intelligibility benefit may further be linked to the familiarity with foreign speech and its impact on intelligibility scores (see 2.3.2.).

All this leads Low (2015, p. 130, as cited in Low, 2021, p. 24) to support "listenerdominated norms": the goals to adopt then depend on the listener's origin. This may however pose the same problem as the LFC, that is, one may never know who they may speak to.

#### 3.3.3. The Learner

The learner is not to be forgotten in pronunciation teaching, especially since many of their characteristics influence their performance: their L1, their age, "the educational context" (Rogerson-Revell, 2021, p. 196), among others. The L1 may determine the pronunciation of specific segmentals: it seems common amidst Japanese speakers to mispronounce /l/ and /r/, but does not seem to be the source of any trouble for French speakers (Levis, 2007, p. 188). Besides, some pronunciation textbooks explicitly mention units or aspects to focus on depending on the L1 (e.g., Bowler & Cunningham, 2015; Hancock, 2007).

As far as the age of acquisition is concerned, it is considered "a crucial variable" (Ioup, 2008, p. 57): "late learners are not likely to achieve native-like pronunciation, but also … native-like L2 phonology is normally found only with very early onset and the likelihood diminishes as the onset age increases" (Ioup, 2008, p. 48, see Ioup, 2008 for a review of articles investigating age and pronunciation proficiency). The first years of learning are critical in preventing fossilised forms<sup>9</sup> (Derwing & Munro, 2014, p. 51), which may be a great source of frustration when, for example, adult learners have become highly proficient in vocabulary and have mastered the grammar but still stumble over basic words (Hu et al., 2013, p. 366). As reminded by Lightbown (2008, p. 6), it is necessary to keep in mind the goals targeted in order to determine if age is really so crucial.

<sup>&</sup>lt;sup>8</sup> Mismatched refers the case where the non-natives do no share their L1.

<sup>&</sup>lt;sup>9</sup> *Fossilised forms* are language forms that an L2 learner may have difficulties improving despite teaching; it may even be unlikely that these forms ever improve (Selinker, 1972, as cited in, Derwing & Munro, 2014, p. 35).

There exist other personal characteristics than the age of acquisition or the mother tongue that influence pronunciation proficiency and that are then important to take into account when tackling pronunciation in class (see Gut, 2009, p. 299 for a comprehensive overview). When testing 139 advanced German university students who had started to learn English after they had turned 10 years old, Hu et al. (2013, p. 371) assessed that "phonetic coding ability [PCA], music aptitude, empathy and openness to experience" were factors related to intelligible pronunciation. *PCA* is defined as the ability to recognise and identify speech sounds, associate them with others and with written symbols and to remember these combinations (Carroll, 1962, p. 128). PCA and empathy were actually the only significant predicators of pronunciation ability (Hu et al., 2013, p. 374). Still, musical education has also been found to be advantageous to perceiving stress (Degrave et al., 2011, p. 98) and to better produce and perceive individual sounds (Gottfried, 2007, p. 236).

#### 3.3.4. Differences Between English and French

Authors agree that both English and French are different in terms of phonology (Frost, 2011, p. 67; Vaissière, 2002, p. 16; Walter, 2001, p. 52). To Frost (2011, p. 69), one of the main differences is how stress is used, leading to an empirically attested "deafness" to stress from francophone learners (Dupoux et al., 1997, p. 406). Walter (2001, pp. 52–55) agrees on the difficulties francophone speakers have with stress in general (see 2.4.2. for a definition of stress), as well as with specific segmentals,<sup>10</sup> which has been corroborated by Collins & Mees (2008, p. 211, as cited in Diana, 2010, p. 14). Another main difference, which again relates to stress, is that French is argued to be a syllable-timed language and English is said to be a stress-timed one (Frost, 2011, p. 71; Setter & Jenkins, 2005, p. 8). This particular notion was discussed previously (see 2.4.2.). If these differences are useful in better understanding some difficulties learners may have, it may be relevant to start from there in pronunciation teaching (Couper, 2021, p. 138).

#### 3.3.5. Oral Teacher Input

One may wonder about the role the teacher's input plays in acquiring pronunciation. One study looked at the comprehensibility and accentedness of learners who had received pronunciation classes from either an English native or a non-native teacher (Levis et al., 2016). Eighteen university students with an above-intermediate level of English were split up into two groups: one was taught English pronunciation by a NS of American English and the other by a Turkish

<sup>&</sup>lt;sup>10</sup> The concerned segments are : /i:/, /ɪ/, /əʊ/, /p/, /ɔ:/, /h/, /r/, /p/, /t/, /k/, / $\theta$ /, and / $\delta$ /.

native. The classes lasted over a period of 7 weeks, were given twice a week during 75 minutes and focused on suprasegmentals. Both pre- and posttest were conducted and required listeners to rate the 18 students on their comprehensibility and accentedness. Overall, both groups improved similarly, meaning that there was no significant difference between them (p. 17). In addition, despite having a preference for a native teacher when it came to speaking (p. 22), all students valued their teacher highly (p. 21). These findings led the authors to the conclusion that pronunciation learning depends on factors others than the teacher's L1, just like any other skill learned in class (p. 23), and that it should therefore "bolster the confidence of skilled [non-native English speaker teachers] and [native English speaker teachers] alike" (p. 25).

Ioup & Weinberger (1987, as cited in Rasier, 2011, p. 121) slightly touched upon this point when they noticed that input exposure did not lead to improvement in every sector of pronunciation, meaning that input is not always enough. In their study on the importance of feedback (see 4.6.), Saito & Lyster (2012, p. 625) as well state that teacher input is not sufficient to help learners improve their pronunciation. Also, Iverson et al. (2012, p. 157) have found evidence to support that claim (see 4.5.). Finally, some pronunciation approaches almost seem to do away with oral teacher input, such as The Articulatory Approach (Messum & Young, 2021) and The Silent Way (Richards & Rodgers, 2001b) (see 4.7.).

#### **3.4. Segmentals Versus Suprasegmentals**

If the segmental and the suprasegmental levels make up pronunciation, teachers should know where to put the focus: Should they teach solely segmentals or suprasegmentals? Should they emphasise one over the other?

Yenkimaleki & van Heuven (2021) split 80 Iranian university students into four groups who received specific pronunciation training: all were taught or given explanations on one aspect of pronunciation (segmentals or suprasegmentals) and then exercised either perception or production specifically; the course accounted for 15 hours in total. A control group (without any form of explicit pronunciation training) provided comparison. Intelligibility and comprehensibility were tested before and after the pronunciation course. If all groups improved (p. 6), the segmental-production students scored best in intelligibility and the suprasegmentalperception group scored best in comprehensibility (p. 7).

Derwing et al. (1998) conducted a similar experiment. Forty-eight adult students of intermediate level were separated into three distinct groups: the control group, the segmental-based approach group and the suprasegmental-based approach group (also called the global

group); the course lasted 20 hours over a 20-week period, started with a pretest and ended with a posttest. The participants' pronunciation skills when reading were rated by native listeners for accentedness and for comprehensibility, and the same native listeners rated the students' fluency, accentedness and comprehensibility in spontaneous speech. The results indicated that the two test groups improved in comprehensibility and accentedness when reading, but only the suprasegmentals group improved their comprehensibility and fluency in spontaneous speech (p. 394). Put differently, both groups could successfully transfer their acquired knowledge to reading, but solely the global group transferred their new skills to actual speech (p. 406). Still, the authors do not favour a sole focus on prosody but rather advocate taking both levels into account for, when miscommunication occurs, repeating a word may clearly help (p. 407).

Just like the previous studies, Gordon & Darcy (2016) tested the effect of segmentalversus suprasegmental-based teaching approaches. Thirty high-intermediate students were divided into three groups: the control group, the segmental-based group and the suprasegmental-based group. Each trained pronunciation and received theoretical explanations on their respective focus, but the control group only repeated words and sentences; the segmental group specifically trained four vowels. Pre- and posttests were conducted. Twelve listeners rated the 30 speakers for comprehensibility after they had taken the course which accounted for 3 hours and 45 minutes spread over a 3-week period. Despite all groups having received similar grades on the pretest, they differed in the posttest, pointing to effective training (p. 73). According to the posttests, only the suprasegmental group showed significant improvement in comprehensibility leading the authors to state that "explicit phonetic instruction can benefit L2 learners when it is not restricted to vowel training only" (p. 81). Indeed, the segmental group's comprehensibility even decreased, which may have come as the result of too much attention drawn to four sounds, other important features then being deemphasised (p. 82). Another possible explanation is that more time is to be allocated to segments for their production to significantly improve (p. 83).

From the three above-mentioned studies and others that rather emphasised the positive impact of suprasegmentals on intelligibility and comprehensibility, the logical conclusion to draw would be to strongly emphasise the teaching of suprasegmentals. Yet, Yenkimaleki & van Heuven (2021, p. 10), Derwing et al. (1998, p. 406), Gordon & Darcy (2016, p. 84) advocate spending time learning about both levels in class, which should help keep "a balance between fluency and accuracy" (Gordon & Darcy, 2016, p. 65). This view is shared by other researchers who regret that teachers generally put too much emphasis on segments (Chun, 2002, p. xiv;

Diana, 2010, p. 12; Levis & Grant, 2003, p. 14; Munro & Derwing, 2006, p. 521; Pennington, 2021, p. 16). Eventually, suprasegmentals, unlike segments, are argued to make up "a skill [rather than a body of knowledge]" (Fraser, 2010, p. 10) and are therefore relevant to teach. Therefore, Chun (2002, p. 144) pleads for a mix of a top-down and bottom-up approach (see 3.1.).

#### 3.5. When to Teach

If one knows what goals to pursue and what to teach, another important aspect is knowing when to teach. Pennington (2021, p. 6) deplores the usual integration of a pronunciation component in secondary schools or in universities: pronunciation explanations are given on the spot, as "a response to learner performance" (p. 6), instead of being integrated as one specific component, and if it is ever the case, it is limited to an advanced and highly proficient public. Nevertheless, Marks (2011, p. xxvi), among others (Bowen, 1972, p. 87; Derwing & Munro, 2014, p. 48; Galante & Thomson, 2017, p. 117; Gordon & Darcy, 2016, p. 58; Kissling, 2013, p. 735; Levis, 1999, p. 373; Levis & Grant, 2003, p. 14; Rasier, 2011, pp. 126–127; Taylor, 2006, p. 53; Thomson, 2012, p. 19) does insist on pronunciation exercises catering to a unique need, that is, helping students work on segments or on the suprasegmental level if they have issues communicating in speaking activities.

Pronunciation classes may not be included because of various reasons such as time and number constraints, what is then important is to integrate pronunciation teaching into other activities, like listening and speaking exercises (Derwing, 2010, p. 28). The following steps could serve as a pattern for pronunciation integration: perception exercises (see 4.5.), sound production and speaking (Couper, 2021, p. 139). Likewise Celce-Murcia, Brinton, and Goodwin (1996, as cited in Levis & Grant, 2003, p. 13) propose a comparable five-step model that starts with awareness raising and ends with speaking exercises. The keyword is here contextualising (Bowen, 1972, pp. 83–85) for pronunciation does not have to stand apart but can be explicitly linked to more form-focused lessons. For example, stress could be approached in class in a variety of ways: practising it with regard to language functions like disagreement or contradiction<sup>11</sup> (Hahn, 2004, p. 217), or with regard to vocabulary like numbers (e.g., 13 / $\theta_3$ :'ti:n/ and 30 /' $\theta_3$ ..ti/), countable and uncountable nouns (Foote et al., 2012, p. 19) or compound nouns (e.g., "greenhouse" /'gri:n.haos/ and "green house" /gri:n. 'haos/) (Bowen,

<sup>&</sup>lt;sup>11</sup> Even if Hahn (2004, p. 217) does not explicitly say that stress could be practised with grammar, disagreement or contradiction could involve comparatives; other language functions could easily be paired with particular grammar points.

1972, p. 93). Addressing stress when seeing vocabulary makes much sense to Field (2005, p. 420) because of the very nature of lexical stress. Marks (2011) also insists on ensuring the good pronunciation of any word and suggests presenting rhythm patterns for chunks such as: "whenever you like, whoever you ask, whatever they said" (p. xxvi).

#### **3.6.** Conclusion

After reviewing the literature, it may be suggested that the goals pursued can vary, but that most researchers emphasise intelligibility (Couper, 2021; Diana, 2010) over native proficiency for it represents, to them, an unnecessary ideal (Bowen, 1972; Marks, 2011), but nativeness appears to influence today's teaching practices (Levis, 2005) and teacher training (Detey et al., 2016). Taking the learners' needs into account also matters a great deal (Bowen, 1972; Derwing & Munro, 2014; Galante & Thomson, 2017), for it ultimately comes down to what the learner desires to achieve (Taylor, 2006) (Q<sub>1</sub>, "What goals should teachers pursue in pronunciation teaching?").

In the intelligibility paradigm, using non-native models makes sense and is thus recommended since intelligible and comprehensible NNSs form realistic goals (Kissling, 2013; Murphy, 2014) ( $Q_6$ , "Do non-native accents have a place in class?"): learners are indeed more likely to be as intelligible as these proficient NNSs than to sound native. In either case, it is important not to focus too much on traditional (i.e. British or American) native models, but rather introduce learners to these non-traditional (e.g. non-native) models and highlight their positive features (e.g. being both intelligible and accented) (Jenkins, 2000; Low, 2021).

The learning contexts may however determine objectives ( $Q_{11}$ , "Do teaching contexts impact pronunciation goals?"), which are then different if English is learned as a second, foreign or international language (Jenkins, 2002; Setter & Jenkins, 2005). In EFL contexts, it does make sense to strive to sound native, but it does not in EIL contexts where several norms are acknowledged (Low, 2021). In this latter context, one specific set of pronunciation elements should be mastered and thus forms the goal to achieve, that is, the LFC (Jenkins, 2002). Going even further, some researchers suggest moving away from what the speaker wants and focusing on the listener (Low, 2021), which reveals a bit problematic because one can hardly anticipate who they are going to talk to.

Moreover, the learners' characteristics affect pronunciation learning as well. For instance, speakers from specific language backgrounds may have more difficulties pronouncing some sounds (Levis, 2007) or perceiving stress (Dupoux et al., 1997), likelihood of native
pronunciation depends on the age of acquisition (Ioup, 2008), awareness of sounds, empathy (Hu et al., 2013), and musical education (Degrave et al., 2011; Gottfried, 2007) positively influence pronunciation capacity.

One point teachers should all focus on, no matter the learning contexts and the learners is spending time on both segments and on suprasegmentals (Chun, 2002), even if teaching suprasegmentals have been shown to increase intelligibility more than teaching segments (Derwing et al., 1998; Gordon & Darcy, 2016; Yenkimaleki & van Heuven, 2021). Addressing both segments and suprasegmentals should be done when addressing other language components, such as lexis, grammar or language functions (Bowen, 1972; Field, 2005; Foote et al., 2012; Hahn, 2004; Marks, 2011).

Eventually, teacher input has been demonstrated to be less impactful than teaching ( $Q_{12}$ , "How does teacher input influence the pupils' pronunciation?"): what appears to matter to improve the learners' pronunciation is then not the teachers' L1 (meaning the quality of input pupils get from their teachers), but other factors (Levis et al., 2016).

### 4. Overview of Pronunciation Teaching Methods

#### **4.1. Introduction**

The purpose of the following chapter is to give an overview of techniques discussed in the literature in order to teach pronunciation. There exists a wide array of techniques, but this chapter introduces a few pronunciation teaching methods, so the inventory is not comprehensive. These research questions are answered: "What are effective techniques to teach pronunciation ( $Q_2$ )?", "How does feedback impact pronunciation teaching ( $Q_4$ )?", "Is the use of a metalanguage effective to teach pronunciation ( $Q_5$ )?"

### 4.2. Phonetic Notation

*Phonetic notation* may be defined as "the use of special written symbols to refer to the sounds or sound features of one or several languages" (Mompean & Lintunen, 2015, p. 2). Because of the lack of correspondence between spelling and pronunciation in English (Jones, 2011b, p. vi; Marks, 2011, p. xxvi; van Oostendorp, 2020, p. 20), phonetic notation offers a convenient approach to consistently represent one sound with one unique symbol (Mompean & Lintunen, 2015, p. 4). In other words, phonetic notation is one kind of metalanguage, which is a key component of pronunciation teaching (Couper, 2011, p. 12) that will enable students to express what they hear or think they hear (Fraser, 2010, p. 11). Mastering phonetic notation should then not be "an end in itself", but should be seen as a way of efficiently talking about pronunciation (Mompean & Fouz-González, 2021, p. 164).

The most commonly used metalanguage (Mompean & Fouz-González, 2021, p. 157) is the phonetic symbols from the International Phonetic Alphabet (IPA), devised by the International Phonetic Association around 1887 (Murphy & Baker, 2015, p. 7). The symbols include notation sets for segments, as well as icons to represent the suprasegmental level like "a superscript stress mark diacritic [']" (Mompean & Lintunen, 2015, p. 6).

There exist other means of representing segments (e.g., keywords), and suprasegmental features, (e.g., the ToBI system) (Estebas-Vilaplana, 2017, p. 75; Mompean & Fouz-González, 2021, p. 157), or simply using arrows, circles, capitals or underlining parts of a word can serve as ways to indicate stress (Mompean & Fouz-González, 2021, p. 158; Mompean & Lintunen, 2015, p. 6).

The potential advantages of using phonetic notation are numerous, but the main one seems to be autonomous learning (Marks, 2011, p. xxvii; Mompean & Fouz-González, 2021,

p. 157; Mompean & Lintunen, 2015, p. 6), for the symbols are used outside and inside the classroom: textbooks and dictionaries often make use of IPA or of other sorts of phonetic notations (Mompean & Lintunen, 2015, p. 3). In addition, it may prove useful when working on perception in order to provide learners with crucial adequate feedback (Fouz-González & Mompean, 2021, p. 300; Mompean & Fouz-González, 2021, p. 162) (see 4.6.). One last and general asset of IPA in particular is that it has iconic representative means for every sound of every language (Setter & Jenkins, 2005, p. 2), which then in my opinion offers interesting applications in different language classes.

Nonetheless, phonetic notation has some downsides. When looking up a word in a dictionary, one must bear in mind that the phonetic transcript, if given, is not "how English ought to be pronounced [but] how [the authors] believe some native speakers of English actually do pronounce the word" (Jones, 2011b, p. vi), which is confirmed by Lewis (2011, p. xxiv). This phonetic transcript or "citation form" is further problematic for it is unlikely to be as clear in actual speech (Cauldwell, 2011, p. xx). The Cardinal Vowel quadrilateral that vowels are often represented in, also suffers from the same disadvantage: it merely represents and is then approximate, which is why circles and not points are used (Diana, 2010, p. 16; Jones, 2011b, p. vii). One obvious disadvantage of phonetic notation is that some may consider it to be an extra burden to teach and to learn (Fouz-González & Mompean, 2021, p. 300; Mompean & Lintunen, 2015, p. 3).

Theoretically, phonetic notations are useful and a few studies have been conducted testing the potential benefits of such systems. Fouz-González & Mompean (2021) separated 71 proficient (B1 level) Spanish university students into three groups: a control group, a phonetic symbol group and a keyword group. While the control group did not receive any instruction at all, the two other groups followed a 2-hour High Variability Phonetic Training<sup>12</sup> (see 4.5.) program, spread over 4 weeks, focusing on identification tasks, namely the students heard a word and had to choose the correct symbol or keyword corresponding to the sound (p. 297). The learners trained on eight English vowels that are typically problematic for L1 Spanish speakers. Three perception tests were run: a pre-, post- and delayed posttest. The posttests revealed that both experimental groups performed twice as good (significantly better then) as the control group (p. 311), but that the scores were always higher for the symbol group, with

<sup>&</sup>lt;sup>12</sup> *High Variability Phonetic Training* (HVPT) is a teaching method to train perception, typically of segments; the learners have to listen to a lot of input from different speakers; an often-used task in HVPT is a discrimination activity, meaning identifying whether two items belong to a same category (Fouz-González & Mompean, 2021, p. 299; Thomson, 2018, p. 209).

the exception of one vowel  $/\Lambda/$  (as in "cup") (p. 316). Important conclusions of that study are then that phonetic symbols and keywords alike are valuable tools to help improve perception (p. 318) even with little training (p. 323), and that learners seem to better like symbols for they see them as "more beneficial and engaging" (p. 320).

As far as the learners' opinions on phonetic notation are concerned, Mompean & Lintunen (2015) surveyed them with a questionnaire given to 177 advanced (B2-C1) university learners from three different countries (52 from Finland, 59 from France and 66 from Spain). Noteworthy is the fact that they had received classes on a phonetic notation which is said to be "based on the International Phonetic Alphabet" (p. 11). If they generally all agreed on the potential for "awareness-raising" and the visual help that notation provides (pp. 14-15), they reacted more negatively towards the advantage of "autonomous learning", although they remained quite certain that it indeed was one of the advantages (p. 16). Despite finding it relatively simple to use (p. 19), few of them claimed to actively use phonetic notation when confronted with a new word (p. 17). Finally, a majority agreed that it would be too hard to learn for pupils under 12, but that it would be easy for those above 12 (p. 19). The main divergence between the three countries turned out to be where the participants had become acquainted with phonetic notation: a majority of Finnish said it was before university, while a large number of French and Spanish participants noted that phonetic notation was paid attention to in secondary schools (p. 20). Overall, students seem to acknowledge the potential benefits of phonetic notation for teenagers, in spite of not using it often themselves.

## 4.3. Technology

In the literature, there seems to be a broad consensus (Levis, 2007; Rogerson-Revell, 2021) that technology, or in this case computer assisted pronunciation teaching (CAPT) may be beneficial. Technology is versatile in the sense that it may provide learners with a freedom that is almost impossible to achieve in classroom environments (Busà, 2007, p. 165): with technology, pronunciation learning becomes individualised and this could then make the learner autonomous (Levis, 2007, p. 184; Rogerson-Revell, 2021, p. 190; Venkatagiri & Levis, 2007, p. 276). Technology may even offer students friendlier environments where they could express themselves free of stress (Rogerson-Revell, 2021, p. 190), like in the "virtual world" Second Life (Derwing, 2010, p. 30). Also, technology offers learners awareness-raising devices in the form of personal feedback, that may well be visual above all things (Chun, 2002, p. 97; Jilka, 2007, p. 94; Levis, 2007, p. 184; Mehlhorn, 2007, p. 224; Rogerson-Revell, 2021, p. 196; Thomson, 2012, p. 20). Moreover, Levis (2007, p. 188) posits the idea that CAPT should ensure

pupils may pick whatever element they need to practise on and that feedback should assist them in doing so.

Because of their many interests, authors name specific programs teachers could use as support in any pronunciation course, like PRAAT, WASP, CSL (Levis, 2007, p. 196), Moo-O, Hooked on Phonics, Endless ABC, Twinkl, Khan Kids (Low, 2021, p. 30). In keeping with recent research and trends on the use of non-native models, Low (2021, p. 30) mentions Well Said! as a good application involving NNSs. Talking heads, that is, visual computerised representations of a human head with precise inside and outside features of the mouth, may also help in pronunciation teaching (Badin et al., 2010, p. 502).

Studies have shown that technology could prove efficient in pronunciation improvement. Busà, (2007) ran a "pseudo"-experiment with 30 of her own students who, for 20 hours in the language laboratory, recorded themselves, got visual feedback and tried to make their own waveform resemble a native's. If the students ended being able to modify their waveform, there was not any pre- or posttest to assess and quantify any effects, making this experiment a first step in the studies regarding technology.

Thomson (2011) conducted a study on 22 adult Mandarin beginner learners of English. In eight 15-to-20-minute autonomous classes spread over 3 weeks, the learners trained their perception of 10 English vowels in two contexts: "/bV/ [and] /pV/" (p. 751). During the sessions, through a computer program, the learners had to associate sounds with "nautical flag[s]" (p. 752), which were chosen over words "to avoid any confounding effects from potentially past associations between sounds and English orthography" (p. 752). Recorded in pre- and posttests, the learners' pronunciation was scored on intelligibility by four natives in familiar (/bV/, /pV/) and new contexts ("/gV/ and /kV/ and /zV/ and /sV/") (p. 752). The results show that improvement had taken place, despite not having been taught to produce, and that the acquired knowledge transferred to some new contexts, namely /zV/ and /sV/ (p. 758). One important drawback of the study is that the training and test stimuli consisted of non-words, meaning that transfer to actual words and to spontaneous production remains to be tested (pp. 759-760).

Fouz-González (2020) tested a commercially-available app, English File Pronunciation, with 52 proficient (B2 level) Spanish university students; four vowel sounds and two consonant sounds were chosen as the targets of training because of their tendency to be problematic for Spanish speakers. Learning was completely autonomous, with the only condition being a daily

15-to-20-minute practice. The researcher used a control group, a pretest and a posttest. The researcher tested both perception and production, in imitation, controlled and spontaneous situations, while the application itself revolves only around recognition of segmental sounds and thus around identification tasks. After a 2-week training, the learners had improved their perception and production of both known and new words, but there were differences in terms of sounds and tasks (p. 73). If the results are said to be "encouraging" (p. 74), it comes at a literal cost: 5.49 euros per application, which, according to a majority of the participants is a deterrent (p. 76).

On that note, Thomson (2012) reviewed an IPA-related, perception-based, freelyavailable play pronunciation website: English Accent Coach. Being similar to English File Pronunciation in its approach (i.e., identification of sounds in syllables and in words using phonetic notation), it may offer a suitable alternative, which does lead to improvements in both perception and pronunciation (p. 23).

If the previously-mentioned studies deal with segments, Levis & Pickering (2004, pp. 517–520) argue that it may be possible to practise changing topics in continuous discourse through pitch visualisation because these changes are marked by pitch (p. 510). Predicting where stress falls in a transcription, followed by visualising pitch is one idea put forward by both researchers.

Derwing & Munro (2009) rejoice at what technology has to offer but warn against some possible problems: "Technology offers a great deal of promise, provided that technological applications are guided by pedagogical specialists who understand appropriate goals and priorities in teaching pronunciation" (p. 487). Two authors (Levis, 2007, p. 185; Rogerson-Revell, 2021, pp. 190–192) denounce this. To them, much CAPT material seems to do away with today's pedagogy and simply showcases technological advances: mere mimicry or drill-based exercises are often central to many pronunciation apps. As underlined by Low, (2021, p. 30), a necessary first step to avoid such resources or any inefficient resource is for teachers to check the efficiency and accessibility of the apps themselves first, before using them in class, or before advising pupils on pronunciation applications. In either case, teachers should lead their learners to pedagogically-accurate resources ( Derwing & Munro, 2014, p. 51). Detey et al. (2016, p. 16) go a step further by reminding teachers that technology may be useful only if it helps learners meet specific needs. This is eventually doable assuming that teachers do not lack "training in pronunciation and in the use of technology", which may sometimes be the case (Levis, 2007, p. 185).

Furthermore, many applications dedicated to pronunciation do not align with what research advocates. Despite research clearly encouraging putting more emphasis on the suprasegmental level, the focus in apps lies solely on segmentals, which then fosters the overemphasis on vowel and consonant sounds (Pennington, 2021, p. 3; Rogerson-Revell, 2021, p. 192). Besides, native-pronunciation is often promoted in CAPT, regardless of researchers' calls for a shift to the intelligibility paradigm (Rogerson-Revell, 2021, p. 192).

At the heart of CAPT stands Automatic Speech Recognition (ASR), software that recognises speech and makes feedback on pronunciation possible. Unluckily, ASR appears to be faulty for it may lack accuracy (Levis, 2007, p. 185) and may misrecognise adequate pronunciations as errors and vice versa (Rogerson-Revell, 2021, p. 194); this inability of ASR to precisely identify mistakes may lead to more general, and consequently less helpful, comments (Levis, 2007, p. 193). To illustrate these points, Rogerson-Revell (2021, p. 195) highlights the fact that an "unnaturally [slow] and [careful]" diction performs best at Australia's computerised immigration oral fluency test.

Derwing et al. (2000) put a speech recognition software to the test: Dragon Naturally Speaking Preferred (1997). They had 30 highly-proficient speakers record a set of 60 sentences each; out of the 30 speakers, 20 were non-native (Cantonese and Spanish speakers), and 10 were native Canadian English. The ASR software transcribed the sentences, and so did 41 native Canadians, who also rated the sentences for both accentedness and comprehensibility; thanks to the transcriptions, the speakers also received an intelligibility score. If both the human listeners and the computer could transcribe NSs equally well (99.70% of the time for the humans and 90.25% for Dragon Naturally Speaking Preferred), they very much differed in their skills when they had to transcribe accented speech, despite the NNSs having been identified as mildly accented and fairly easy to understand:<sup>13</sup> the transcription scores for the humans were 94.99% and 95.71% for the Cantonese and Spanish speakers respectively, while the ASR software scored 72.45% and 70.75%, which means that "more than one of every four words [was transcribed] inaccurately" (p. 600). This leads the researchers to state that ASR could help students see what they may not pronounce correctly, or provide "negative feedback<sup>14</sup> in a

<sup>&</sup>lt;sup>13</sup> The accent scores were set on a 9-point scale: the Cantonese speakers were rated 4.53 and the Spanish speakers were rated 4.64. Comprehensibility was rated on a 9-point scale as well: the Cantonese speakers were rated 3.53 and the Spanish speakers were rated 3.35 (Derwing et al., 2000, p. 597).

<sup>&</sup>lt;sup>14</sup> Negative feedback or negative evidence is the opposite of positive feedback. *Negative feedback* shows that something is missing or did not occur (Kosko, 2017, para. 1). In the case of ASR, negative feedback shows that a correct sound was not pronounced.

nonthreatening context" (p. 601). However, seeing when the study was conducted (i.e., 2000), one may expect that technology has improved a lot.

Although feedback is fairly important in CAPT material, the way it is provided is even more important. Visual representations are common amidst pronunciation resources, but these are often "spectrograms, waveforms and pitch tracings" (Levis, 2007, p. 190) and they are said to be too difficult for learners to fully make use of because they lack transparency and may require formal training (Busà, 2007, p. 175; Detey et al., 2016, p. 26; Fouz-González, 2020, p. 63; Levis, 2007, p. 190; Rogerson-Revell, 2021, p. 193). Visual representations are still worthwhile to some and are incorporated into programs like PRAAT and WASP (Levis, 2007, p. 191), but others (Derwing, 2010, p. 30; Setter & Jenkins, 2005, p. 10) caution teachers against using the last two resources for they were first made in order to thoroughly research speech; yet, the WinPitch program is available in several versions, with one being specific for research and two others being specially conceived for teachers and learners. On the other hand, comparison between two visualised productions to find pitch contours, as well as the pitch levels may reveal fairly easy nonetheless (Busà, 2007, p. 175).

# 4.4. Drama

"Roleplay, imitation [,] communicative 'mirroring' [,]<sup>15</sup> and imaginary enactment or dramatization [sic] of different characters and voices" are valuable activities which all involve adopting a certain style of pronunciation (Pennington, 2019, p. 378). To Pennington (2019, p. 377), they are valuable because they allow learners to familiarise themselves with specific pronunciations, which is a necessary first step to acquisition. The links between such activities and the benefits of exposure to various models, as explained by researchers (see 3.2.) are also quite apparent.

Evidence of advantages of drama is first found in internalised voices. When analysing the speech of an adult L2 English speaker, Moreno (2016, p. 39) showed that the suprasegmentals were altered and that they better fitted the speaker's local norm when the L2 speaker enacted someone else's voice; for example, when telling a story, the speaker would quote what someone had said and change his voice accordingly, resulting in better pronunciation. Despite the impossibility of generalisations based on one case study only,

<sup>&</sup>lt;sup>15</sup> Mirroring involves copying a speaker, that is their pronunciation, along with their non-verbal features (see 4.4.1.).

Moreno's finding is a first step in understanding how playing a role may help pronunciation teaching (p. 40).

LaScotte & Tarone (2019) obtained similar results. Ten people from various English levels and from various language backgrounds took part in intensive English lessons. When the authors examined their speech, they found that they were more grammatically accurate (p. 109) and generally more fluent (pp. 104-105) when they enacted others' voices, if the imitated people were proficient speakers. This leads the authors to recommend involving students in "classroom theatrics, [or] roleplays, acting out student-selected scenes from a film, show, novel or play ..." in order to get a better view of what pupils may actually be capable of (p. 108). A last possible advantage of such activities the authors mention is better pronunciation (p. 108).

As far as actual drama is concerned, it may potentially be stress-relieving, as investigated by Galante (2018). If her study is not directly linked to pronunciation teaching, she does emphasise that stress may weaken the learner's speaking abilities and that drama does impact pronunciation (p. 282). She split up 24 B1-level Brazilian teenagers into two groups: a drama and a non-drama group. For 4 months, twice a week, the teenagers followed a 2-hour class with different goals, either the enactment of a 15-minute play (drama group) or a personal presentation (non-drama group). At the end of the two programmes, the anxiety level, or Foreign Language Anxiety, had reduced in both groups, when compared to their anxiety levels at the beginning of the course. Nevertheless, the difference was slightly bigger for the drama group (p. 273). In addition, the drama-group learners felt more comfortable when speaking English, which may be explained by the new identity speakers had to adopt (p. 280) (see 2.3.3. for more on identity). Galante mentions that drama, apart from enhancing comfort when speaking a foreign language, does bring "added values [or] particular [advantages] [relating] to communication skills such as accuracy, eye-contact, and body language" (p. 282). Pronunciation accuracy can then be worked on through drama.

Galante & Thomson (2017) investigated drama more thoroughly and also focused on enhanced fluency (p. 116). Twenty-four B1-level Brazilian teenagers were split up into four different groups; the two experimental groups included five and eight pupils respectively, the two control groups included five and six learners respectively. Similarly to Galante's study (2018), the experimental groups (drama groups) trained to perform a 15-minute play and the control groups (non-drama groups) to give an oral presentation. Every group was given exercises on segmental and suprasegmental features, but only the drama groups took on roleplaying activities, such as giving advice in specific situations. Different types of pre- and posttests were conducted on all pupils; these demanded the pupils to describe a story in the first or third person, to retell a story after having watched a video, inventing a monologue and roleplaying with the teacher. Based on these tests, 30 Canadians rated speech samples to assess the speakers' comprehensibility, accentedness, and fluency. If all learners were considered comprehensible even though they were also said to have strong accents (p. 133), the results showed that fluency and comprehensibility only significantly improved within the dramagroups (pp. 128, 130), but the improvement in comprehensibility was little (p. 130). The drama training did not seem to have any impact on accentedness (pp. 130-131). Interestingly, the scores were dependent on the type of task: when they told a picture-based story in the first person, the learners were the most comprehensible and the least accented (p. 134).

### 4.4.1. Shadowing and Mirroring

Shadowing and mirroring are two techniques that resemble drama and other roleplaying activities. *Shadowing*, or sometimes called "echoing" (Derwing & Munro, 2014, p. 50) is a repetition exercise where learners echo or repeat what a model is saying; *mirroring* on the other hand involves sound reproduction as well as copying non-verbal features (Derwing & Munro, 2014, p. 50; LaScotte et al., 2021, p. 150; Luo et al., 2008, p. 2807).

Mirroring activities ending with a presentation in the style of a chosen model has been shown to have benefits: with that approach, seven highly proficient adult learners significantly improved their intelligibility (LaScotte & Tarone, 2022, pp. 753–754), managed to sound more native (p. 754), and increased their self-confidence (p. 758). Such projects support exposure to numerous speakers as (proficient) learners seem to be able to copy ways of speaking and may then improve their intelligibility. As far as shadowing is concerned, two studies demonstrated its efficiency in terms of improved imitation abilities, fluency, accentedness, comprehensibility in spontaneous speech (Foote & McDonough, 2017, p. 52), and in terms of accentedness, intonation, stress placement, and segment production in reading and spontaneous speech (Martinsen et al., 2017, as cited in LaScotte & Tarone, 2022, p. 748).

#### **4.5.** Perception

*Perception* is defined as "an internal mental (and physiological) process by which the perceiver recognizes [sic] incoming stimulus events as instances of mental categories" (Strange & Shafer, 2008, p. 159). It is argued to be the "underlying foundation of pronunciation" (Thomson, 2012, p. 19), the tenet being that perceiving comes before producing and that some pronunciation issues hence find their source in perception (Detey et al., 2016, p. 21; Mehlhorn, 2007, p. 218;

Setter & Jenkins, 2005, p. 6), which may be itself impeded by a phonological barrier (see 3.3.2.). An example of perception helping production is how children who had been exposed to English passively (i.e., through songs or television, without any explicit instruction) started producing contrasts that did not exist in their mother tongue (Simon et al., 2014, p. 18, 2016, p. 740): new phonological categories were forming. Pronunciation problems may be further complicated by the separation and sometimes chasm between "what we say, the physical sounds we produce, and what we think we say" (Couper, 2014, p. 37). Working on perception then only seems logical, especially with regard to the essential (VanPatten, 2002, p. 108) "noticing hypothesis" (Schmidt, 2001, p. 26), which states that learners first need to notice or have their attention drawn to a particular feature of the language before acquiring it (Schmidt, 2001, pp. 29–31; Schmidt, 1990, p. 149). In the literature, perceiving is then pointed to as the necessary condition to adapt and change one's pronunciation (Couper, 2021, p. 130; Derwing & Munro, 2014; Setter & Jenkins, 2005, p. 6; Thomson, 2012, p. 20), and perception exercises, such as marking a stressed syllable or comparing one's own recorded production with a native's (Couper, 2003, p. 59, 2011, p. 11), are advocated in classroom contexts (Chun, 2002, p. 150). Eventually, working on perception may not only serve the speaker, but the listener as well (see 3.3.2.), since perceiving helps, for example, decipher the emotion behind an utterance (Chun, 2002, p. 13).

As noted by Henderson (2015), perception goes hand in hand with a switch of attention, "from the language's features to [the] learner's conceptions" (p. 16). When she asked 57 French university students to identify whether a speaker was British or American and to justify their choice, Henderson noticed a great disparity in the answers given (p. 15). So did she when she asked 56 of those students to explain how they would sound more British or American. Besides, some of the answers were plainly wrong (p. 15). Working on correcting these mistakes, to then improve production and basing feedback on what students perceive or think they do is advocated by the author (p. 16).

In the literature, training perception often takes the form of High Variability Phonetic Training (HVPT), which typically focuses on the segmental level, and which requires the learner to listen to large amounts of stimuli produced by varied and numerous speakers. A typical HVPT task is a discrimination exercise where the learner listens to several language forms and must discriminate, or choose, whether they correspond to the same category (Fouz-González & Mompean, 2021, p. 299; Thomson, 2011, p. 749, 2018, p. 209). Although authors do not clearly name HVPT, they do insist on making students listen to diverse speakers and

several varieties (Derwing, 2010, p. 28; Jenkins, 2000, p. 91; Low, 2021, p. 28; Marks, 2011, p. xxvi; Murphy, 2014, pp. 258, 266).

Iverson et al. (2012) tested how efficient HVPT was. They separated native adult French speakers into two test groups: one included 15 experienced subjects who had lived in an English-speaking country from 6 months to 10 years and whose comprehension of English grammar was correct, the other included 21 inexperienced subjects, who had all but one never lived in an English-speaking country and who had mastered the basics of English. The treatment consisted of eight 45-minute sessions spread over the course of 1 or 2 weeks; during the HVPT program, the learners trained 14 vowel sounds through identification exercises. After analysis, Iverson et al. (2012, p. 151) concluded that both groups had improved thanks to the training sessions in the three kinds of tasks that were tested: identification, discrimination and production. What is more, the groups were tested on new words, meaning that the subjects were able to generalise what they had previously learnt (p. 147). Because of the limited nature of the exposure to English during the study, which is far less than what the experienced subjects would be exposed to in their daily lives, the authors suggest that oral input is not sufficient to improve production but that drawing attention to specific sounds "improves L2 vowel perception" (p. 157).

Other numerous studies have been done on HVPT. In a meta-analysis, Thomson (2018) analysed and criticised 32 studies focusing on HVPT. His conclusions are fairly clear: "the studies surveyed provide compelling evidence that HVPT is a very effective pronunciation training tool, and that resulting improvements are long lasting" (p. 208); in other words, exposure to various input from different speakers helps learners forge robust phonological categories. Among that evidence, the author notes the following benefits: it is suited for adults (p. 215), acquired knowledge is generalizable to new words (p. 215), significant improvement of both perception and production occurs despite the latter being smaller than the former (p. 217), improvements last at least 4 months (p. 217). Thirty out of the 32 studies analysed tested HVPT with adult learners, only two tested this teaching process with younger learners, namely 12- and 16-year-olds. Thomson notes that only the study conducted on the 12-year-old children did not result in enhanced production or perception, but he highlights some flaws that may explain the failure and thus remains enthusiastic about the efficacy of HVPT with learners of all ages (p. 214).

Finally, perception is undoubtedly linked to the helpful concept of phonological awareness that is defined as "conscious knowledge of the sounds, syllable structure,

phonotactics and prosody of [a] target language" (Venkatagiri & Levis, 2007, p. 265). When Venkatagiri & Levis (2007) measured 17 university students' phonological awareness level and asked 12 NSs of English to measure the students' comprehensibility in both controlled and spontaneous speech, they concluded that the more phonologically aware they were, the more comprehensible they were (p. 273). They however state that phonological awareness could not be taken as the sole explanation for comprehensibility (p. 275). Since perception is recognition of input, explicit theoretical knowledge on production and properties of segments and suprasegmentals (i.e., phonological awareness) may reveal useful in teaching pronunciation (see 2.4.); this claim might also be supported by the impact of PCA in pronunciation (see 3.3.3.).

#### 4.6. Feedback

The importance of feedback has been slightly touched upon: it is a central matter to technology among other fields (Levis, 2007, p. 194; Rogerson-Revell, 2021, p. 194) (see 4.3.) and is made possible by phonetic notations (Fouz-González & Mompean, 2021, p. 162) (see 4.2.). In general, researchers agree on the importance of feedback to raise learners' awareness of the function of pronunciation (Couper, 2021, p. 137), or to simply indicate how close they are to the norm or goal pursued (Rasier, 2011, pp. 134–135). The question might then be whether feedback is truly efficient and if so, what kind of feedback.

Saito & Lyster (2012) tested the effectiveness of one specific type of feedback: recasts. *Recasts* are defined as repeating the correct form of a mistake (Lightbown & Spada, 2013, p. 140), as in the following example: "Look at the field, I can see 20 /ʃip/!" "Oh yes, I can see 20 /ʃi:p/." Saito & Lyster (2012) studied the pronunciation of the phoneme / $\mu$ /<sup>16</sup> (as in "race") among Japanese adult learners after they had received instruction with or without feedback. Sixty-five Japanese adults were split into three different groups: 29 received feedback during form-focused instruction (i.e., exercises that focused specifically on the perception and production of the consonant sound), 25 followed form-focused instruction solely and 11 students served as the control group. Pre- and posttests were conducted and aimed to get acoustic measurements of their pronunciation of the phoneme / $\mu$ / in spontaneous speech and in controlled contexts. If the authors cannot ascertain that the groups could generalise their instruction to new words, the pronunciation significantly improved exclusively in the form-focused instruction and feedback group (p. 624). Therefore, the authors suggest that both

<sup>&</sup>lt;sup>16</sup> This symbol /1/ may not be found in the symbols-used section in some dictionaries like the *Cambridge English Pronouncing Dictionary* (Jones, 2011a); it is replaced by /r/.

positive (in the form of input) and direct negative (in the form of feedback) evidence is necessary for pronunciation improvement. For the learner, the feedback would then serve as "a double-check [and] [revision of] their own output" (p. 625). Yet, the two authors indicate that further research is needed on the type of feedback that would be most efficient (p. 627).

## 4.7. Physical Skills and the Silent Way

Pronunciation involves the physical production of sounds (see 2.2.); thus, learning how to pronounce may be compared to the process of learning a physical activity (Messum & Young, 2021, p. 172) and that is why Messum & Young (2021) support an "Articulatory Approach, based on a motor skill coaching paradigm" (p. 169). The Articulatory Approach is based on the need of explicit knowledge of how a sound is physiologically produced to guarantee correct pronunciation (Billières et al., 2013a) A first step to that approach may be a short "resensitisation" session (Messum & Young, 2021, p. 171) that enables to think about the invisible, automatic and unconscious movements needed to produce sounds; that session may include blowing air through the mouth to feel how the abdominal muscles contract (PronSci -Pronunciation Science Ltd, 2017).<sup>17</sup> Then, come awareness raising activities such as using whispers and stage (or loud) whispers to better perceive how stress is physically produced (PronSci - Pronunciation Science Ltd, 2017),<sup>18</sup> or stammering to better understand the production of schwa (Messum & Young, 2021, p. 172); others talk of muscle gymnastics whereby the jaw, the lips, the tongue, with the help of mirrors are focused on in the production of isolated segments (Billières et al., 2013a). Imitation, supported by computer imagery of the tongue position, is possible as well (Messum & Young, 2021, p. 173). Talking heads may in my view reveal useful in that respect (see 4.3.). What Messum and Young (2021) warn against is "modelling the results [that] will immediately induce attempts to match-to-target" (p. 177): the teacher should rather "coach" (p. 175) and focus on the physical skills involved.

Taking this coaching principle further, Messum & Young (2021) state that "the belief that students need an aural model to work from is incorrect" (p. 175). In other words, teachers should not, in their view, provide a model to imitate, which is strongly reminiscent of another pronunciation teaching technique: The Silent Way (Gattegno, 1972, as cited in Richards & Rodgers, 2001, p. 81). Gestures of the mouth to show sound formation, clapping to indicate stress, quickly moving the hand to ask a pupil to speed their pronunciation up, or putting one

<sup>&</sup>lt;sup>17</sup> The time code for the video is: 4:32.

<sup>&</sup>lt;sup>18</sup> The time code for the video is 8:50.

finger up to indicate a mistake in the first word of sentence are key elements of The Silent Way (silentway.online, 2021). Often, a chart full of coloured rectangles is used: these represent English sounds, that can then be matched, through pointing, with sounds the pupils are familiar with (i.e., from their L1) (silentway.online, 2021). For instance, the teacher can write the letter "a" on the board, make student repeat how the vowel is pronounced, point to a rectangle, nod to show both represent the same sound, and eventually make the whole class repeat. Once several sounds have been introduced and learned, the teacher can point to a sequence of coloured rectangles to form words. Summarised, the technique revolves around the teacher remaining silent and the learner speaking as often as possible (Howells et al., n.d.; Richards & Rodgers, 2001b, p. 81).

### 4.8. The Verbo-Tonal Method of Phonetic Correction

The Verbo-Tonal Method (VTM) of phonetic correction aims at improving learners' pronunciation. The method is described in the following subsections, starting with its origins, its principles and finishing with concrete exercises within the VTM perspective.

### 4.8.1. Origins and the Structuro-Global Audio-Visual Method

The Verbo-Tonal Method was developed in the 1950s by linguist Petar Guiberina who focused on providing speech stimulus (hence verbo) to assess perception at various frequencies (hence tonal) (Billières et al., 2013b). VTM is to be integrated into another, larger teaching method, namely the Structuro-Global Audio-Visual (SGAV) methodology,<sup>19</sup> because of the goals and guidelines both methods share, that is, oral communication, the importance of perception, and emphasis on real communication situations (Renard, 2000b, p. 5). Successful communication entails appropriate phonetics and pronunciation, so that is why they should be taught (Renard, 2000b, p. 6).

In VTM, there are not any pronunciation-dedicated moments in class (Renard, 2000c, p. 15), but rather pronunciation is integrated into communication activities, since it serves communication (De Vriendt, 2000a, p. 250; Renard, 2000c, p. 18). Still, the teacher should avoid pinpointing and correcting any mispronunciation in spontaneous speech acts (e.g., when a learner wants to share their opinion spontaneously), but should preferably do it when the class

<sup>&</sup>lt;sup>19</sup> The Structuro-Global Audio-Visual method (SGAV) encompasses exercises such as imitation of dialogues supported by clear pictures and illustrations. The emphasis is put on real communication situations and on orality, the written aspect of language coming much later in the process of language learning (Simons, 2021, pp. 15–16), 4 to 6 weeks after the beginning (De Vriendt, 2000a, p. 249); in that sense, SGAV is really "global" (Rivenc, 2000, p. 30).

is repeating a dialogue for example (De Vriendt, 2000a, p. 251). Eventually, the target pursued with VTM is an "all-purpose" accent that should resemble a native's (Billières, 2000, p. 39).

### 4.8.2. Perception, Intonation and the Body

The Verbo-Tonal Method puts emphasis on three elements: perception, or teaching learners to listen (Renard, 2000b, p. 7, 2000c, p. 12), intonation (Billières et al., 2013b; Billières, 2014b; De Vriendt, 2000a, p. 254), and the role of the body in the process of perceiving and producing speech (Billières et al., 2013b).

Perception is considered crucial because it is assumed that production errors first arise because of perception errors. Foreign language speakers have, prior to their L2 learning, formed perception habits that stem from their mother tongue (Intravaia, 2000, p. 218; Renard, 2000c, p. 14): a phonological barrier has formed (Borrell, 2000, p. 204) which then acts as a "filter" in perceiving L2 sounds (Borrell & Salsignac, 2000, p. 166). This involves perception of specific sounds which may not exist in the L1, or sounds that do not exist in certain positions in the L1, but do in the L2 (see Borrell, 2000, pp. 205–207); for example, the phoneme /u/ will be mostly recognised as /u/ by French speakers when its frequency is between 150 and 300 Hertz, while English speakers will recognise it when its frequency is between 300 and 600 Hertz (Cureau, Vuletic, 1976, as cited in, Billières et al., 2013b). Each sound has then its own perceptive characteristics (Billières et al., 2013b). Intravaia (2000, p. 229) extends that concept to other fields within and beyond phonology and phonetics: melody and rhythm, physical movement, spatial distribution, stylistics, dialogues. Since the filter is here quite broad, the author insists on having a global approach (p. 233). Lastly, many other researchers who do not appear to apply VTM also emphasise both perception (Detey et al., 2016, p. 21; Mehlhorn, 2007, p. 218; Setter & Jenkins, 2005, p. 6; Thomson, 2012, p. 19) and the impact of the L1 (Dupoux et al., 1997, p. 418; Dupoux & Peperkamp, 2002, p. 2; Iverson & Evans, 2007, p. 2853; Levis, 2007, p. 188) (see 3.3.2., 3.3.3. and 4.5.).

Intonation is given priority with VTM, but this does not entail that segments are not paid any form of attention - on the contrary (Billières, 2014b; Borrell & Salsignac, 2000, p. 173). The reason why intonation is considered as more class-attention worthy is because it connects segments together and organises them through rhythm (Billières et al., 2013b); *rhythm* is defined in VTM as the height, intensity, and length differences across speech (Billières, 2000, p. 50), where repetition of a certain unit (e.g., word, stress, pause) is key (Roberge, 2000a, p. 124). Additionally, intonation carries meaning (Billières, 2000, p. 42), but also renders speech intelligible (Borrell & Salsignac, 2000, p. 165) and less accented (Billières, 2000, p. 42); VTM is thus in line with what other researchers advocate on the influence of intonation in regard to intelligibility (Field, 2005, p. 410; Hahn, 2004, p. 210) and to accentedness (Jilka, 2007, p. 91) (see 2.3.2. and 2.3.3.). If several authors (Chun, 2002, pp. 77–78; Gussenhoven, 2002) mention three codes or four functions of intonation (see 2.4.2.), VTM makes a distinction between neutral and affective intonation, the former referring to a neutral intonation, without any particular meaning (e.g., a plain statement), and the latter referring to the expression of emotions (Roberge, 2000b, pp. 134–135). Yet, others state that such a thing as a neutral intonation does not exist (Billières et al., 2013b).

The last primary guideline is working on the body, for it is quite involved in speech production; two levels are distinguished: the micro level comprises the speech productive organs, such as the vocal folds, and the macro level comprises the whole body, as the heart beating faster when speaking, or the breathing rhythm changing (Billières, 2000, pp. 43–44; Billières et al., 2013b). The body may help in sound, rhythm or intonation realisation, as relaxation eases speech production and perception (Billières et al., 2013b). Researchers stress how bodily gestures may convey contradiction, or help make matters clearer (Billières, 2000, p. 45; Pavelin, 2000, p. 72). Why VTM works on the body has to do with the communicating goal, which involves interaction, leading to greater "body investment" (Intravaia, 2000, p. 222).

#### 4.8.3. Concrete Applications of the Verbo-Tonal Method

Looking at the principles of VTM and of SGAV, it is not surprising that researchers advocate orality exclusively: Total Physical Response exercises,<sup>20</sup> illustrated dialogues (with pictures and recordings), listening comprehensions, question and answer exercises, model imitation (mirroring and shadowing), constantly speaking the L2 (De Vriendt, 2000a, pp. 248–249), drawing attention to acoustic cues (Borrell & Salsignac, 2000, p. 172) are activities they recommend.

Intonation and rhythm are extremely important in VTM and researchers seem to agree on the benefits of counting-out rhymes in that regard (Krnic-Wambach & Wambach, 2000, p. 94; Roberge, 2000a, pp. 114–118, 2000b, p. 142; Wlomainck, 2000, p. 159). Practising counting them is argued to help learners cope with rhythm; clapping or walking to the rhythm of a dialogue is also possible (Wlomainck, 2000, p. 156). Intonation in general can be addressed

<sup>&</sup>lt;sup>20</sup> Total Physical Response exercises entail learners following precise actions, so they use their body totally (De Vriendt, 2000a, p. 248).

with the body, through using hand movements to represent falling or rising pitches for instance (Billières, 2000, p. 56). Uttering a sentence with every intonation one can think of is another exercise typical of verbo-tonal practitioners (Wlomainck, 2000, p. 158). If a learner struggles with lexical stress, the teacher may ask them to first repeat the stressed syllable ("grand" as in "grandfather") and then gradually add segments ("grandf", "grandfa", "grandfath", "grandfather"), and the learner should repeat each step (De Vriendt, 2000b, p. 263).

To get back in touch with the body component of pronunciation, authors make a number of suggestions: pronouncing vowel sounds while standing up, sitting down or laying down to perceive the breathing process more accurately (Krnic-Wambach & Wambach, 2000, pp. 98–99). Detailed pictures (e.g., a man saying it is hot while painfully climbing stairs) may help learners become aware of the influence of the body in certain situations (Roberge, 2000b, pp. 146–147).

A big point VTM makes is the different contexts segments appear in and how they might be influenced by these. Authors speak of combined phonetics, nuanced (Renard, 2000c, pp. 16– 17) or inverted pronunciation (Billières, 2014a). De Vriendt (2000b, pp. 260–266) provides clear cases where the teacher works on the segment environment to enhance its pronunciation.

In some contexts,<sup>21</sup> the phoneme /ŋ/ can be pronounced as is ("strong") or has to be pronounced with an extra phoneme /g/ ("hunger") (De Vriendt, 2000b, pp. 260–261). Francophone learners may have trouble knowing when to pronounce /g/, and may find it the following combination difficult: vowel sound, /ŋ/, vowel sound (the learner may erroneously add /g/ after /ŋ/) (p. 261). What may be proposed to the learner is an unusually slow sentence, wherein the word containing the troublesome phoneme is pronounced faster: "They're singing for us", where "singing" is uttered faster than the rest of the unit (p. 261). If the learner adds a phoneme (like /g/ or /k/) after /ŋ/ at the end position, De Vriendt (2000b, p. 264) suggests making it so /ŋ/ does not appear in final position anymore, and making the learner repeat: "I have a ring" (mispronounced /rıŋg/) should be turned into "I have a ring, right?" and be repeated. This is called "neighbouring sounds" (p. 264) and comes close to what others call combined phonetics<sup>22</sup> (Renard, 2000c, p. 17).

<sup>&</sup>lt;sup>21</sup> If /ŋ/ appears between vowel sounds, it must be followed by /g/ ("hunger"). If /ŋ/ is followed by an added morpheme starting with a vowel ("er"), and if /ŋ/ consequently appears between two vowel sounds, it either remains /ŋ/ if the root is verbal ("singer"), or it must be followed by /g/ if the root is not verbal ("stronger") (De Vriendt, 2000b, pp. 260–261).

<sup>&</sup>lt;sup>22</sup> Combined phonetics (Renard, 2000b, p. 17) insist on sounds influencing one another; several principles are named and explained in Renard (200b, p. 17) but seem to be of application particularly to French.

*Nuanced* (Renard, 2000c, p. 16), *inverted* (Billières, 2014a), or *modulated* (De Vriendt, 2000b, p. 264) *pronunciation* equals to pronouncing exaggerated sounds. If a learner has issues with the phoneme /e/ (as in "bed") and makes it sound like  $/\epsilon/$  (as in the French word "aide"), the teacher should try pronouncing a "close [e], similar or close to the French /e/ of 'bédé'" (p. 264); a close phone [e] means the tongue is on a higher position in the mouth.

De Vriendt (2000b, p. 266) provides an example of how the teacher may shape the learner's pronunciation by taking both the neighbouring and nuanced segments into account: her approach concerns the mispronunciation of the phoneme /I/, as in "rat", which is often pronounced /r/ (as in the French word "réussir"). What the author advocates is replacing the /I/ by /w/ (as in "wet") and using contexts whereby the pronunciation is eased, that is, "labial<sup>23</sup> consonants" (e.g., /p/ or /b/) and "rounded<sup>24</sup> vowels" (e.g., /p/ as in British "got", /o:/ as in "saw", or /u:/ as in "too"). Working on "the broom's on the roof" or on "our professor is prudent" (p. 266) should lead to closer native-like pronunciation of /I/. Still, the author does not mention what is meant by "working on" (i.e., perception, repetition, etc.).

#### **4.9.** Conclusion

The literature offers various methods to ensure intelligibility or at least improvement in pronunciation. Among these, one finds phonetic notations, whether it being the popular IPA symbols or keywords ( $Q_5$ , "Is the use of a metalanguage effective to teach pronunciation?"). Phonetic notations offer many advantages, such as the potential to learn autonomously (Marks, 2011; Mompean & Fouz-González, 2021; Mompean & Lintunen, 2015) and the possibility to provide precise feedback on pronunciation (Fouz-González & Mompean, 2021; Mompean & Fouz-González, 2021), but phonetic notations have some downsides as well, such as the extra amount of work it represents to teach and to learn (Fouz-González & Mompean, 2021; Mompean & Lintunen, 2015). Empirically speaking, learning a phonetic notation has been shown to improve perception (Fouz-González & Mompean, 2021).

Then, technology offers interesting and efficient ways to improve the learners' pronunciation (perception and production alike [Fouz-González, 2020; Thomson, 2012; Thomson, 2011]) as individualised learning and personalised feedback (Levis, 2007; Rogerson-Revell, 2021; Venkatagiri & Levis, 2007) become possible with various apps for example, but

 $<sup>^{23}</sup>$  Labial is a feature and means the consonant is pronounced at or with the lips (van Oostendorp, 2020, pp. 26–31).

<sup>&</sup>lt;sup>24</sup> Rounded refers to the position of the lips, which are then rounded (Crystal, 2008, p. 65).

teachers should always ensure that these apps are pedagogically relevant (e.g. not solely repetition-based) (Derwing & Munro, 2009). Additionally, drama and mimicking speakers' voices appear to lead to comprehensibility, intelligibility, accentedness, and fluency improvements (Foote & McDonough, 2017; Galante, 2018; Galante & Thomson, 2017; LaScotte & Tarone, 2022; LaScotte & Tarone, 2019). Working on production is effective, and so is working on perception, and that is what HVPT focuses on: through exposure to various sorts of input, learners increase their phonological awareness (Thomson, 2018), which may then solve production problems issuing from perception ones (Detey et al., 2016; Mehlhorn, 2007; Setter & Jenkins, 2005). Other methods to teach pronunciation are: The Silent Way (Richards & Rodgers, 2001b), motor-based techniques (e.g., using mirrors to see mouth movements) (Messum & Young, 2021) and the Verbo-Tonal Method (Renard, 2000a) (Q<sub>2</sub>, "What are effective techniques to teach pronunciation?").

As far as feedback is concerned, it is argued to be a central matter to helping learners situate themselves in relation to their pronunciation goals, whether feedback is positive (i.e. correct input) or negative (i.e. explicit correction) (Saito & Lyster, 2012). There does not however seem to be conclusions on the most efficient type of feedback (Q4, "How does feedback impact pronunciation teaching?").

In conclusion, there does not appear to be a clear consensus on what method is best to address pronunciation. What researchers agree on are several guidelines though. It seems that enhancing perception (e.g., HVPT, VTM, the Silent Way, phonetic notation enabling to differentiate what one hears and what one thinks one hears), dealing with suprasegmentals (see e.g., technology, VTM), exposing learners to various input (e.g., perception exercises, feedback, HVPT), providing precise feedback (e.g., phonetic notations, feedback, technology), raising awareness of the role the body plays (e.g., drama, Articulatory Approach, VTM), repeating and imitating (e.g., drama, shadowing, mirroring, VTM), and making use of technology are generally advocated. Still, looking at comprehensive approaches (e.g., the Silent Way, VTM, the Articulatory Approach) these sometimes appear to be opposite: the Silent Way does away with input, while VTM insists on the teacher constantly speaking English and modelling the learners' production, which is not recommended by the Articulatory Approach.

### 5. Teaching: Actual Practices

#### **5.1. Introduction**

This last theoretical chapter offers insights on what some teachers say they do, what other teachers actually do and what learners have to say on pronunciation. The main research question answered in this chapter is: "What is the pronunciation teaching situation in some countries  $(Q_{11})$ ?" Textbooks are also dealt with and this provides an answer to the question: "Are textbooks suitable ways to teach pronunciation  $(Q_3)$ ?" The answers are useful in positing hypotheses in Chapter 6. Prior to finding these answers,  $Q_3$  must shortly be addressed. "Suitable" is used instead of another adjective like "effective" for I do not discuss any data on the supposed efficiency of textbooks (which would then be tested measuring intelligibility improvements through pre- and posttests for example). In addition, the efficiency of textbooks is "supposed" because one must distinguish what textbooks contain and how they are used by the teacher or the pupil. In short,  $Q_3$  pertains to the contents of textbooks, or to what they have to offer.

Surveys on teachers' point of views and their teaching practices are valuable tools to "[help] ensure research and pedagogical advice are appropriately directed" (Couper, 2017, p. 820). Yet, compared to other branches of teaching (e.g., grammar, focus on form, etc.), surveys on pronunciation seem to have been given little attention (Baker, 2011a, p. 82; Murphy & Baker, 2015, p. 2). In spite of their interest to better gain "insight into how teachers' beliefs and knowledge interact in the language classroom and influence their pedagogical behaviors [sic], actions and activities" (Baker, 2011b, p. 82), such surveys must be carefully considered because they may not be representative: some types of participants may want or not to take part in it (Foote et al., 2012, p. 16), most surveyed (and researched) teachers teach proficient adults (Pennington, 2021, pp. 6–7), opinions and answers may be wrong (as in Couper, 2021, p. 140; Galante & Piccardo, 2022, p. 383) and what the participants say they do, is not often what they actually do, as shown in Sato & Kleinsasser (1999, p. 513). That is why questionnaires on stated data are simply not enough to really comprehend what is done in class, and must then be supplemented with "classroom observations [and] ... student reports" (Baker, 2014, p. 155). They may also suffer from various biases (see 6.3.3.). In other words, results from the following surveys must be taken with a pinch of salt. They nevertheless constitute useful tools, as explained above; other advantages are detailed in the next chapter (see 6.3.3.).

## 5.2. Teachers' Reported Practices and Beliefs

Because of the great variety found in articles dealing with teachers' reported practices and beliefs, the methodology is detailed in an appendix (see Appendix A). The articles are examined based on the similarity of the conclusions they draw.

A first major theme that arises from most articles on teachers' reported practices and beliefs is the training. This is apparent in the amount of questions teachers ask themselves regarding pronunciation (Couper, 2021, p. 134): they have questions because they lack formal or scholarly education in pronunciation. They wonder about the learner (e.g., how to make them realise the importance of pronunciation, how to make them aware of their possible shortcomings, how to test and measure their pronunciation abilities), the contents (e.g., what aspects to teach in class, whether L1 and L2 comparison is relevant), the ways to teach (e.g., what activities to use), the legal texts, textbooks, and time (e.g., how to implement pronunciation despite it not being addressed in textbooks or in curricula, how to cope with time issues) and about their own pronunciation (e.g., how to improve their own pronunciation, especially with regard to the differences between their L1 and English) (pp. 133-136).

Moreover, Henderson et al. (2012) summarise that lack of training in one term: "participants' comments reveal that many, if not most appear to be amateurs when it comes to teaching pronunciation" (p. 13), amateurs in the sense of being passionate, but also in the sense of not having the necessary skills: both the evaluation of their "training in relation to teaching pronunciation" and the evaluation of its quality turned out to be poor, except in Finland<sup>25</sup> (p. 13). Often, working teachers did not learn about the pronunciation methodology (Foote et al., 2012, p. 10), but were provided with knowledge of more formal fields (i.e., phonetics and phonology) which may still be incomplete, especially as far as stress and intonation are concerned (Couper, 2017, pp. 829-830). Having taken phonology or phonetics classes might however not concern a majority of practitioners (Baker, 2011a, p. 87; Foote et al., 2012, p. 10): pronunciation training may have revolved around repetitions of different kinds (Baker, 2011a, p. 87; Couper, 2017, p. 829), or may have been fulfilled during continuous learning (Couper, 2017, p. 830; Foote et al., 2012, p. 10). The poor knowledge of pronunciation and its teaching may lead to a reliance on coursebooks and colleagues (Baker, 2011a, p. 90), reduced confidence (Baker, 2011a, p. 91; Couper, 2017, p. 831; Foote et al., 2012, p. 14), reluctance (Macdonald, 2002, pp. 7–11), or even avoidance of pronunciation in class (Couper, 2017, p. 831). Besides, the teachers themselves view training as influential (Baker, 2011a, p. 88), but most consider

<sup>&</sup>lt;sup>25</sup> The different countries surveyed were: Finland, France, Germany, Macedonia, Poland, Spain and Switzerland (Henderson et al., 2012, pp. 8–9).

what they have received in terms of teaching as "woefully inadequate" (Henderson et al., 2012, p. 23). In conclusion, based on the results they got, Henderson et al. (2012) state that "limited or no specific training in teaching pronunciation seems to be the norm, but non-native English speaker respondents have usually received training in improving their own pronunciation" (p. 15). Nevertheless, this overall finding does not mean that teachers are not willing to learn and fill in the gaps in their education (Couper, 2017, p. 830; Foote et al., 2012, p. 14).

Also, this same conclusion does not entail a total absence of pronunciation teaching in class, for it is deemed important (Henderson et al., 2012, p. 9) at every level (Foote et al., 2012, p. 15) and pronunciation instruction is seen as effective and not boring (pp. 13–14). Its implementation ranges from "up to a quarter of [the] weekly teaching time" (Henderson et al., 2015, p. 269), a well-established component of their lesson, and on-the-spot feedback (Foote et al., 2012, p. 10), to "an [unintegrated] and [non-fundamental] [position as an] element of language or language learning" (Macdonald, 2002, p. 10). The assumption often shared by teachers that "having a good pronunciation is sufficient for teaching pronunciation" (Henderson et al., 2012, p. 14), coupled with their highly-self-rated pronunciation skill (p. 12) might explain this paradoxical situation (i.e., global absence of training but teaching nonetheless).

On goals, teachers advocate intelligible communication (Couper, 2017, p. 831; Henderson et al., 2012, p. 10), but some express a preference for "accuracy, clarity and fluency" (Couper, 2017, p. 831) or for native-like proficiency (Timmis, 2002, p. 243). Details may be added as their attitude towards accents and norms. "Accented intelligibility [is] the most desirable outcome" for most instructors (Timmis, 2002, p. 243), with more NS teachers having such a goal than NNS ones. This "accented" precision pertains to many who reject accent deletion because accent is part of the speaker's own identity (Couper, 2017, pp. 833–836; Foote et al., 2012, p. 19); the social dimensions of pronunciation then make up a discouraging factor to some: because it is said to be "very personal" (Macdonald, 2002, p. 9), attempting to change a learner's pronunciation may feel "intrusive" (p. 9) or other instructors may be concerned about imposing a specific norm (Couper, 2017, p. 832). Yet, this is precisely why other teachers pursue accented intelligibility, that is, in order to keep the learner's identity whole (Timmis, 2002, p. 245). Still that goal may not suit learners, who most long for native proficiency (p. 248) for it is seen as "a benchmark of achievement" (p. 242), or as a helping factor in integrating an L1 community (2002, p. 242).

59

As far as the norms are concerned, only a European survey provides data on that matter and the results suggest that RP<sup>26</sup> is the standard most widely employed, despite a stated learners' preference for GA (Henderson et al., 2012, p. 20). Another norm was mentioned but not defined: international English (p. 20).

The final common conclusions are linked to the range of exercises done in class. Textbooks and curricula sometimes play a role in determining the activities (Couper, 2017, p. 832; Foote et al., 2012, p. 11; Macdonald, 2002, pp. 7-11) and the availability of suitable resources and material is another determining element (Macdonald, 2002, pp. 7–11). One may then think of the language laboratory and the many possibilities it offers, but the access to one seems to depend a lot on the country: 7% of the German respondents claimed they had a language laboratory available in their school, against 100% in Switzerland. (Henderson et al., 2015, p. 269). The most generally used exercises are: "minimal pairs ..., employing some repetition ..., using mirrors,<sup>27</sup> having students record and listen to themselves, and using diagrams of the mouth to explain correct articulation and having students feel where/how they were speaking" (Foote et al., 2012, pp. 12–13). Perception was found to be commonly trained as well (Henderson et al., 2015, p. 270). Phonetic symbols were also discussed and a minority of instructors teach how to write them while an overwhelming majority teaches how to read them, but their selection depends on the difference between the L1 and L2 segments, the believed complexity of a sound, the teachers' mastery of the symbols, and the teachers' degree of ease with technology, which is considered crucial to phonetic symbols teaching (Henderson et al., 2015, pp. 271–272). Evaluation of pronunciation involves "oral performances ... reading aloud and listening comprehensions" (p. 275). If there are not any details on what teachers evaluate, an analysis of the pronunciation activities teachers speak of reveals that most exercises focus on segments (55 out of 144) (e.g., sound pairs), then on suprasegmentals (39) (e.g., intonation marking), then on the two levels (26) (e.g., repetitions) and on "other" things<sup>28</sup> (24) (Foote et al., 2012, p. 13).

## **5.3. Teachers' Observed Practices**

<sup>&</sup>lt;sup>26</sup> RP is used here, in place of BBC Pronunciation, to match the contents of the survey, so is GA.

<sup>&</sup>lt;sup>27</sup> Although there are not any details as to what is referred to when talking of mirrors, it seems to be actual mirrors (and not mirroring activities) to work on lips positioning in comparison with a model.

<sup>&</sup>lt;sup>28</sup> Only one instance of an activity within that category is given in the article, "working one on one" (Foote et al., 2012, p. 13), but this activity is not detailed or explained at all.

As mentioned previously, there may exist a gap between what instructors say they do and what they actually do (see Sato & Kleinsasser, 1999, p. 513). It is therefore interesting to observe teachers at work.

Foote et al. (2016) recorded 40 hours of three teachers' practices who taught 11- and 12-year-old francophone learners in Quebec in an intensive ESL context (pp. 181-182). A first observation is the fact that none of the teachers used pronunciation-dedicated coursebooks and that pronunciation did not constitute separate teaching lessons (p. 186). Then, percentages were calculated and pronunciation made up 10% of the "language-related episodes", against 70% for vocabulary and 20% for grammar (p. 187). More on these language-related episodes (i.e., moment dedicated to either pronunciation, vocabulary or grammar), they accounted for 17% of the total classroom time, with the remaining percents being "classroom management" (75%), and "personal anecdotes, text-based input and discussion of text-based input" (8%) (p. 187). These numbers are far less than the quarter of weekly teaching time mentioned in surveys (Henderson et al., 2015, p. 269). During the 40 hours, only the segmental level was addressed, encompassing "possessive and plural 's', past tense endings, 'th' sounds and phonemes /f/, /v/, /w/, /p/, /b/, /i/ and /I/" (Foote et al., 2016, p. 188). Thus, this finding matches teachers' stated practices (Foote et al., 2012, p. 13). Still, there was individual variation as to the preferred use of words or of sounds to teach segments.

Amidst the activities used or pronunciation-related episodes, spontaneous feedback, recasts in this case (Foote et al., 2016, p. 193), occurred the most, followed by repetitions, "often done as a choral repetition of a tongue twister [sic]<sup>29</sup>" (p. 189). The researchers try to explain the relative "lack of emphasis on pronunciation [in] a high communicative, intensive teaching context" (p. 191) and the mismatch between what is done and what is said to be done in surveys (e.g., activities relying on suprasegmentals) (p. 191) and so they call upon familiarity with pronunciation errors, that is especially important in classes where all have the same L1: because pupils and teachers have got used to mispronunciations, these do not entail miscommunication anymore (p. 192), which is an effect stressed by several researchers (Gass & Varonis, 1984, p. 81). Another explanation is the fact that segments are less difficult to correct than suprasegmentals (Roberge, 2000b, p. 137), which may require "specialised terminology" (Foote et al., 2016, p. 192), and specialised material (e.g., computers to display waveforms to work on pitch [see 4.3.]); the overall lack of formal training among practitioners may be a last

<sup>&</sup>lt;sup>29</sup> Bowen (1972, p. 86) stands very critical towards tongue-twisters, qualifying them as "wholly without merit" for he does not understand how putting a large number of problematic sounds one after the other may help.

reason (Baker, 2011a, p. 91; Couper, 2017, p. 831; Foote et al., 2012, p. 14; Macdonald, 2002, pp. 7–11).

Baker (2014) interviewed and observed five teachers who taught in the same North American Intensive English Program, the goal of which is to "help L2 learners of English achieve sufficient language proficiency to succeed in undergraduate or graduate programs in North American universities" (p. 140), so the teaching public consisted here of international students. The observations account for 200 minutes per teacher (p. 142). The researcher did not focus her analysis on the level (segmental or suprasegmental) addressed in class, but rather on the type of activity used, that is "controlled, guided [or] free" (p. 143). Controlled exercises are exercises where the teacher has the leading role (e.g., repetitions, minimal pairs); guided exercises are exercises where a general frame is provided (e.g., information gap, open-ended questions and answers); free exercises are exercises where the learner has the dominating role (e.g., roleplays, oral presentations) (pp. 143-144). A total of 25 techniques<sup>30</sup> were identified and fell, for a great majority (i.e., 15), on the controlled end of the spectrum, six techniques were identified as guided and four as free (p. 145). The results match surveys indicating the most widely used activities (Foote et al., 2012, pp. 12–13). Baker suggests that the teachers who had followed pronunciation methodology classes made use of more varied activities (Baker, 2014, p. 148), assessing the stated influence of training (Baker, 2011a, p. 80).

### 5.4. Learners' Points of View

Whether pupils or students, learners offer interesting insights into views on pronunciation and how it is integrated into the classroom.

Derwing & Rossiter (2002) investigated adult's views on pronunciation: they questioned 100 adult immigrants from various L1 backgrounds on the possible communication problems they had encountered in Canada. A majority indicated difficulties to make themselves understood and in their view, pronunciation was one of the reasons why (p. 160), so it explains the high percentage (90%) of respondents saying they would follow pronunciation instruction courses if they had the opportunity to do so (p. 161).

Similarly, Kang (2010) surveyed 238 adult ESL learners in the United States (123 respondents) and in New Zealand (115 respondents) on pronunciation instruction (p. 107) and

<sup>&</sup>lt;sup>30</sup> Here are instances of activities (see Baker [2014, pp. 146–147] for a detailed list): clapping when reading a word and discrimination (controlled), peer feedback in pairs and class identification of forms uttered by a learner (guided), enacting a movie scene and oral presentations (free).

the results echo those found in Derwing & Rossiter (2002), that is, almost all participants "reported that pronunciation is important for communication, [that] they are concerned about it, and therefore want to improve the way they sound very much" (p. 108). One point the students differed in depending on the country they were studying in was that of willingness to achieve native-like pronunciation proficiency. In New Zealand, 37% of the students "did not think it was desirable to sound like a native speaker", and 26% sometimes made a conscious choice to sound foreign (reminiscent of results found in Gatbonton et al. [2005, p. 504]), while only 5% of the American students expressed such negative opinions towards desirability to sound native, and 8% said they intentionally avoided a native accent (Kang, 2010, p. 110). Eventually, the New Zealand learners pinpointed a lack of a proper model to follow (i.e., the teacher) and of proper guidelines in their pronunciation lessons (pp. 111-112). Therefore, Kang calls for a deeper investigation of the impact of the teacher's pronunciation (p. 114), and this desire can be contrasted, in my opinion, with studies suggesting how little impact teacher input has (see 3.3.5.): despite not clearly helping learners improve their pronunciation, teacher input (and especially stance with regard to accents and models) does seem to play a role in the learners' perception of what to expect from pronunciation instruction, as hinted at by Kang's conclusions (pp. 111-112).

On the other hand, Tergujeff (2013) based a discussion of pronunciation on an overlook of textbooks<sup>31</sup> with 10 EFL Finnish learners, six of whom were 15 or 16, two of whom were 10 and two of whom were 18. In spite of not having a specific pronunciation norm as a target, the learners expressed preferences for "an accent that is familiar to them, … British Received Pronunciation" (p. 84), but only as an ideal: the learners did insist on targeting intelligible and fluent speech (p. 84). Besides, the RP<sup>32</sup> accent is familiar because it was the accent they were mostly exposed to in class (p. 88), corroborating data from Henderson et al. (2012, p. 20). According to the learners, common pronunciation activities involve marking lexical stress after listening to a word, spontaneous feedback, choral repetitions, aloud readings and audio segments that can be listened to at home (Tergujeff, 2013, pp. 85–87), but other less controlled activities (such as oral presentations) may not have been mentioned by the learners for they do not, in their minds, fall within pronunciation activities (p. 86). Phonetic notations are not part of these commonly used activities (p. 87), except at primary schools (p. 92), which is in line with comments on the pronunciation approach in high schools, that is, the learners believe "[it]

<sup>&</sup>lt;sup>31</sup> The books were used to start a discussion with the learner based on exercises they would regularly do in them (Tergujeff, 2013, p. 85).

<sup>&</sup>lt;sup>32</sup> RP is used here instead of the preferred BBC Pronunciation to better match the contents of the article.

is not paid enough attention [while] [it] is taught extensively at the primary level" (p. 92). Overall, Tergujeff's findings match what has been observed in classrooms and stated by teachers.

### 4.5. Textbooks

One may argue that the last actor in the classroom, next to the learners and the teacher, is the textbook, which may constitute the skeleton of the lessons (Thornbury, 2013, p. 217), specifically for beginner teachers who may be in dire need of material (Diepenbroek & Derwing, 2014, p. 2) or for any teacher who may need information they lack on a specific subject (e.g., pronunciation) (Derwing et al., 2013, p. 25); to teachers, they also form one essential resource in integrating pronunciation in class (Baker, 2011a, p. 90; Couper, 2017, p. 832, 2021, p. 135; Foote et al., 2012, p. 11). In addition, they are one of the ways to bridge the gap between researchers and practitioners (Gut et al., 2007, p. 5). Thus, textbooks may be interesting to analyse in terms of what they have to offer with regard to pronunciation teaching.

Derwing et al. (2013, p. 22) quantitatively and qualitatively analysed 48 popular ESL coursebooks out of 12 series (which were identified as the most sold in Canada by their respective publishers<sup>33</sup>), as well as six teacher's guides, and their conclusion is telling: "many of the series reviewed ... provide inadequate support to either the teacher or the learner, evidenced by a limited range of task types, few clear explanations ... and limited review of pronunciation features covered" (p. 37). As far as quantity is concerned, the percentage of the series dedicated to pronunciation varies greatly, from 0.4% to 15.1% (p. 28), with some individual books containing no pronunciation exercises at all (p. 30). The most common topics addressed are the following: intonation (290 activities counted out of a total of 1844 activities), sentence stress (256), lexical stress (218), rhythm (182), "reductions" (152) (e.g., "wanna" or "dunno"), vowels (136) (p. 28). The suprasegmental level is then more often covered than the segmental level, in spite of a great variety of vowels and consonants addressed (p. 33). Even if many elements are covered, the authors' criticism stems from the sporadic frequency of theoretical explanations that may not help understand and learn suprasegmentals (pp. 34, 36). If the number and the different types of activities is inconsistent across all books, one kind of exercise is "most consistently used across all series analysed": listen and repeat tasks, which

<sup>&</sup>lt;sup>33</sup> These publishers were: Oxford University Press, Pearson Education, Longman, Pearson Longman and Cambridge University Press (Derwing et al., 2013, p. 26).

may reveal ineffective if not supplemented by adequate and explicit guidelines and feedback (p. 35). Other authors criticise such practices as well (Levis & Grant, 2003, p. 14).

Careful use of textbook material is pinpointed by Levis and Cortes (2008, as cited in Foote et al., 2012, p. 5) who noted that minimal pairs in coursebooks are peculiar: one of the words of the pair is seldom used in actual speech. Then, the models provided by textbooks consist most often of GA and BBC pronunciation (Levis, 2005, p. 371), or of native speakers (Thornbury, 2013, p. 215). Furthermore, Tergujeff (2013, p. 85) decided to use coursebooks to prompt reactions on pronunciation instruction from Finnish pupils but noticed that it revealed difficult to simply find a pronunciation-related exercise; she also listed the activities used: "phonetic training, reading aloud, imitation, rhymes, rules and instructions, awareness-raising activities, spelling, dictation and ear training" (p. 85). Yet, the pupils' reactions revolved around far fewer task types (p. 85), maybe pointing to a limited use of the made-available activities. Finally, Rossiter et al. (2010, p. 599) report that fluency is very seldom trained in ESL textbooks.

Overall, textbooks do not appear to offer suitable material to solely base one's teaching of pronunciation on. Yet, this does not mean that the textbooks that were surveyed and analysed cannot be efficient as there is not any data on their efficiency, on the improvement they could lead to, and on their actual implementation and use in class. In other words, a teacher could use a textbook to instruct their pupils in pronunciation and complete or improve its teaching with other material.

#### **4.6.** Conclusion

Two research questions may now be answered. Although the approach to pronunciation teaching varies across countries, a number of common points have been made clear thanks to several surveys (Q<sub>11</sub>, "What is the pronunciation teaching situation in some countries?"). Training in formal fields seems to have been incomplete (Couper, 2017) and training in pronunciation methodology seems to be generally lacking (Foote et al., 2012), with various consequences. Among these, one finds reduced confidence (Couper, 2017), reliance on external pronunciation resources (Baker, 2011b), and avoidance of pronunciation overall (Couper, 2017; Macdonald, 2002). Yet, teachers tackle pronunciation in class nonetheless (Henderson et al., 2015), and they are even willing to learn more and fill in the gaps in their training (Foote et al., 2012). Their teaching goals revolve around either native-like proficiency (i.e., BBC Pronunciation or undefined international English) (Henderson et al., 2012; Timmis, 2002), or,

for most of them, (accented) intelligibility (Couper, 2017; Henderson et al., 2012; Timmis, 2002) because they are fully aware of the social aspects of pronunciation (Macdonald, 2002).

Turning to what teachers actually do, they state the activities first depend on the material available, or on what the textbooks provide (Couper, 2017; Macdonald, 2002), which deal with pronunciation differently in terms of models (e.g., NSs only), quantity (e.g., low variety of exercises, mainly listen and repeat tasks) and quality (e.g., very few clear explanations) (Derwing et al., 2013; Levis, 2005; Thornbury, 2013) (Q<sub>3</sub>, "Are textbooks suitable ways to teach pronunciation?"). Then, classic classroom pronunciation activities are repetitions, perception-based or phonetic-notation-based exercises, oral performances and various controlled activities (e.g., minimal pairs) (Baker, 2014; Foote et al., 2012; Henderson et al., 2015). Eventually, common to many teachers, is their emphasis on segments over suprasegmentals (Foote et al., 2012).

In spite of addressing pronunciation in class, what the teachers claim they do, does not appear to satisfy their learners who demand more of its teaching (Derwing & Rossiter, 2002; Tergujeff, 2013) because they consider pronunciation highly (Kang, 2010). Interesting is the learners' goals which range from sounding native, to consciously not sounding native or being intelligible (Kang, 2010; Tergujeff, 2013), and they therefore do not always match what the teachers aim for.

Learners may not be happy with how pronunciation is dealt with in class because actual practice does not correspond to stated practice. Researchers observed teachers for up to 40 hours and realised pronunciation is barely introduced in class, when compared to grammar or vocabulary, and when pronunciation is indeed introduced, it concerns the segmental level only (Foote et al., 2016).

### 6. Survey: Stated Practices

### **6.1. Introduction**

As mentioned previously (see 5.1.), surveys are valuable tools. Most interestingly, the survey used here and the answers it has received allow to compare what research recommends doing and what practitioners say they do. The aim of the questionnaire is to look at the state of English pronunciation teaching in francophone Belgium, in EFL contexts, in secondary and further education. Several themes are<sup>34</sup> surveyed: the goals pursued and the norms used in pronunciation teaching, the reasons influencing pronunciation teaching as a whole, the training teachers have received in pronunciation (teaching), the contents taught, the activities used, and the integration of pronunciation into the classroom.

## **6.2. Hypotheses**

In total, this dissertation targets 23 different questions, 12 of which are answered in the previous chapters and 11 of which are answered through the results obtained via the questionnaire. Almost every question<sup>35</sup> answered theoretically is answered practically thanks to the participants' responses. The research questions, along with the hypotheses ( $H_{13}$ ), are presented in the Table 2.

### Table 2

Research Questions			Hypotheses	
Q13	What goals do teachers pursue in pronunciation teaching?	H <sub>13</sub>	The pronunciation goals revolve around either being intelligible or having a BBC Pronunciation accent.	
Q14	What techniques do teachers use to teach pronunciation?	$H_{14}$	Teachers mainly use repetitions and oral presentations.	
Q15	Do teachers use textbooks to teach pronunciation?	H15	Teachers rely on (dedicated) textbooks to teach pronunciation.	
Q <sub>16</sub>	How do teachers use feedback regarding pronunciation?	$H_{16}$	Teachers mainly use recasts when a segment impairs intelligibility.	
Q17	Do teachers use any kind of metalanguage to teach pronunciation?	$H_{17}$	Teachers may use IPA.	

# Research Questions and Their Hypotheses for the Survey

<sup>&</sup>lt;sup>34</sup> Present tenses instead of past tenses are preferred in this chapter, especially when the structure of the survey is detailed (see 6.3.2.). At the time of writing Chapter 5 (beginning of May 2023), the survey is still open; the results are presented in the present because they reflect common practices or the teachers' habits.

<sup>&</sup>lt;sup>35</sup> The two questions which do not receive a practical answer in Chapter 5 are: "What is the pronunciation teaching situation in some countries  $(Q_{11})$ ?" and "How does oral teacher input influence pronunciation  $(Q_{12})$ ?"

	Research Questions		Hypotheses
Q18	Do teachers call upon non-native	$H_{18}$	Teachers solely use native speakers in
	accents in class?		class.
Q19	What norms do teachers refer to when teaching/evaluating	$H_{19}$	Teachers refer to BBC Pronunciation.
Q <sub>20</sub>	pronunciation?		
	What features do teacher	H <sub>20</sub>	Teachers rather focus on segments and on lexical stress.
	pronunciation?		
	-	H <sub>21.1</sub>	Teachers lack training in pronunciation theory.
Q <sub>21</sub>	Why do teachers (not) teach pronunciation (much)?	H <sub>21.2</sub>	Teachers lack training in pronunciation teaching methodology.
		H <sub>21.3</sub>	Teachers are not confident when teaching pronunciation.
		H <sub>21.4</sub>	Teachers believe exposure to input is enough.
		H <sub>21.5</sub>	The lower teachers rate their pronunciation abilities, the less likely they are to teach pronunciation thoroughly.
Q22	What kind of training in (teaching) pronunciation did/do teachers get?	H <sub>22.1</sub>	Teachers received little training in pronunciation teaching methodology.
		H <sub>22.2</sub>	Teachers received some training in phonology and/or phonetics.
		H <sub>23.1</sub>	Pronunciation-dedicated moments mainly arise whenever communication breaks
Q <sub>23</sub>	How is pronunciation integrated into the classroom?	H <sub>23.2</sub>	down. The implicit or explicit integration of pronunciation into the classroom depends on the teachers' self-rated pronunciation level, in terms of pronunciation, ability to teach it, and mastery of theoretical concepts.

The hypotheses are derived from conclusions and reports from various surveys, observations and from authors' comments (see Chapter 5); personal experience<sup>36</sup> with the Belgian school system clarified some hypotheses. The following paragraphs explain where the hypotheses are taken from.

It is hypothesised that pronunciation goals in francophone Belgium revolve two poles: being intelligible or having a BBC Pronunciation accent ( $H_{13}$ ). It comes from teachers who

<sup>&</sup>lt;sup>36</sup> My personal experience with the Belgian teaching system consists of a bachelor's degree in Germanic languages (Agrégation de l'enseignement secondaire inférieur, orientation langues germaniques) obtained in 2019 at Les Rivageois; this schooling comprised 160 hours of traineeship in schools. I also taught various English and Dutch classes in 2019-2020, and I taught English for 3 months in 2022.

reported on their teaching goals (Couper, 2017, p. 831; Henderson et al., 2012, p. 10; Timmis, 2002, p. 243). The norm most used in Europe, that is,  $RP^{37}$  (Henderson et al., 2012, p. 20) and the norm used in Finland specifically back the hypothesis (Tergujeff, 2013, p. 88). If teachers target BBC Pronunciation, it only seems logical that they use it as a norm for teaching, evaluating, and correcting pupils (H<sub>19</sub>), and that they use it, along other native accents, in listening comprehensions (H<sub>18</sub>). These hypotheses are also supported by analyses of textbooks that mostly provide native models (Levis, 2005, p. 371; Thornbury, 2013, p. 215).

It is assumed that teachers mainly use repetitions and oral presentations when teaching pronunciation ( $H_{14}$ ). This hypothesis comes from generally used pronunciation exercises (Foote et al., 2012, pp. 12–13; Tergujeff, 2013, pp. 85–87), evaluation practices (Henderson et al., 2015, p. 275), attested common exercises (Baker, 2014, p. 148; Foote et al., 2016, p. 189), and textbook activities (Derwing et al., 2013, p. 35). It is quite possible that instructors use a wider range of activities, and these may include other exercises research bring up, such as aloud reading for example (Henderson et al., 2015, p. 275).

It is advanced that teachers rely on (dedicated) textbooks to teach pronunciation (H<sub>15</sub>). This assumption stems from surveys where textbooks were found to be influential in pronunciation teaching (Baker, 2011, p. 90; Couper, 2017, p. 832; Foote et al., 2012, p. 11; Macdonald, 2002, p. 7-11). The occasional reliance on and the global importance of coursebooks also point towards that hypothesis (Derwing et al., 2013, p. 25; Diepenbroek & Derwing, 2014, p. 2; Thornbury, 2013, p. 217).

It is posited that teachers mainly use one type of feedback, that is, recasts, in one specific situation, that is, when a segment impairs intelligibility ( $H_{16}$ ). This hypothesis is based on comments made by a researcher on pronunciation implementation (Pennington, 2021, p. 6), on observed teaching practices (Foote et al., 2016, p. 193), on learners' statements (Tergujeff, 2013, pp. 85-87) and on the level that teachers seem to mostly focus on, that is, segments (Foote et al., 2012, p. 13, 2016, p. 188). Because the use of feedback seems so important, it is also used to suggest another hypothesis: pronunciation-dedicated moments mainly arise whenever communication breaks down ( $H_{23.1}$ ). This statement must be linked to the exercises used for pronunciation ( $H_{14}$ ).

<sup>&</sup>lt;sup>37</sup> RP is used here instead of BBC Pronunciation, because it is explicitly mentioned by the authors (Henderson et al., 2012, p. 20).

Besides, how pronunciation is tackled (i.e., explicitly or implicitly) is argued to depend on how teachers rate themselves in terms of how good they believe their own English pronunciation is, how good they believe their mastery of theoretical concepts is, and how good they believe to be at teaching pronunciation (H<sub>23.2</sub>). In this dissertation, pronunciation is assumed to be dealt with implicitly if teachers rate themselves poorly overall, but it is not assumed that a high self-given rating on pronunciation always entails a highly explicit integration of pronunciation, for grasp of pronunciation theory and adequate teaching capacities may play a role as well in the explicit or implicit nature of pronunciation teaching. It is nonetheless expected that teachers give themselves low scores when it comes to pronunciation theory for they appear to lack training in phonetics and/or phonology (Baker, 2011, p. 87; Couper, 2017, p. 829-830; Foote et al., 2012, p. 10; Henderson et al., 2012, p. 13), and for textbooks seem to offer inadequate theory (Derwing et al., 2013, pp. 34–36).

Furthermore, because segments are said to be what teachers mostly focus on, it is hypothesised that the features stressed in class are segmentals and lexical stress ( $H_{20}$ ), since segments (Foote et al., 2016, p. 188), along with lexical stress (Tergujeff, 2013, pp. 85-87) are reported to be taught. Suprasegmentals are said to be addressed in class (Foote et al., 2012, p. 13), hence lexical stress also standing at the heart of pronunciation teaching.

The use of a metalanguage is claimed to involve IPA only, and is claimed not to be regular (H<sub>17</sub>). This hypothesis relies on the statements made by teachers on their use of phonetic notations (Henderson et al., 2015, pp. 271–272); IPA was chosen over other metalanguages because of its wide use in the pronunciation field (Mompean & Fouz-González, 2021, p. 157). Despite teachers stating they use a phonetic notation, it is expected to be occasional (therefore "may" in the hypothesis), based both on supervisor Mr Simons's input and on personal experience.

In addition, seeing how training influences practices (Baker, 2014, p. 148; Baker, 2011, p. 80), it is advanced that reasons for (not) teaching pronunciation (much) are: the believed positive impact of exposure to input that thus replaces formal teaching (H<sub>21.4</sub>), the lack of training in pronunciation theory (H<sub>21.1</sub>), the lack of training in pronunciation methodology (H<sub>21.2</sub>) (Baker, 2011, p. 91; Couper, 2017, p. 831; Foote et al., 2012, p. 14; Macdonald, 2002, p. 7-11), and the lack of confidence that results from the supposed absence of training (H<sub>21.3</sub>) (Baker, 2011, p. 91; Couper, 2017, p. 831; Foote et al., 2012, p. 14). The impact of teacher input is hypothesised to play a determining role in how much of pronunciation is taught, because it is strongly recommended by researchers to constantly speak English in class (De Vriendt, 2000,

pp. 248-249) and by legal Belgian documents (Fédération de l'Enseignement Secondaire Catholique, 2018, p. 60; Wallonie-Bruxelles Enseignement, 2020, p. 55). Then, how thorough teachers are in their approach of pronunciation (e.g., a wide array of elements seen in class, introduction to IPA, explicit instruction, varied techniques, use of resources, etc.) is claimed to be dependent on their self-ratings of their own pronunciation, of their mastery of pronunciation theory and of their teaching abilities ( $H_{21.5}$ ).

Finally, the reported lack of training in teaching methodology, in phonetics or in phonology, coupled with the global dissatisfaction with that training (Henderson et al., 2012, p. 23) form other hypotheses, namely that teachers received little training in pronunciation teaching methodology (H<sub>22.1</sub>) and that they received some training in phonology and/or phonetics (H<sub>22.2</sub>).

### **6.3.** Methodology

#### 6.3.1. Constructing the Survey

Throughout the whole process of constructing the questionnaire, four variables were kept in mind: the public, the language used, the software used and the General Data Protection Regulation. The survey was designed for English teachers in EFL contexts in francophone Belgium, hence the use of French, for it was assumed to be the mother tongue of the targeted public. Then, the respondents could either work in secondary schools, whether lower or upper secondary, and in further education, at any stage. These two criteria were important as to coming up with the necessary items (e.g., in collecting information on the respondent, or specific words used in the questions<sup>38</sup>) and as to the translation of some terms used (e.g., functional load, motor skills). Because of its integrated save feature (see 6.3.3.) and because it is unlimited in the number of responses it can get, Google Forms (https://docs.google.com/forms) was chosen over other software (e.g., LimeSurvey, SurveyMonkey). Question design was highly dependent on the software used because it has a limited number of options; discussions on answer modality were then driven by what Google Forms allows. Legal requirements concerning the General Data Protection Regulation were addressed in the introduction and applied in the questionnaire.

<sup>&</sup>lt;sup>38</sup> For instance, the fourth question in Section 3 had to be written in such a way that both university-trained teachers and non-university-trained teachers could answer: the question is on training in linguistics (which caters for university-trained teachers) or in language (which caters for non-university-trained teachers).

The first step in designing the survey took place in late October 2022 during a meeting with PhD student Mrs Renson, who went into detail about the basics of designing a survey: the types of questions preferred, the order they should appear in, the way to write them, the tools and software available, the sections and what they encompass, the ideal date to launch the survey, and the test periods. She also provided advice on enhancing the number of respondents and insisted on the necessity to have every question from the questionnaire linked to a research question. In early November 2022, a very early draft of the survey was deeply looked at to ensure the demanded connections and to apply the given advice, and was thus modified accordingly.

On 30<sup>th</sup> November 2022, supervisor Mr Rasier revised and commented the very first version of the questionnaire and suggested modifications with regard to pronunciation theory. December marked the beginning of questionnaire proofreading with supervisor Mr Simons. Prior to every meeting (i.e., 19<sup>th</sup> December 2022, 9<sup>th</sup> February 2023, 21<sup>st</sup> April 2023, 26<sup>th</sup> April 2023, 12<sup>th</sup> May 2023), a revised version of the draft was sent to him. With Mr Simons, every question and item were analysed in terms of research questions, hypotheses, relevancy, easiness to analyse and language. On 13<sup>th</sup> March 2023, Mr Rasier corrected the definitions that were used in the questionnaire. When an adequate version of the questionnaire had been put up, it was sent to Mrs Renson for her to proofread it. She thus received the sixth version on 23<sup>rd</sup> April 2023. Her relevant comments and remarks were discussed and taken into account with Mr Simons a few days later. A total of eight versions came before the testing phase, with the ninth one being the final version (see Appendix B).

The trial run took place as soon as Mr Simons and Mr Rasier had validated the questionnaire, that is, on 26<sup>th</sup> April 2023. A crucial factor in determining who could be a tester was their teaching network. It was decided to test six teachers from subsidised free education, three from public schools organised by Wallonie Bruxelles Enseignement, and one from subsidised public education, as recommended by Mr Simons and as to have a number of teacher-testers in line with the current landscape of the Belgian school system. Indeed, there are more schools from subsidised free education than public schools (Renson, 2023, p. 188) and the number of testers from each network matches the amount of pupils in each network: 61.57% of all Walloon pupils attend subsidised free education, 23.38% public schools, and 15.05% subsidised public education (Fédération Wallonie-Bruxelles, 2020). Thus, 10 teachers were chosen as testers and were sent a questionnaire via email; these were the five teaching assistants at the University of Liège (who were teachers in secondary schools as well), two
previous trainee supervisors, two other teachers suggested by Mr Simons, and one student colleague who was simultaneously teaching as well.

Some minor changes were made to the trial-run questionnaire in order to better collect opinions from the testers. For each multiple-choice question (MCQ), an item "other" was made available for the trial run, and at the end of every section, an open-ended question was added so that the respondents could report on any encountered problems, whether it was technical, language-related, knowledge-related, or whether an item or definition was missing. Another open-ended question required them to put in the time they had taken to complete the questionnaire; a last open-ended question asked the testers if they thought any major theme relating to pronunciation was absent. These added questions were made compulsory to ensure comment collecting, so, for example, if the testers had not come across any issue, they were asked to type in a message saying so (e.g., "OK", "/"). These details were explained in the email they received (see Appendix C).

The trial run ended on 12<sup>th</sup> May, when the 10 respondents had answered the questionnaire. Every suggestion, remark, comment and added item were then sorted out in a table (see Appendix D), which was sent to and discussed with Mr Simons on 12<sup>th</sup> May 2023. Despite first wanting to take large tendencies into account (i.e., at least two similar comments), the trial run revealed so fruitful that every remark was looked at and taken into consideration, but this does not mean that every remark indeed made its way onto the final version of the survey.

The changes made to the eighth version, after the trial run, were the following. The form was modified as spelling mistakes were corrected and some lexical units were put in bold to highlight differences between following questions. As far as content is concerned, adverbs, such as, "primarily" or "most frequently" were added whenever a question demanded solely one answer. Some questions and sometimes their guidelines were detailed or slightly changed in order to add items and in order to improve comprehensibility. The detailed changes are provided in the appendices (see Appendix E), so is the trial run questionnaire (i.e., the eighth version) (see Appendix F).

### 6.3.2. Structuring the Survey

The survey is divided into nine different sections, counting the introductory presentation and conclusion. The questionnaire comprises four questions aimed at gathering data on the participant and 36 other questions about pronunciation in general, with each section dealing

with a specific theme. All questions are in line with the legal requirements provided in the General Data Protection Regulation (General Data Protection Regulation, 2016): every answer is indeed anonymous and confidential.<sup>39</sup>

Thirty-eight questions are compulsory, with only two being optional because they require a specific answer to the previous question. For instance, question number 1 (Q1) is about the level the teachers work in (i.e., secondary schools and/or further education), and the second question asks for information as to what subject field they may teach in (i.e., general, technical and/or vocational education), if they selected secondary schools; Q2 had to be made optional, for further education does not make a distinction between different subject fields. Question 20 is the other non-compulsory question.

All of the questions are either closed MCQs or scales (e.g., importance scales, Likert scales<sup>40</sup>), meaning that the participants never have to write any answer themselves, but only have to choose whatever options best suit them, with the reason behind this choice being the relative ease of compiling and analysing such data. The choice of having closed questions only may however have proved harmful to the survey as a whole (see 6.3.3.). There is one open-ended question and it is the non-obligatory Q20: in that question, the respondents have to write the name<sup>41</sup> of the resources they said they use in teaching pronunciation.

Most of the questions allow for one answer only (15 questions in total) or are scales where the respondents have to choose one option only (15 questions in total), while the remaining questions allow for several answers to be selected (nine questions) or are open-ended (one question). Generally, the participants are forced to select an answer and cannot add their own if none of them correspond to their practice. In other words, an "other" option is not available,<sup>42</sup> which may be one flaw of the survey (see 6.3.3.).

Before being questioned, the teachers are provided with definitions of various technical terms used in the subsequent questions. Questions from Section 2 to Section 6 included are preceded by such explanations. Because there are quite a few definitions<sup>43</sup> and because they are

<sup>&</sup>lt;sup>39</sup> Other required measures are ensured at an institutional level.

<sup>&</sup>lt;sup>40</sup> *Likert scales* contain a certain number of similar-themed statements that participants have to choose an option on, that is, they have to select one option among several, such as "totally disagree", "disagree", "agree", "totally agree" (Dörnyei & Taguchi, 2010, p. 27; Krosnick & Presser, 2009, p. 20).

<sup>&</sup>lt;sup>41</sup> Here, "names" refer to actual brand names of pronunciation resources, such as Cambridge's *Pronunciation in Use* for example.

<sup>&</sup>lt;sup>42</sup> There is one exception : Q3 in Section 1 (i.e., "What is your mother tongue?") where teachers can select either "French", "English" or "Other." The participants can then add their mother tongue.

<sup>&</sup>lt;sup>43</sup> There are five definitions for Section 2, nine for Section 3, eight for Section 4, 15 for Section 5, and two for Section 6, which bring the total up to 39 definitions.

from 2-, to 7-line-long, they were much discussed with both Mr Rasier and Mr Simons as to whether they should remain on the Google Forms questionnaire or be put on another document that would be available through a hyperlink. Eventually, they were kept within the questionnaire for they were considered crucial to comprehensibility. On the one hand, they make the questions comprehensible, and they guarantee that both the respondents and I are more or less on the same page as to what the terms mean; the participants may indeed skip the definitions or may understand these differently than expected. On the other, the definitions may have worked against the survey (see 6.3.3.).

The introduction presents the goals of the questionnaire and the method used. It details for whom the questionnaire is, how many questions there are, and how long it should take to answer them all. The introduction also stresses that the answers are anonymous and confidential and that the research interest lies in teachers' practices and opinions, meaning that there are not any right or wrong answers. The introduction invites the respondents to send an email if they are interested in getting the results, and it eventually closes on thanks.

Section 1 aims at collecting information on the respondent's profile through four questions. The teaching level (Q1) is divided between lower and upper secondary education, and so is further education (UF 1-2 and UF 3-12), with the reason being that they require different degrees. The precise years of what lower and upper secondary education cover, are detailed to ensure clarity. Other pieces of information are collected, that is, the subject field (general, vocational, technical [technique de transition, technique de qualification]) (Q2), the mother tongue (Q3), and the amount of teaching experience (Q4), in spans of 5 years, as recommended by Mr Simons.

Section 2 encompasses two questions, revolves around goals and norms, and thus directly relates to the subsection of the same name in this dissertation (see 3.2.). The questions are which norm is used in evaluating the pupils' pronunciation (Q1) and which goal the teachers pursue in terms of pronunciation at the end of the pupils' secondary education (Q2); this time period helps respondents answer with one specific type of pupils in mind. The first question relates to  $H_{19}$  and the second question concerns  $H_{13}$ .

Section 3 contains six questions and relates to various opinions, beliefs about pronunciation and also relates to teacher training. A first Likert scale (Q3) confronts the participants with a wide range of 23 statements and asks whether they totally disagree, disagree, agree or totally agree. Four possibles answers instead of five (e.g., "neutral", "no opinion") are

available to force respondents to make a choice (Dörnyei & Taguchi, 2010, p. 28). After the Likert scale, come other questions on training, that is, what it consisted of in both linguistics (Q4) and language teaching methodology (Q5), if they had ever followed pronunciation classes after graduating (Q6), and if they desire to do so (Q7). The last question of the section (Q8) requires respondents to choose one out of four different schooling periods, which one would be best to start teaching English pronunciation; the four periods were chosen because they are the start of four traditional stages of the Belgian schooling system, and the goals for the end of each of these stages are laid out in legal documents. These periods are: the fifth year of primary education (i.e., the second stage of compulsory schooling), the first year of secondary education (i.e., the second at a go f secondary education (i.e., the third and last stage of secondary education). The hypotheses tested in this section are  $H_{21.1}$ ,  $H_{21.2}$ ,  $H_{21.3}$ ,  $H_{21.4}$ ,  $H_{21.4}$ ,  $H_{22.1}$  and  $H_{22.2}$ .

Section 4 examines teaching contents through seven questions. The first two questions are about how teachers primarily teach (i.e., focus on theoretical explanations or solely on production) (Q9), and about what they actually teach (e.g., vowel/consonant sounds, stress) (Q10). Questions 11, 12 and 13 (as well as Q16 in Section 5) have specific response modalities: they allow respondents to rank a definite set of items regarding one characteristic. For instance, in Q11, participants have to rank seven pronunciation elements in terms of how important they are in their English class, but every item has to be given a different value, from 1 to 7. In Q12, teachers have to choose three pronunciation elements in terms of how troublesome they are for their pupils, and then have to rank three methods of selecting pronunciation elements worth to teach. The last two questions of Section 4 are linked to phonetic notations such as IPA and the possible reasons why teachers use it or not; Q14 is an MCQ and Q15 is a Likert scale. The hypotheses tested here are  $H_{17}$  and  $H_{20}$ .

With 14 questions, Section 5 investigates the teaching methods and thus relates to Chapter 4 of this dissertation. The themes are varied: Q16 deals with different activities (e.g., discrimination, drama) and requires teachers to rank three that they have chosen out of the 14 proposed, Q17 assesses, with a Likert scale, how explicit or implicit<sup>44</sup> teachers are when they

<sup>&</sup>lt;sup>44</sup> *Explicit teaching* is a type of teaching whereby the teacher makes the objectives and steps to achieve these objectives clear; the learners know what is expected from them and are provided with feedback (Renson, 2023, p. 111). *Implicit teaching* is automatic learning of knowledge without explanations as to the reasoning behind that knowledge (p. 116).

tackle pronunciation in class, Q18 is about ways to present a new pronunciation element (e.g., through pictures or various comparisons), Q19 is about the potential use of textbooks or any other resources (e.g., sites, phone applications, posters), and Q20 is one of the two optional questions and asks teachers to type in the name(s) of the textbook(s) or site(s) mentioned in Q19. Questions 21 through 24 offer insights into how immediate feedback is and on what kind of mistakes it is on (i.e., every, main or intelligibility mistakes). Question 21 and Q22 form a pair that differ only in activity researched (i.e., spontaneous speech acts and aloud reading); Q23 and Q24 form another pair that differs only in activity researched (i.e., spontaneous speech and aloud reading). Question 25 is a Likert scale on the precise type of feedback instructors use. Question 26 is another Likert scale on the pronunciation elements focused on. Question 27 asks teachers how frequently they use native and non-native accents in listening comprehensions. Question 28 and Q29 are two Likert scales that deal with the same theme, that is, how pronunciation is integrated into the classroom, but Q28 is about whether pronunciation is a stand-alone class every now and then, is integrated into other modules, or whether it is its own independent module, while Q29 pertains to the more classical elements (i.e., vocabulary, grammar, syntax, pragmatics) combined with pronunciation. The hypotheses investigated in Section 5 are H<sub>14</sub>, H<sub>15</sub>, H<sub>16</sub>, H<sub>17</sub>, H<sub>18</sub> and H<sub>23.1</sub>.

Section 6 focuses on vocabulary lists and evaluation with four questions. Question 30 and Q31 examine when there is information regarding pronunciation in a vocabulary list and what it is on. Question 32 looks at how heavy pronunciation weights in tests, and Q33 looks at whether pronunciation can be a criterion in oral tests resulting in failure if not met. Rather than checking hypotheses, Section 6 zooms in on some aspects of pronunciation to detail assumptions made earlier (H<sub>23.1</sub> and H<sub>23.2</sub>).

Section 7 consists of three 10-point scales participants have to situate themselves on regarding three values: how good their own pronunciation is (Q34), how good their mastery of the theoretical concepts in pronunciation is (Q35), and how good their pronunciation teaching ability is (Q36). The point of this section is to compare the participants' self-assigned scores with their answers (e.g., the range of items focused on in regard to their mastery of theory), but it also tests  $H_{21.5}$  and  $H_{23.2}$ .

The conclusion thanks the respondents for their gracious help and invites them to send an email in case they agree on letting me observing one of their classes dealing with pronunciation. The order of the questions is not random at all, but follows a simple pattern: from the more general questions to the more precise ones. For example, the later questions (i.e., Q21 to Q33) tackle very precise themes, like feedback (Q21 to Q26), listening comprehensions (Q28), vocabulary lists (Q30 to 31), or tests (Q32 and Q33). Because they ask respondents to rate themselves and because this can dishearten some if they consider themselves poorly, Q34, Q35 and Q36 are placed at the very end of the questionnaire.

#### 6.3.3. The Shortcomings of the Survey

Questionnaires or surveys such as the one used here suffer from a number of biases, among other possible flaws. The following subsection explains the potential shortcomings of the survey used, after highlighting some interesting aspects of surveys in general.

Questionnaires remain advantageous to researchers in terms of amount of possible themes, and in terms of "time ... effort, and ... financial resources" (Dörnyei & Taguchi, 2010, p. 21): through software, questionnaire answers may be gathered and analysed quite fast, in comparison to one-to-one interviews for instance. If they target pronunciation or any classroom-related subject, they offer possibilities to fine-tune "research and pedagogical advice" (Couper, 2017, p. 820) for the researcher gains access to potential correlations between what teachers believe and what they claim they do (Baker, 2011, p. 82) (see 5.1.).

Still, biases are present in the survey. The *social desirability bias* is one bias inherent to the questionnaire (Krosnick & Presser, 2009, p. 37; Oppenheim, 2001, pp. 138–139): the questions may be obvious as to what answer is expected, or socially "desirable/acceptable" (Dörnyei & Taguchi, 2010, p. 8), leading the respondents to choose that one socially expected answer over what they actually do or believe. Because quite a few respondents<sup>45</sup> are trainee supervisors who have worked with the university of Liège before, the social desirability bias may have seen its effects increased: they indeed know that the questionnaire is supervised by Mr Simons, head of the modern languages methodology department at the university of Liège, and may choose the academically adequate answer. *Self-deception* is another bias whereby a respondent chooses an answer in order to avoid self-deception (Dörnyei & Taguchi, 2010, pp. 8–9). One question where both of biases may influence the answers is the last item of Q25: the respondents are asked how frequently they correct a pronunciation mistake, or not. The answers "often" and "always" (i.e., they hardly ever or never correct pronunciation mistakes) may be

<sup>&</sup>lt;sup>45</sup> It is impossible to tell the percentage of trainee supervisors who answered the questionnaire, but it is quite likely that there are quite a few of them in the respondents, seeing how the survey was made public (see 6.3.4.).

avoided, for leaving mistakes uncorrected may be frowned upon in teaching. The general introduction aims at reducing both theses biases, through stating that there is no right or wrong answers to any of the questions, and assuring the answers are anonymous and confidential (Krosnick & Presser, 2009, p. 39).

The *acquiescence bias* may influence participants as well: humans tend to naturally agree with a statement when they are doubting (Dörnyei & Taguchi, 2010, p. 9; Krosnick & Presser, 2009, p. 21; Oppenheim, 2001, p. 181). To reduce the effects of the acquiescence bias, words with positive and negative connotations in the items were avoided (Dörnyei & Taguchi, 2010, p. 90; Oppenheim, 2001, p. 181), MCQs were included, and overall comprehensibility was ensured (Krosnick & Presser, 2009, pp. 23–24), through using clear definitions and through testing the questionnaire before opening it to a wider audience.

Seeing the amount of questions (four about the respondent's profile and 36 on pronunciation) and of definitions (39 definitions), the *fatigue effect* must be discussed as well: inaccurate answers may arise when a questionnaire is "too long or monotonous" (Dörnyei & Taguchi, 2010, p. 9), so, as a result, the last sections and their questions may suffer from respondents getting tired (Krosnick & Presser, 2009, p. 46). The large amount of definitions, their length, and their complexity in some cases, may increase that fatigue effect. However, the number of questions was kept to a minimum through linking them with research questions<sup>46</sup> and a save feature was added to the questionnaire to allow respondents to leave and come back to the survey any time they wanted, but this feature is not mentioned in the introduction and is usable only if the respondents log on to their internet browser with an email address.<sup>47</sup> Because the questionnaire was sent on its own, meaning that the other master students' questionnaires were sent later on, in a separate email then, fatigue effect was decreased, in comparison with a collective email with several students' questionnaires.

The order the question appear in, or the *order effect*, may impact the participants, especially towards the end (Oppenheim, 2001, p. 112): if a global question follows a precise one, the researcher runs the risk of "putting ideas into the respondents' minds or to suggest that they should have attitudes when they have none" (p. 112). Remaining general at the beginning of the survey, while being more precise and detailed in the last questions aims to reduce the influence of question order. Yet, Q28 and Q29 could have come much earlier in the

<sup>&</sup>lt;sup>46</sup> For example, four questions were deleted after the eighth version of the questionnaire because those did not bear any link to the research questions.

<sup>&</sup>lt;sup>47</sup> None of the email-addresses are collected if the respondents indeed log on to their browser with an email address.

questionnaire for they are quite global (i.e., inclusion of pronunciation modules within the class and combination of pronunciation activities): they could have been put either at the start of Section 5, that is, have become Q16 and Q17, or they could have been put before the IPA subtheme, that is, have become Q14 and Q15.

Additionally, scales are subject to "*error of central tendency*" (Oppenheim, 2001, p. 233), or the tendency for questionnaire participants to avoid the "extreme categories" (p. 233). For example, in Q11, the respondents may be less inclined to use the two ends of the scale for some items, or the respondents may have avoided the items "totally (dis)agree" in Q3.

As recommended (Dörnyei & Taguchi, 2010, pp. 41–42), specific lexis and structures were avoided so that the questionnaire could remain clear and unambiguous. Yet, some "nonspecific adjectives" or "universals" (pp. 41-42) were used, such as, among others, "mostly" "main" (e.g., in Q1), "good" (e.g., in Q3), (e.g., in Q23 and Q24), "never/often/sometimes/always" (e.g., in Qs25 to 29). The frequencies "never", "often", "sometimes", "always" were pinpointed as downsides of the sixth version of the survey by Mrs Renson because they are part of these "universals" (p. 42). More precise replacements, such as "none of the pronunciation modules", "one pronunciation module out of three", "two pronunciation modules of three", "every pronunciation module" for Q29, were submitted to and discussed with Mr Simons in the seventh version, but the initial frequency adverbs were eventually kept throughout the whole questionnaire because they were deemed comprehensible, and more so than the replacements that had been thought of. As far as Q27 is concerned, it contains the adverb "mostly" but the adverb was added after the trial run, for one tester wondered if it would be relevant to distinguish between listening comprehensions as exercises or as tests. The comment was discussed, was not considered to make it as is into the final version but was taken into account, as seen with the addition of the adverb. Furthermore, "good" pertains in every case to "good pronunciation"; "correct pronunciation" should have been used instead, for it is defined at the beginning of Section 3, as referring to the norm (i.e., BBC English or GA) the respondents use when correcting pronunciation mistakes.

However, precise wording was used to avoid ambiguities. The second and eighth questions are good examples as they are both detailed enough to avoid ambiguities that the testers had pointed out. In Q2, the participants have to identify the pronunciation goal aimed for the end of schooling; in Q8, the participants have to choose the ideal moment to start teaching pronunciation in English as a first foreign language (as opposed to English as a second language). One may still argue that it is not precise enough since a subject field (i.e., general,

technical and/or vocational education) in Q2 or a grade in Q8 is not mentioned, and since goals may vary depending on the subject field or on the grade.

Negative constructions were reduced as much as possible to ensure comprehensibility (Dörnyei & Taguchi, 2010, p. 42), but one negative construction is used in the last item of Q25 (i.e., "I do not correct the mistake"). It may be argued that, despite the negation, it is fairly easy to match the item with "never", "sometimes", "often" or "always", that it is a necessary item relative to feedback, and that putting the sentence in the positive form would not make much sense since the other items are types of mistake correction.

Moreover, the survey has another shortcoming which was put forward by several of the testers: the difficulty to generalise over every class, subject field, and level the respondents teach (i.e., secondary schools and/or further education), and the difficulty to pick one definite answer. Many times, the testers commented that the level they targeted depended on the class they taught, or they simply said that it was difficult to choose one goal among the three possible answers (Q2, Section 3). Questions 14 and 15 were given similar comments as to the lack of nuance available in the items. The testers gave identical comments on Q22 and Q23: sometimes they delay and sometimes they do not postpone their feedback. These comments were taken into account and the adverbs "mostly", "primarily" and "most frequently" were added to the MCQs, and precisions were given in the guidelines to help participants generalise their answers (e.g., Q2).

Some "forced questions"<sup>48</sup> are also open to criticism, such as Q11 that demands respondents to list pronunciation features in terms of their relative importance on a 7-point scale, with the condition that each feature be on a different point in the scale. Some testers found it difficult to deal with that question and that condition, for they had wished to assess vowel and consonant sounds at the same level of importance, for instance. The results of such forced questions will then have to be considered carefully.

## 6.3.4. Administrating the Survey

On 16<sup>th</sup> May 2023, around 8:30 a.m., the survey was sent to the headmasters of 43 schools in Belgium, from all school networks. The school were selected through the Belgian education

<sup>&</sup>lt;sup>48</sup> *Forced questions* are defined here as questions whereby the respondents must make a choice and are thus not free in their answers (see the explanation for Q11). Although the term "forced question(s)" does not appear as is in the literature, authors mention "forced choice" (Dörnyei & Taguchi, 2010, p. 34), "a set of forced choices" (Oppenheim, 2001, p. 140), and "forced choice questions" (Krosnick & Presser, 2009, p. 22) or "forced choice items" (p. 45).

system's official website (i.e., http://enseignement.be) which provides lists of schools in Belgium: 35 were chosen from the province of Liège, eight schools were selected for the other four francophone provinces. If no balance was reached between the two school networks in Liège, a 1/1 ratio was stricken for the other provinces. The schools were randomly selected. The survey was sent via email, directly to the headmasters if their address was available on the school website (which was mostly the case); if not, it was then sent to the only email mentioned. The email (see Appendix G) introduced who I was, where and what field I was studying, what and for whom the survey was, as well as a few details about the survey. The email also explicitly asked whether they would accept to forward the survey to their EFL teachers.

On 16<sup>th</sup> May 2023, around noon, an email (see Appendix H) was sent to one of the teaching assistants at university, Mrs Van Hoof, to ask her kindly to transfer the survey to the list of trainee supervisors. An email for them specifically was provided as well. The same day, around 3 p.m., the survey was sent to previous colleagues I had worked with in 2019 and in 2020 (see Appendix I), to previous teachers who had supervised my traineeships (see Appendix J), to Mrs Renson (see Appendix K), and to the teaching assistants at university (see Appendix L). I also called upon one teacher from Les Rivageois, with whom I had graduated in 2019, to ask if she would be willing to share the survey with her previous students (see Appendix M). A message was shared on the private university group with old classmates who had already got their teaching degree (see Appendix N). A private message was sent to a former teacher of mine in secondary school with whom I had kept contact (see Appendix O). In total, 15 private messages were sent. Eventually, a message was published on my Facebook wall and was made visible to both friends and their friends (see Appendix P).

With every email, private message or publication, the goal was to entice friends, former colleagues, former teachers to either complete the questionnaire themselves or to share it on social media or privately with teachers they know: *snowball sampling*, or "a chain reaction" (Dörnyei & Taguchi, 2010, p. 61) was thus pursued (Oppenheim, 2001, p. 43).

To enhance participation rates and start a chain reaction, each message contained more or less the same essential pieces of information, that is, the public targeted, the time it would take to complete the questionnaire, the most frequent type of questions (i.e., closed questions), the ensured confidentiality, the absence of right or wrong answers, the research interest, and an invitation to spread the questionnaire on social media or amidst friends and acquaintances. The content did not differ much between the various kinds of messages, but the form did, in order to suit the formality needs: for example, private emails made mention of the headmasters' or teachers' names and the Facebook publication contained emojis.

Strategic choices to ensure better visibility were made as to what time the various messages should be sent: the headmasters received the questionnaires early, for they may take the most time out of all reached candidates to share it, teachers were contacted around 3 p.m. for it is the time most of them finish school and get home, the same logic applies to the time of the Facebook post (i.e., 3 p.m.).

The survey remained open for a bit more than a month, that is, from Tuesday 16<sup>th</sup> May 2023 until Monday 19<sup>th</sup> June 2023. Ideally, it should have been kept open longer, but time constraints urged me to close it sooner. Still, more than half of the responses (i.e., 44) were collected in the first four days (16<sup>th</sup> to 19<sup>th</sup> May). Monday was chosen as the final day to give respondents the weekend to answer.

No reminders were sent to the headmasters or to any of the desired participants, with the reason being the busy periods that the end of May and the beginning of June are for headmasters and teachers alike. Still, a comment was written on the Facebook publication, every week, to make sure it remained visible on people's wall.

#### 6.3.5. Processing the Survey

When it closed, the survey had received 57 answers in total (N = 57). It had also been shared more than 55 times on Facebook, and at least by six headmasters, who confirmed that they had forwarded the survey to their English teachers. It is quite possible and plausible that the questionnaire was shared more times than the given numbers: some Facebook users may choose to send the survey privately, or to share it with particular confidentiality parameters, and some headmasters may have transferred the questionnaire without having said so.

Since most of the questions were compulsory, there was no need to sort out answers and delete the incomplete ones. Google Forms provides various kinds of charts and detailed numbers or percentages and these were used for the analysis.

# 6.4. Analysing the Survey

The following section presents and describes the answers to the questionnaire in chronological order. An actual thorough interpretation and discussion of the results with regard to the hypotheses follow this section; short comments linking both theory and the results are given if relevant. Whenever possible and relevant, tables and figures are shown to ease comprehension

and these are further clarified by explanations and colours (e.g., light green to show the highest percentage and light orange to show the lowest). All percentages are rounded to the first decimal in order to remain coherent in regard to Google Forms, which rounds percentages automatically. Eventually, the questions are referred to either in the text or in the tables or figures in French, so the French punctuation norms are consequently applied (e.g., empty space before a question mark).

#### 6.4.1. Section 1: The Respondents' Profiles

## Figure 1



Number of respondents

Overall, most of the teachers surveyed (57.9%; N = 57) teach in an upper secondary level, but 28 (48.1%) teach at a lower secondary level (see Figure 1). Very few respondents teach in tertiary education: 10.5% in UF 1-2 (which equals to the lower secondary level) and 1.8% in UF 3-12 (which equals to the upper secondary level). The reason why the percentages do not add up is because respondents could<sup>49</sup> select one or more options (this situation occurs multiple times further in this dissertation). In Q1, different combinations then come up: 26 participants teach in the upper secondary level only, 18 teach in the lower secondary level only, 7 teach in both levels, 3 teach in the lower secondary and lower tertiary education, 2 teach in the lower tertiary education level.

<sup>&</sup>lt;sup>49</sup> In the section "Analysing the survey", past tenses are preferred to refer to how the respondents answered (e.g., answer modality), but present tenses are used to refer to the teachers' practices or to the results in general.

Share of Respondents in the Secondary Education Subject Fields Q2) Si vous enseignez dans l'enseignement secondaire, dans quelle(s) filière(s) enseignez-vous ?



When it comes to the subject field (see Figure 2), almost all participants out secondary education teach in general education (96.3%), 29.6% of them teach in the technical education (technique de transition), 33.3% teach in the other technical education (technique de qualification) and 14.8% teach in vocational education. Once again, the participants could select multiple answers: 30 teachers work in general education only, 7 in general and technical (technique de transition), 6 in all subject fields, 5 in general and technical (technique de transition), 4 in all but vocational, 1 in all but general and 1 in professional solely.

#### Figure 3

*Years of Experience Among the Respondents* Q4) Combien d'année(s) d'expérience avez-vous dans l'enseignement ?





Question 3 ("Quelle est votre langue maternelle ?") and Q4 ("Combien d'année(s) d'expérience avez-vous dans l'enseignement ?") are more straightforward as the respondents could only select one item. Regarding Q3, all but one participant are francophone natives.<sup>50</sup> Regarding Q4 (see Figure 3), a bit less than one third of the teachers (29.8%) have from 2 to 5

<sup>&</sup>lt;sup>50</sup> That participant speaks Bosnian as a first language.

years of experience in teaching, 17.5% have been teaching for 6 to 10 years, 15.8% have an 11to-15-year experience, 10.5% have 1 or less than 1 year of experience, 8.8% have been teaching for 16 to 20 years, 7% have from 21 to 25 years of practice, 5.3% have been teachers for 26 to 30 years, 3.5% have a 31-year or 35-year field experience, and 1.8% have more than 41 years of teaching practice. None of the respondents have between 36 and 40 years of teaching experience.

The typical respondent is then an upper secondary, general education, francophone teacher with 2 to 5 years of experience behind them. This average teacher also points to one shortcoming of this questionnaire and of questionnaires in general: the results that are introduced here may not represent the practice of every Belgian EFL teachers. Because there are not that many answers (N = 57), because they probably most teach in the province of Liège,<sup>51</sup> because an important share of respondents may be traineeship supervisors, and because they mostly teach in the same level and subject field, the results collected cannot pretend to any generalisation over all of the Belgian EFL teachers.

## 6.4.2. Section 2: Goals and Norms

Section 2 looked at goals and norms in pronunciation teaching and the results are surprising in regards to the hypotheses (see 6.2.). As far as Q1 ("Quelle norme utilisez-vous majoritairement pour évaluer la prononciation anglaise des élèves ?") is concerned, 27 teachers (47.4%) state that they mostly use BBC English or RP (that were both considered as synonyms in the Google Forms) when they evaluate pronunciation, 27 others (47.4%) claim they do not use any norm when evaluating their pupils' pronunciation, and 3 (5.3%) say they use GA.

One may be perplexed as to how a teacher does not refer to a particular norm to assess a pupil's pronunciation abilities, but Q2 ("Quel but voulez-vous atteindre au niveau de la prononciation anglaise des élèves au terme de l'enseignement secondaire/de promotion sociale ?") may provide hints as to what such a teacher refers to instead. In terms of goals pursued at the end of secondary or further education, a majority of participants (63.2%, n = 36) indicate they aim for intelligible pronunciation, a minority of respondents (35.1%, n = 20) report they strive for a pronunciation without a foreign accent, while 1 respondent (the remaining 1.8%)

<sup>&</sup>lt;sup>51</sup> Thirty-five headmasters from schools in Liège were contacted, all of my private messages were sent to teacher from that province, and Mrs Van Hoof sent the survey to traineeship supervisors who had worked with the university of Liège, hence that issue with representativeness. However, eight schools from the other francophone parts of Belgium were reached out to, and some headmasters even positively replied to my request (i.e., sharing the survey).

says they want their pupils to achieve native-like pronunciation. This data is in line with what surveys on teachers' reported practices put forward (Couper, 2017; Henderson et al., 2012; Timmis, 2002) and especially with what authors plead for, that is, the intelligibility paradigm (Couper, 2021; Diana, 2010; Galante & Piccardo, 2022; Mompean & Fouz-González, 2021).

# 6.4.3. Section 3: Reasons, Beliefs, Opinions and Training in Pronunciation

The results of the first question of Section 3 (Q3) are presented in Table 3. The results are discussed and analysed in groups, if they have the same theme (e.g., age of the learner).

# Table 3

Numbered Item	Pas d d'ac	u tout cord	P d'ac	as cord	D'accord		Tout à fai d'accord	
	n	%	n	%	n	%	n	%
1. J'aime enseigner la prononciation anglaise.	0	0	3	5.3	35	61.4	19	33.3
2. La prononciation est importante en anglais.	0	0	0	0	22	38.6	35	61.4
3. Un natif accorde de l'importance à la prononciation anglaise du locuteur non-natif avec lequel il parle.	2	3.5	14	24.6	32	56.1	9	15.8
4. Enseigner la prononciation anglaise est efficace et améliore la prononciation de l'élève.	0	0	6	10.5	35	61.4	16	28.1
5. Enseigner la prononciation anglaise peut aider à corriger les formes fossilisées.	0	0	6	10.5	42	73.7	9	15.8
6. Les élèves de 12 à 15 ans sont trop jeunes pour apprendre la prononciation anglaise.	40	70.2	16	28.1	1	1.8	0	0
7. Les élèves de 15 à 18 ans sont trop jeunes pour apprendre la prononciation anglaise.	47	82.5	10	17.5	0	0	0	0
8. Les élèves de plus de 18 ans sont trop vieux pour apprendre la prononciation anglaise.	40	70.2	15	26.3	1	1.8	1	1.8
9. Les élèves de plus de 12 ans sont trop vieux pour apprendre la prononciation anglaise.	42	73.7	14	24.6	0	0	1	1.8
10. Seul un natif peut enseigner la prononciation anglaise.	29	50.9	24	42.1	3	5.3	1	1.8

*Percentages of Agreements With Statements About Pronunciation Teaching* Q3) Veuillez indiquer votre degré d'accord avec les affirmations suivantes.

	Pas d	Pas du tout		Pas		D'accord		Tout à fait	
Numbered Item	d'ac	cord	d'ac	cord	Dat	coru	d'ac	cord	
	n	%	n	%	n	%	n	%	
11. Avoir une prononciation anglaise									
correcte, sans avoir de connaissances	10	175	20	40.1	17	20.0	2	25	
sur la prononciation anglaise, suffit	10	17.5	28	49.1	1/	29.8	2	3.5	
pour faire acquerir une bonne									
12 Le me sons préneré à enseigner le									
repondiction angleise per mes divers	4	7	20	25 1	28	40.1	5	00	
séjours à l'étranger	4	/	20	55.1	20	49.1	5	0.0	
13 Je me sens prénaré à enseigner la									
prononciation anglaise par ma									
formation initiale en langue en haute	2	3.5	9	15.8	35	61.4	11	19.3	
école/à l'université.									
14. Je me sens préparé à enseigner la									
prononciation anglaise par ma	0	150	17	20.9	25	12.0	6	10.5	
formation initiale en didactique reçue	9	15.8	1/	29.8	25	45.9	0	10.5	
en haute école/à l'université.									
15. Je me sens préparé à enseigner la									
prononciation anglaise par ma	19	33.3	18	31.6	19	33.3	1	1.8	
formation initiale en littérature reçue à	17	0010	10	0110	17	0010	•	110	
l'université.									
16. Je me sens préparé à enseigner la									
prononciation anglaise par ma	24	40.1	01	26.9	11	10.2	1	1.0	
formation en secondaire (pas par ma	24	42.1	21	30.8	11	19.3	1	1.8	
supérieur)									
17 Je suis satisfait de ce que i'ai									
appris sur les concepts théoriques									
relatifs à la prononciation anglaise	5	8.8	14	24.6	28	49.1	10	17.5	
dans ma formation initiale en langue	C	010		2.110	_0	.,,,,,	10	1110	
et/ou en linguistique.									
18. Je suis satisfait de ce que j'ai									
appris sur les techniques et procédés									
relatifs à l'enseignement de la	9	15.8	20	35.1	23	40.4	5	8.8	
prononciation anglaise, dans ma									
formation initiale en didactique.									
19. Mes élèves sont majoritairement									
motivés lorsque j'enseigne la	5	8.8	21	36.8	29	50.9	2	3.5	
prononciation anglaise.									
20. Une exposition a un input anglais									
indiananachla nour qua las álàvas	0	0	5	00	25	12.0	27	17 1	
aient une bonne prononciation	0	0	5	0.0	25	43.9	21	47.4	
anglaise									
21. Un enseignement théorique de la									
prononciation anglaise est	~		~ ~			10			
indispensable pour que les élèves	8	14	22	36.6	23	40.4	4	7	
aient une bonne prononciation.									

Numbered Item	Pas du tout d'accord		Pas d'accord		D'accord		Tout à fait d'accord	
	n	%	n	%	n	%	n	%
22. Un enseignement pratique de la prononciation anglaise (divers exercices) est indispensable pour que les élèves aient une bonne prononciation.	0	0	3	5.3	29	50.9	25	43.9
23. Les dictionnaires électroniques fournissant une version sonore de la prononciation (Word Reference ou Cambridge Dictionary) ont rendu l'enseignement de la prononciation anglaise obsolète.	30	52.6	22	38.6	4	7	1	1.8

**6.4.3.1. General Reasons for Focusing on Pronunciation**. A first look at the results may indicate reasons why teachers focus (or not) on pronunciation. In general, the teachers surveyed seem to like teaching pronunciation (94.7%, n = 54) (Item 1), and seem to believe in its efficiency (89.5%, n = 51) (Item 4), even when it comes to improving fossilised forms (89.5%, n = 51) (Item 5), which is actually attested by research (Derwing & Munro, 2014), and pronunciation teaching is assumed to still be relevant, in spite of electronic dictionaries providing audio files (91.2%, n = 52) (Item 23). The instructors then enjoy teaching that subject, and their pupils are said to be mostly motivated (54.4%, n = 31), but it is not so for 45.6% (n = 26) of the respondents (Item 19).

They all find it an important component of the English language (100%, n = 57) (Item 2), but, if everyone values it, more than one fourth of the respondents (28.1%, n = 16) do not think that a NS values the English pronunciation of the NNS to whom they are talking (Item 3). Although there does not appear to be any data available on pronunciation tolerance depending on the L1, some of the researchers' conclusions may indeed imply NSs are more tolerant towards non-native pronunciation than NNSs are. It has been demonstrated that natives rely more on the context than on the actual sounds (which is what non-natives do) to decode an utterance (Jenkins, 2002, p. 89), showing pronunciation mistakes do not always impede the understanding of a NS. Research also highlights the prestige natives grant some accents (Beebe & Giles, 1984; Derwing & Munro, 2009). Lastly, Timmis, (2002, p. 243) noticed in her questionnaire that there are more NS teacher who want their learners to achieve accented intelligibility than NNS teacher, pointing to a greater accent tolerance from NSs.

**6.4.3.2. Learners' Age**. As far as the learner's age is concerned, the overwhelming majority of respondents (98.2%, n = 56 for Item 6; 100%, n = 57 for Item 7) have the same opinions, that is, pupils in lower secondary (from 12 to 15 years old) and in upper secondary education (from 15 to 18 years old) are not too young to learn English pronunciation (Items 6 and 7). In addition, almost all of the participants (96.5%, n = 55 for Item 8; 98.2%, n = 56 for Item 9) think that 12- and 18-year-olds are not too old to learn pronunciation (Items 8 and 9). In other words, there does not seem to exist, in the participants' minds, a critical period after which learning English pronunciation becomes useless. Research stresses this as well: age does not play a major role in acquiring intelligible pronunciation, but only comes into play if one wants to achieve native-like pronunciation (Ioup, 2008).

**6.4.3.3. Training**. The respondents were asked on the kind of training they received, and especially on how prepared they felt to teach pronunciation, and on how satisfied they are with what they learned after high school. Most (66.6%, n = 38) agree that they are content with what they learned about pronunciation theory, which means that 33.3% (n = 19) disagree or totally disagree (Item 17). The opinions are less clear-cut when it comes to training in pronunciation methodology: 49.2% of the practitioners (n = 28) agree that they are content with what they learned about pronunciation teaching in language teaching methodology, while 50.9% (n = 29) disagree or totally disagree (Item 18). These numbers can be explained by the fact that some universities offer one general methodology class only, like it is the case with the university of Liège (Université de Liège, 2023), opposed to some universities which offer one methodology class per language, like it is the case with Université Libre de Bruxelles, 2022) and with Université catholique de Louvain (Université catholique de Louvain, 2023).

These numbers are reflected elsewhere, as 54.4% of the teachers surveyed (n = 31) agree that they feel prepared to teach pronunciation thanks to their language teaching methodology training, whereas 45.6% (n = 26) disagree or totally disagree (Item 14). English language classes (see above) seem to score better as to teaching preparation: 80.7% (n = 46) feel these classes prepared them well enough, but 19.3% (n = 11) do not feel they did (Item 13). Nevertheless, literature classes do not score as much, since a bit less than two thirds of the participants (64.9%, n = 37) disagree that they prepared them to teach pronunciation, which still means that a bit more than one third of the participants (35.1%, n = 20) think that literature classes did prepare them to teach pronunciation (Item 15). A majority of respondents (78.9%, n = 45) do not think that secondary education got them ready to teach pronunciation, but this is

not true for 21.1% (n = 12) of them (Item 16). On the question of the impact of staying abroad, the panel appears to be divided, but there is a slight tendency towards a positive influence: 57.9% (n = 33) claim they do feel prepared to teach pronunciation thanks to their stays abroad, while 42.1% (n = 24) claim that they do not (Item 12).

**6.4.3.4.** Natives and Input. As to who can teach pronunciation and the way it should be dealt with, the responses suggest that teachers do not think only a NS is able to teach pronunciation, but some (7.1%, n = 4) think it is the case (Item 10). These opinions also come up in Item 11: two thirds of the respondents (66.7%, n = 38) do not think that pronouncing English correctly without theoretical background is enough to lead to correct pronunciation acquisition (Item 11). Yet, one third of the respondents (33.3%, n = 19) agree or totally agree that input is enough. Those respondents have different opinions than researchers, who put forward that sheer input is not sufficient to improve pronunciation (Iverson et al., 2012; Levis et al., 2016; Saito & Lyster, 2012).

However, the results regarding Item 11 do not entail that sufficient, correct, oral input is of secondary importance to ensure correct pronunciation, for an overwhelming majority (91.2%, n = 52) agrees that such input is needed (Item 20). Input then seems to make up one needed facet of pronunciation teaching, and so does practice as 94.7% (n = 54) believe that practice is indispensable (Item 22), but the importance of theory is much discussed and does not seem to be really needed: 47.4% (n = 27) reckon it is indispensable to ensure correct pronunciation, while 50.6% (n = 30) disagree that it is (Item 21).

These results are more or less in line with what a few researchers have concluded. Sheer input, as explained above, is not enough to improve one's pronunciation (Iverson et al., 2012; Levis et al., 2016; Saito & Lyster, 2012), but remains important in helping pupils producing contrasts they may not have in their L1 (Simon et al., 2014, 2016). Then, bringing the learners' attention to sounds, through theory for example, is linked to enhanced comprehensibility (Venkatagiri & Levis, 2007), and is overall the first step in acquisition (Schmidt, 2001, pp. 26, 29–31; Schmidt, 1990, p. 149; VanPatten, 2002, p. 108). In other words, a few teachers, when contrasted to research, appear to value input a bit too much and appear not to see the added value of pronunciation theory.

**6.4.3.5.** Activities During Teacher Training. The results on how satisfied teachers are with their training strike one as mixed: some teachers are happy with what they learned in general English classes, and others are dissatisfied with their training in pronunciation

methodology (see Item 13 and Item 14). Question 4 and Q5 give insights into the activities both sorts of training involved. Question 4 is presented in Figure 4.

### Figure 4

*Linguistic Pronunciation Activities Done During Teacher Training* Q4) Durant votre formation initiale en langue et/ou en linguistique pour devenir professeur, quelle(s) activité(s) liée(s) à la prononciation anglaise avez-vous effectuée(s) ?



Number of Respondents

Two activities come up as most common in language/linguistic training: learning IPA (80.7%) and transcribing phonetic symbols into English words (80.7%). Not as frequent but still so is another type of phonetic transcription, that is, from oral English input to phonetic symbols (75.4%). The last quite usual activity is learning a phonetic notation (70.2%); what phonetic notation involves was not detailed, and may thus refer to various kinds of phonetic notations (e.g., ToBI, keywords). Then, there is even a chance that some respondents selected both the item regarding IPA and the one on phonetic notations, but that, to them, they meant the same. The last activity a small majority of participant (52.6%) say they did is exercises on prosody or suprasegmentals. A minority (47.4%) say they learned theory on the physical properties of segments and of intonation.<sup>52</sup> Finally, the least common activity is analysis of these physical properties (24.6%). The precise results are quite varied as to who did what, as, for example, some respondents selected as few as one exercise (e.g., learning IPA, or transcribing symbols to written English).

<sup>&</sup>lt;sup>52</sup> In Q4, "suprasegmentals" or "prosody" should have been used instead of "intonation", since suprasegmentals include intonation, but the reverse is not true. One may still argue that "intonation" remains a clearer term to teachers.

As far as language teaching methodology is concerned, examining Q5 reveals that one activity was selected by a large majority of participants. The results are presented in Figure 5.

### Figure 5

Pronunciation Activities Done in Language Teaching Methodology Q5) Durant votre formation initiale en didactique pour devenir professeur, quelle(s) activité(s) liée(s) à la prononciation avez-vous effectuée(s) ?



Number of Respondents

Clearly, the data suggest that one of the only quite common methodology activity linked to pronunciation is comments from trainee supervisors on pronunciation (80.7%). More than half of the respondents (59.6%) say their own university teachers<sup>53</sup> commented on their pronunciation during the traineeships. A minority of the respondents (38.6%) say they learned pronunciation teaching techniques specific to English, whereas 15.8% say they learned such techniques for all languages (see 6.4.3.3. for comments on universities and English classes).

Question 6 ("Dans le cadre de la formation continue, avez-vous participé à des journées pédagogiques sur l'enseignement de la prononciation anglaise ?") asked whether the respondents had ever taken part in pronunciation teaching training courses during in-service training and all said they had not, in spite of a clear demand, since 68.4% of the participants (n = 39) say they seek such courses (Q7, "Êtes-vous demandeur de journées pédagogiques sur l'enseignement de la prononciation anglaise ?"). This demand matches what other surveys found out, that is, teachers are willing to spend more time learning on pronunciation (teaching) (Couper, 2017; Foote et al., 2012), but the same surveys indicate that continuous learning did provide instructors with the missing information.

<sup>&</sup>lt;sup>53</sup> In Q5, the last item should have included "didacticiens et professeurs de langues" to cater to every respondent, that is, those who went to university and those who went to college.

The last question of Section 3 was about when one should start teaching pronunciation. A clear tendency arose out of Q8 ("Quand pensez-vous qu'il faut commencer à enseigner la prononciation anglaise dans le cours de langue anglaise 1 ?"): most participants (66.7%, n = 38) reckon one should start early, that is, in the fifth year of primary education. A bit more than one fourth (26.3%, n = 15) think one ought to begin in the first year of secondary education, and few (7%, n = 4) believe one should commence in the second stage of secondary education. The final stage of secondary education was not selected once.

# 6.4.4. Section 4: Contents

Section 4 started with Q9 ("Quel aspect de la prononciation anglaise travaillez-vous le plus en classe ?") and the answers are unanimous. In class, every teacher surveyed focuses more on practice than on theory and this matches the greater importance they give practice over theory (see 6.3.2.1.). More interesting is which pronunciation elements teachers most focus on. This was looked at through Q10, Q11 and Q12 (see Figure 6, Table 4 and Table 5).

#### Figure 6





Number of Respondents

Overall, the data suggest that segments (i.e., vowel sounds, consonant sounds and diphtongs) are what most teachers focus on. When looking at all the individual answers, only one respondent did not select either vowel sounds, consonant sounds or diphtongs (they indicate that they teach word stress only). Yet, this does not mean that suprasegmentals are never seen in class; as the results show, they are all addressed in varying degrees. The most taught suprasegmental is word stress (73.7%), followed closely by intonation (70.2%). A minority of respondents (38.6%) indicate that they teach rhythm and en even smaller amount of participants (35.1%) state they teach sentence stress. A deeper look at the data shows that 21.1% of the surveyed (n = 12) selected all options; other large tendencies include teaching vowel, consonant

sounds, and diphtongs, with either intonation only (12.3%, n = 7), with lexical stress only (10.5%, n = 6), or with both intonation and lexical stress (14%, n = 8).

Question 11 provided more details as to what teachers find of greatest importance. As mentioned in the methodology (see 6.3.2.), Q11 is peculiar with regard to its modalities: the respondents had to select three pronunciation elements and rank them in terms of how important they are in their class (see Table 4).

## Table 4

#### Importance of Pronunciation Elements in EFL Classes

Q11) Veuillez sélectionner un chiffre en fonction de l'importance relative que vous accordez, en tant qu'enseignant, à chacun des éléments suivants dans votre cours d'anglais (1 = le moins important pour moi ; 7 le plus important pour moi). Les sept chiffres doivent être utilisés, et chaque élément doit avoir un chiffre différent.

Itom			Re	lative Imp	ortance		
Item	1	2	3	4	5	6	7
Sons voyelles	11	0	8	2	9	12	15
Sons consonnes	3	15	3	11	8	12	5
Diphtongues	0	2	12	7	15	13	8
Intonation	0	2	19	16	5	10	5
Rythme	7	21	9	7	5	5	3
Accent tonique	6	7	3	9	13	2	17
Accent de phrase	30	10	3	5	2	3	4

The chosen way to analyse Table 4 is to add the values from the first and second columns, and from the sixth and seventh columns. The three items that have the highest number in each category (i.e., first and second columns, sixth and seventh columns) can then be considered as the least and as the most important items. Columns 3, 4 and 5 are then not taken into account because the results for these columns may be subject to error of central tendency (see 6.3.3.).

With this method, the three least important pronunciation elements appear to be: sentence stress (70.2%, n = 40), rhythm (49.1%, n = 28) and consonant sounds (31.6%, n = 18). Sentence stress seems to be the undisputed least important element as 30 teachers put it at the bottom of the scale, which is the highest number any item received. In addition, the three most important items appear to be: vowel sounds (47.4%, n = 27), diphthongs (36.8%, n = 21) and lexical stress (33.3%, n = 19). Despite being third among the three most important elements, lexical stress received the most votes for the top of the scale (two more than vowel sounds and seven more than diphthongs); the importance teachers give lexical stress corresponds to the

importance lexical stress has according to the researchers: listeners heavily rely on word stress to understand one's speech (Hahn, 2004; Zielinski, 2008).

Yet, two other items closely follow lexical stress in terms of importance: consonant sounds (29.8%, n = 17) and intonation (26.3%, n = 15). This shows that consonant sounds are given variable levels of important, as 18 teachers and 17 others consider them of either low importance (see Columns 1 and 2) or of rather high importance (see Columns 6 and 7).

Question 12 pertained to the same pronunciation items, but asked the participants to take three of them and rank them in terms of how problematic they are for the pupils. The results are presented in Table 5.

#### Table 5

#### Level of Problems Pronunciation Elements May Pose

Q12) Quels éléments posent le plus de problèmes à la majorité de vos élèves ? Veuillez en sélectionner 3 en fonction du niveau de problème posé pour vos élèves (1 = pose des problèmes, mais moins que le 2 ; 2 = pose des problèmes mais moins que le 3 ; 3 = pose le plus de problèmes).

			Level of	Problem		
Item		1		2		3
	n	%	n	%	n	%
Sons voyelles	13	22.8	11	19.3	9	15.8
Sons consonnes	11	19.3	4	7	6	10.5
Diphtongues	15	26.3	13	22.8	9	15.8
Intonation	6	10.5	7	12.3	5	8.8
Rythme	5	8.8	8	14	5	8.8
Accent de mot	6	10.5	8	14	20	35.1
Accent de phrase	1	1.8	6	10.5	3	5.3

To do away with error of central tendency (see 6.3.3.), exclusively the two ends of the scale are analysed. The most problematic element (see Column 3) is by far lexical stress, as it received more than double the votes of any other feature in the third column. On the other hand, diphthongs appear to be ranked first in the first level of difficulty, closely followed by vowel and consonant sounds. Segments thus appear to be problematic for pupils: in total, 68.4% of the respondents (n = 39) indicate that segments are somewhat difficult (see Column 1). Besides, 24 teachers (42.1%) report segments are the most problematical elements for their pupils. When comparing this data to what researchers found, stress is indeed considered to be problematical for francophone learners (Dupoux et al., 1997; Frost, 2011; Walter, 2001, p. 200), and so are

some segments (Walter, 2001). In others words, the difficulties teachers report match the ones reported by researchers.

Question 13 resembles Q12 in terms of answer modality, so it will be analysed in a similar manner (see Table 6). For Q13, the participants had to select three criteria that help them select what they teach with regard to pronunciation. Just like in Q12, 1 equals to the least important of the three and 3 to the most important. After rereading Q13 and its guideline, the question and more precisely what 1 and 2 represent seem to suffer from a lack of details. Contrary to what is done with Q12, the scale is not as thoroughly detailed, resulting in unclear definitions. Indeed, the intended meaning of 1 is the least important criterion among three important criteria, but it could have been interpreted as the least important criterion out of the proposed items.<sup>54</sup> Therefore, the first level of importance, as well as the second for both clarity and central tendency reasons, are not analysed here.

## Table 6

Importance of Criteria to Select Pronunciation Elements to Teach

Q13) Sur la base de quel(s) critère(s) sélectionnez-vous les éléments de prononciation anglaise que vous enseignez ? Veuillez en sélectionner 3 en fonction de leur importance (1 = le moins important ; 3 = le plus important).

		In	iportai	nce Lev	el	
Item	1	-	2		3	3
	n	%	n	%	n	%
Une liste d'erreurs fréquentes commises par mes élèves	9	15.8	10	17.5	30	52.6
Mon intuition de ce qui est compliqué pour les élèves francophones	16	28.1	24	42.1	7	12.3
Des éléments identifiés par des ouvrages de ressources (comme <i>Pronunciation in Use</i> ) comme compliqués pour les élèves francophones	13	22.8	4	7	9	15.8
Des éléments identifiés par la recherche comme compliqués pour les élèves francophones	3	5.3	6	10.5	2	3.5
Le concept de charge fonctionnelle	2	3.5	2	3.5	3	5.3
Des éléments identifiés par mon cours de didactique comme compliqués pour les élèves francophones	2	3.5	4	7	1	1.8
Mes propres difficultés en prononciation quand j'étais élève	12	21.1	7	12.3	5	8.8

<sup>&</sup>lt;sup>54</sup> To ensure clarity, Q13 should have been as detailed as Q12, and the guidelines should then have been: "1 =important, mais moins que le 2"; "2 =important, mais moins que le 3"; "3 =le plus important."

"Une liste d'erreurs fréquentes commises par mes élèves" is by far the most important criterion to select pronunciation elements. Two other criteria were selected by more than 10% of the respondents, that is, "Mon intuition de ce qui est compliqué pour les élèves francophones" and "Des éléments identifiés par des ouvrages de ressources (comme *Pronunciation in Use*) comme compliqué pour les élèves francophones." The first selection criterion (i.e., the list of frequent errors), corresponds to what literature advocates (Couper, 2021), but the one research appears to emphasise a lot, that is, functional load (Brown, 1988; Derwing & Munro, 2014; King, 1967; Munro & Derwing, 2006), does not appear to be often called upon by the panel.

The last two questions of Section 4 revolved around phonetic notations and IPA. Question 14 was more general and asked about the most frequently used method to teach pronunciation: the results are presented in the following pie chart (see Figure 7).

## Figure 7

Method Used to Teach Pronunciation Q14) Pour enseigner la prononciation anglaise, quel type de méthode utilisezvous **le plus fréquemment** ? • L'alphabet phonétique international

- Des mots clés anglais courts pour illustrer des sons (cat, ship, put, etc.)
- Des symboles phonétiques simplifiés
- Mon propre alphabet phonétique

The majority of participants (50.9%) claim they most often use short English words, such as "cat" or "ship" to teach pronunciation. IPA remains a relevant way to teach pronunciation as it is claimed to be often used by 28.1% of the teachers surveyed. Besides, IPA and keywords alike have proved their efficiency (Fouz-González & Mompean, 2021), and the alphabet proved it is well considered by students (Mompean & Lintunen, 2015). Finally, the rest of the panel states they either make use of simplified phonetic symbols (17.5%) or of their own phonetic alphabet (3.5%), whatever it may be.

The next question zoomed in on IPA specifically and confronted the respondents with several items. The different statements from Q15 and how many respondents (dis)agreed, and to what extent are presented in Table 7.



## Table 7

## Percentages of Agreements With Statements on IPA

Q15) Veuillez indiquer votre degré d'accord avec les affirmations suivantes. Elles sont en rapport avec l'alphabet phonétique international (API) dans le cours d'anglais langue étrangère.

Numbered Item		Pas du tout d'accord		as cord	D'accord		Tout à fai d'accord	
	n	%	n	%	n	%	n	%
1. J'utilise l'API avec mes élèves.	12	21.1	15	26.3	22	38.6	8	14
2. Je familiarise mes élèves à l'API.	8	14	13	22.8	24	42.1	12	21.1
3. L'API est trop complexe pour mes élèves.	4	7	23	40.4	23	40.4	7	12.3
4. L'API est trop complexe pour moi.	35	61.4	16	28.1	4	7	2	3.5
5. J'ai été familiarisé avec l'API durant ma formation en secondaire en anglais.	25	43.9	14	24.6	15	26.3	3	5.3
6. J'ai été familiarisé avec l'API durant ma formation en anglais en haute école/à l'université.	5	8.8	2	3.5	15	26.3	35	61.4
7. J'ai le temps de voir l'API en classe avec mes élèves.	20	35.1	23	40.4	12	21.1	2	3.5
8. J'estime que l'enseignement de l'API aux élèves est superflu car tous les dictionnaires en ligne proposent une version audio du mot recherché.	16	28.1	28	49.1	11	19.3	2	3.5

After reviewing the numbers, it appears that most respondents (52.6%, n = 30) agree or totally agree that they use IPA in class with their pupils, so quite a few teachers (47.4%, n = 27) do not seem to use IPA in class (Item 1). Yet, if only a small majority uses IPA, a large majority (63.2%, n = 36) introduces IPA to their class (Item 2). Looking at literature and at how little IPA-based exercises are used in class (see Q16 in 6.4.4.), one may guess why teachers use or introduce IPA for: making sure feedback is precise (Fouz-González & Mompean, 2021; Mompean & Fouz-González, 2021), presenting pupils with means to express themselves (Fraser, 2010), or with means to learn on their own (Marks, 2011; Mompean & Fouz-González, 2015).

Reasons that may explain why IPA is not used by all is the supposed complexity of IPA: 52.7% of the participants (n = 30) do think that IPA is too difficult for their pupils (Item 3). Yet, if there are more positive answers (i.e., [totally] agree) than negative ones (i.e., [totally] disagree) for Item 3, the same number of respondents (40.4%, n = 23) agree and disagree (40.4%, n = 23) that IPA is too complex for their classes (Item 3).

Another reason explaining why IPA is not taught by all is time constraints, as 75.5% of the instructors surveyed (n = 43) (totally) disagree that they have time to see IPA in class (Item 7). On that matter, authors (Bowen, 1972; Derwing, 2010; Field, 2005; Marks, 2011) encourage teachers to combine pronunciation (IPA in this case) with, among others, vocabulary lessons in order to solve timing problems (see 6.4.6.). Finally, online dictionaries that provide audio files do not appear to dissuade teachers from introducing IPA, since most (80.7%, n = 46) (totally) disagree that these resources turned IPA teaching useless (Item 8).

Turning to the teachers themselves and their relation to this type of phonetic notation, a great share of participants do not reckon it is too difficult for themselves (89.5%, n = 51) (Item 4), maybe because they got introduced to it during their secondary education (31.6%, n = 18) (Item 5), or because they learned about it after their secondary education (87.7%, n = 50) (Item 6). It indeed appears that pronunciation training in college or in university revolved around IPA for quite a few teachers (see 6.4.3.5.).

### 6.4.5. Section 5: Material and Activities

Section 5 took a deeper look at some precise activities used in teaching pronunciation. The first question of the section (Q16) asked the participants to rank three activities in terms of frequent use, with 1 being the least frequent and 3 the most frequent (see Table 8). After rereading the question and its guideline, there is a high chance that the participants understood the question differently: "1 = 1 a moins fréquente" may have been interpreted as "an activity that is never used", or as "an activity that is quite often used, but less so than 2." This latter explanation was the one I intended, but I may have failed to put it clearly.<sup>55</sup> Because the meaning of 1 I wanted to put across and the respondents' interpretation of 1 may be different, the first and second level of frequency are not analysed; besides what frequency 2 means partly depends on what 1 is (e.g., 2 is a more or less frequent activity, 2 is a frequent activity but less so than 3). Therefore, solely the third column is analysed.

<sup>&</sup>lt;sup>55</sup> Question 16 can see its clarity improved in similar manner to Q13, since the meaning of 1 and 2 need to be detailed: "1 = fréquente, mais moins que la 2"; "2 = fréquente, mais moins que la 3"; "3 = la plus fréquente."

# Table 8

#### Frequent Pronunciation Activities in EFL Classes

Q16) Quelle(s) activité(s) utilisez-vous pour enseigner la prononciation anglaise ? Veuillez en sélectionner 3 en fonction de leur fréquence d'utilisation (1 = la moins fréquente ; 3 = la plus fréquente).

		F	requen	cy Leve	l	
Item		1		2	,	3
	n	%	n	%	n	%
Des répétitions en chœur	16	28.1	2	3.5	3	5.3
Exercices de discrimination	2	3.5	7	12.3	6	10.5
Activités d'expressions orales	4	7	13	22.8	10	17.5
Exercices d'imitation d'un modèle	9	15.8	4	7	4	7
Des transcriptions phonétiques (symboles -> mots écrits en anglais)	2	3.5	2	3.5	3	5.3
Des transcriptions phonétiques (mots écrits -> symboles phonétiques)	1	1.8	0	0	1	1.8
Des transcriptions (input oral -> mots écrits en anglais)	0	0	3	5.3	0	0
Clarification par l'enseignant d'un point de prononciation	4	7	10	17.5	6	10.5
Théâtre	0	0	3	5.3	1	1.8
Lecture à voix haute	7	12.3	4	7	18	31.6
Activité de répétition individuelle après un feedback du professeur	4	7	6	10.5	4	7
Virelangue (ex. Les chaussettes de l'archiduchesse sont-elles sèches ?)	5	8.8	1	1.8	1	1.8
Visualisation des sons et des accents via des ondes sonores	2	3.5	0	0	0	0
Exercices se basant sur les capacités physiques	0	0	2	3.5	0	0

Looking at the third level of frequency, one activity appears to be quite often used: "Lecture à voix haute" (31.6%). Another common exercise is "Activités d'expressions orales" (17.5%), so are "Exercices de discrimination" (10.5%), and "Clarification par l'enseignant d'un point de prononciation" (10.5%). This data corresponds to previous surveys whereby minimal pairs, perception exercises, oral performances and aloud readings were found to be usual classroom practices (Foote et al., 2012, pp. 12–13; Henderson et al., 2015, p. 275).

Interestingly (see 6.5.2. for more details), although teachers say they use IPA and although they learnt about it in college or university, transcriptions are among the least frequently used activities. The perception side is preferred in any case: more often, the pupils have to decipher symbols to then write words or sentences down for example. In addition, in

spite of a stated preference for the practical side (see Q9 in 6.4.4.), explaining or clarifying a pronunciation point seems common practice.

Speaking of clarifying (assumed to be explicit), Q17 pertained to how explicit or implicit the instructors' approach to teaching pronunciation is (see Table 9).

## Table 9

### Explicit and Implicit Pronunciation Teaching

Q17) Veuillez indiquer votre degré d'accord avec les items suivants. Lorsque j'enseigne la prononciation anglaise, je l'enseigne principalement ...

Numbered Item	Pas du tout d'accord		Pas d'accord		D'a	cord	Tout à fait d'accord	
-		%	n	%	n	%	n	%
1. de manière explicite.	2	3.5	7	12.3	29	50.9	19	33.3
2. de manière implicite.	7	12.3	18	31.6	23	40.4	9	15.8

If the results seem to be quite clear for "de manière explicite" since most of the answers (86%, n = 48) are positive (i.e., "agree" or "totally agree"), it is less categorical for implicit teaching: 43.9% of the instructors (n = 25) (totally) disagree that they mainly teach pronunciation implicitly, while 54.4% (n = 31) (totally) agree that they do. Combining both sets of data, teachers appear to mix the approaches, but remain more explicit than implicit, and this approach does not, at first, match the most used activities (see 6.5.1. for further analyses). The teachers' practice then dovetail with the practice encouraged by some (Hu et al., 2013, p. 371; Venkatagiri & Levis, 2007, p. 273) who found oud explicit theoretical knowledge is linked is beneficial to improved pronunciation (see 3.3.3. and 4.5.).

It could be argued that exercises make up the one part of teaching, with the previous necessary part being presenting the subject first, and this is what Q18 looked at (see Figure 8).







Two exercises appear to be quite common to introduce a new pronunciation element as more than 60% of the participants report they use them: "Une comparaison avec un autre élément anglais pour avoir un contraste (sheep/ship)" (89.5%) and "Illustration d'un son via une exagération et un ralentissement de ce son, par exemple, en montrant la position de la langue dans la production du son (three teeth)" (61.4%).

Reviewing the detailed results, that is, which combination got selected most, three combos come up repeatedly: "Une comparaison avec un autre élément anglais pour avoir un contraste (sheep/ship)" alone (24.6%, n = 14), the previous item with "Illustration d'un son via une exagération et un ralentissement de ce son ..." (22.8%, n = 13), and finally both of the above-mentioned elements along with "Des images, dessins, vidéos (ex. positionnement entre les dents)" (10.5%, n = 6). Any other processes association got chosen by less than 8.8% of the respondents (or less than five times) and the precise results are therefore not analysed any further. Eventually, comparing two English sounds is obviously a perception-based activity, and such exercises are strongly recommended by various authors (Couper, 2021; Derwing & Munro, 2014; Setter & Jenkins, 2005; Thomson, 2012), and in presenting a new pronunciation element through a comparison undoubtedly makes the learner aware, or notice the sound difference, thus teachers take into account the influential (VanPatten, 2002) "noticing hypothesis" (Schmidt, 2001, p. 26).

Question 19 questioned the participants about the type of resources they might employ to teach pronunciation features (see Figure 9).

## *Resources Used to Teach Pronunciation* Q19) Quel(s) type(s) de manuel(s), d'ouvrage(s) de ressources et/ou de ressource(s) utilisez-vous pour enseigner la prononciation anglaise en classe ?





As can be seen with the graph, there is not one resource that a majority of participants claim to use, but three were selected by more than or by close to 30% of the participants: "Manuel classique" (35.1%), "Site internet dédié à la prononciation" (31.6%) (a technological mean endorsed by authors [Levis, 2007; Rogerson-Revell, 2021]), and "Poster/tableau mural ..." (29.8%). It thus looks like phonetic symbols are physically present in the classroom, even if they may not be taught (see 6.4.4.). Finally, quite a few respondents (33.3%) do not make use of any of the mentioned resources.

The instructors who had selected any of the options (except the last one), were asked to write the name of the resource they use (Q20, "Si vous utilisez un manuel, un ouvrage de ressource, ou une autre ressource, lequel/lesquelles utilisez-vous ?"). The ones cited more than once<sup>56</sup> are: (*New*) English File (n = 5), Get Up (n = 3), Pronunciation in Use (n = 3).

The next set of questions revolved around giving feedback in two specific classroom contexts: spontaneous speech acts and aloud reading. The first pair of questions (Q21 and Q22) looked at the moment of the feedback (see Figure 10 and Figure 11).

<sup>&</sup>lt;sup>56</sup> The resources that were cited once only are: the online dictionary *Word Reference, Jelly 2* and its pronunciation exercises, *Vitaeducation.org, Choices, Gateway* B1+, a dictionary including pronunciation, *TTsdemo, NA, New Headway Advanced, New Inside Out Intermediate, Step Up. English File* was cited three times, whereas *New English File* was cited twice. *Get Up 6* was mentioned once.

### Figure 11

Moment of Feedback During Spontaneous Speech Acts

Q21) Lorsqu'un élève fait une erreur de prononciation durant une **intervention orale spontanée** en cours d'anglais langue étrangère (expression d'un avis personnel, par exemple), que faites-vous ?



Je donne un feedback discret immédiatement (reformulation), mais j'y reviens explicitement après.

Moment of Feedback During Aloud Readings Q22) Lorsqu'un élève fait une erreur de prononciation durant une **lecture à voix haute** en cours d'anglais langue étrangère (d'un paragraphe d'un texte ou d'une consigne, par exemple), que faites-vous ?



Je donne un feedback discret immédiatement (reformulation), mais j'y reviens explicitement après.

The results suggest one moment or type of feedback is common to both spontaneous speech acts and aloud readings: postponed feedback (31.6% for Q21; 24.6% for Q22). Yet, how immediate feedback is seems to differ substantially based on the activity because immediate correction appears to be unusual practice when a pupil is spontaneously speaking (19.3%), while immediate correction appears to be usual when a pupil is reading aloud (49.1%); this is in accordance with feedback habits in VTM (De Vriendt, 2000a, p. 251). What is more, discrete recasts (i.e., reformulation in the item) followed by explicit correction seem common during spontaneous speech (49.1%) but do not seem so common during aloud readings (26.3%). In other words, one practice (i.e., postponing feedback) seems frequent whatever the exercise, while there appear to exist preferences for two other practices (i.e., immediate correction and discrete recasts followed by explicit correction).

If how immediate feedback is varies depending on the context, teachers behave more similarly in terms of what they correct (Q23 [see Figure 12] and Q24 [see Figure 13]).

Mistakes Corrected During Spontaneous Speech Acts

Q23) Lorsqu'un élève fait des erreurs de prononciation durant une **activité d'expression orale spontanée** au cours d'anglais langue étrangère, que faites-vous ?

*Mistakes Corrected During Aloud Readings* Q24) Lorsqu'un élève fait des erreurs de prononciation durant une **lecture à voix haute** au cours d'anglais langue étrangère, que faites-vous ?



In both contexts (i.e., spontaneous speech acts and aloud readings), a majority of respondents (56.1% for Q23; 61.4% for Q24) state they correct the main pronunciation mistakes. Intelligibility mistakes are taken more into consideration during spontaneous speech (31.6%), than during aloud readings (10.5%). Only a few instructors (12.3%) correct every pronunciation mistake during speech acts, but 28.1% of the participants claim they do indeed focus on every mistake when a pupil is reading aloud.

Recasts and explicit correction seem to constitute important types of feedback at least in two contexts. Question 25 (see Table 10) looked at the frequency of other types of feedback to correct a pronunciation mistake (there was not any context detailed). In spite of defining six different feedbacks in the Google Forms (i.e., explicit correction, recasts, clarification requests, metalinguistic feedback, elicitation, repetition), only five sorts of feedback (plus an item "I do not correct the mistake") were available in Q25: this is purely a mistake for I forgot to add "explicit correction."

## Table 10

# Frequency of Feedback Types

Q25) Lorsque vous corrigez une erreur de prononciation anglaise, que faites-vous ?

Numbered Item		nais	Parfois		Souvent		Toujours	
Numbered Rem	n	%	n	%	n	%	n	%
1. Demande de clarification	5	8.8	24	42.1	24	42.1	4	7
2. Extraction de l'information	8	14	22	38.6	24	42.1	3	5.3
3. Reformulation par le professeur	0	0	2	3.5	31	54.4	24	42.1
4. Feedback métalinguistique	11	19.3	19	33.3	25	43.9	2	3.5
5. Répétition par le professeur sur un ton indiquant une erreur	12	21.1	8	14	27	47.4	10	17.5
6. Je ne corrige pas l'erreur.	34	59.6	22	38.6	1	1.8	0	0

The data imply that teachers do correct pronunciation mistakes since almost every participant (98.2%, n = 56) says they never let a pronunciation mistake slip through the net (Item 6). Solely 1 respondent (1.8%) states they often do not correct a mistake. As far as the type of feedback is concerned, there does not seem to be a clear preference for any, since "souvent" was selected by most for every feedback. Yet, three pieces of feedback received more negative answers (i.e., either "jamais" or "sometimes") than positive ones (i.e., "souvent" or "always"): clarification requests (50.9%, n = 29), elicitation (52.6%, n = 30), and metalinguistic feedback (52.6%, n = 30). Thus, it would mean that the others, which have received more positive reactions than negative ones, are preferred: the experimentally-attested effective (Saito & Lyster, 2012) recasts (96.5%, n = 55) and repetitions (64.9%, n = 37).

The last feedback-related piece of information that is of interest to this research is what feedback is on and Q26 examined just that (see Table 11).

# Table 11

Focus of Feedback

Q26) Lorsque vous faites un feedback sur la prononciation anglaise d'un élève, sur quel(s) élément(s) le faites-vous porter ?

Numbered Item		nais	Parfois		Souvent		Toujours	
		%	n	%	n	%	n	%
1. Les sons voyelles	0	0	16	28.1	27	47.4	14	24.6
2. Les sons consonnes	3	5.3	19	33.3	22	38.6	13	22.8
3. Les diphtongues	0	0	13	22.8	32	56.1	12	21.1
4. L'intonation	4	7	19	33.3	27	47.4	7	12.3
5. Le rythme	12	21.1	26	45.6	12	21.1	7	12.3
6. L'accent tonique	7	12.3	12	21.1	24	42.1	14	24.6
7. L'accent de phrase	20	35.1	24	42.1	10	17.5	3	5.3

Just like with the different kinds of feedback, the teachers surveyed do not seem to focus more on one particular pronunciation feature. However, some tendencies come up when analysing the results: both rhythm and sentence stress are not often subject to feedback as they received more negative answers than positive ones (66.7%, n = 38 for rhythm; 77.2%, n = 44 for sentence stress). Besides, these are the two elements that most respondents specify they sometimes (i.e., "Parfois") give feedback on. Therefore, the other elements (i.e., vowel and consonant sounds, diphthongs, intonation, lexical stress) are often or always subject to feedback. Interestingly, more than one fifth of the respondents state their feedback is always on segments or on lexical stress (see column "Toujours" for Item 1, Item 2, Item 3 and Item 4). The focus of feedback also naturally matches what teachers claim to most address in class (see 6.4.4.).

One of the last questions of Section 5 (Q27) questioned the respondents about the use of native or non-native English speakers in class, through listening comprehensions (see Table 12).

# Table 12

# Frequency of Accents Used in Listening Comprehensions

Q27) Lorsque vous faites des compréhensions à l'audition en classe d'anglais langue étrangère, quel(s) accent(s) ont **majoritairement** les locuteurs ?

Numbered Item		nais	Parfois		Souvent		Tou	jours
Nullibered Itelli	n	%	n	%	n	%	n	%
1. Un accent natif (anglais, américain, australien, etc.)	0	0	1	1.8	27	47.4	29	50.9
2. Un accent non natif (français, espagnol, allemand, etc.)	17	29.8	39	68.4	1	1.8	0	0

The use of native speakers in class appears to be quite common (98.2%, n = 56) for an overwhelming majority of respondents say they either often or always use native speakers in listening comprehensions; only 1 participants thus says they sometimes make their pupils listen to native accents. This does not entail that non-native accents are nowhere to be found in Belgian EFL classes: 1 teacher (1.8%) says they often make use of NNSs, 39 teachers (68.4%) indicate they sometimes let their pupils listen to NNSs, against 17 (29.8%) who say they never exploit NNSs in class. This exposure to NNSs and consequently to non-traditional varieties of English matches the recommendations of many scholars (Derwing, 2010; Jenkins, 2000; Low, 2021; Marks, 2011; Murphy, 2014).
The final questions of Section 5 referred to the integration of a pronunciation module into the classroom environment at two levels: the teaching sequence (Q28, see Table 13) and the more traditional features of English, like vocabulary or grammar (Q29, see Table 14).

# Table 13

*Frequency of Integration of Pronunciation Modules Into a Teaching Sequence* Q28) Comment les modules de prononciation sont-ils intégrés dans votre enseignement de l'anglais langue étrangère ?

Numbered Item		Jamais		Parfois		Souvent		Toujours	
Numbered Rem	n	%	n	%	n	%	n	%	
1. C'est un module à part, déconnecté du reste des séquences (ex. une heure par mois sur un élément spécifique de la prononciation).	35	61.4	18	31.6	2	3.5	2	3.5	
2. C'est un module intégré dans une séquence didactique en fonction des besoins des élèves relatifs à la matière abordée dans la séquence (ex. prononciation du « ED » dans une séquence sur le past simple).	1	1.8	12	21.1	23	40.4	21	36.8	
3. C'est une séquence entière sur la prononciation.	40	70.2	8	14	5	8.8	4	7	

On the whole, 70.2% of the instructors surveyed declare that they never put up a whole sequence dedicated to pronunciation and 61.4% state that they never put up a pronunciation module, unrelated to previous or future sequences. Additionally, both of these options got more negative answers than positive ones: 93% (n = 53) of "Jamais" and "Sometimes" answers for "module à part, déconnecté du reste des séquences" and 84.2% (n = 48) for "séquence entière sur la prononciation." The latter option thus appears to be more common than the former, but both remain rare. This could be linked to the material teachers claim to use (see Q19 in 6.4.5.): few teachers use a dedicated pronunciation textbook, so this may explain why few teachers create pronunciation-dedicated modules.

Clearly, the teachers' responses point to one preferred way of integrating pronunciation modules into a greater whole: "un module intégré dans une séquence didactique en fonction des besoins des élèves relatifs à la matière abordée dans la séquence." For a great number of teachers, it is quite usual (77.2%, n = 44) that they integrate a pronunciation module into a larger sequence depending on the demanded needs, which is just what many researchers advise doing (Bowen, 1972; Derwing, 2010; Derwing & Munro, 2014; Galante & Thomson, 2017;

Gordon & Darcy, 2016; Kissling, 2013; Levis, 1999; Levis & Grant, 2003; Marks, 2011; Rasier, 2011; Taylor, 2006; Thomson, 2012).

These needs could refer to one of four linguistic elements (i.e., grammar, syntax, vocabulary, and language functions) and Q29 looked at that in detail (see Table 14).

# Table 14

Frequency of Pronunciation Combined With Grammar, Lexis, Syntax and Language Functions

Q29) Avec quoi sont combinées les clarifications et applications de la prononciation anglaise dans votre enseignement ?

Numbered Item		Jamais		Parfois		Souvent		Toujours	
Numbered Rem	n	%	n	%	n	%	n	%	
1. Avec des éléments grammaticaux (ex. prononciation des « ED » avec le past simple).	1	1.8	9	15.8	37	64.9	10	17.5	
2. Avec des éléments syntaxiques (ex. intonation des questions).	7	12.3	25	43.9	22	38.6	3	5.3	
3. Avec des éléments lexicaux (ex. prononciation des mots de vocabulaire).	0	0	8	14	33	57.9	16	28.1	
4. Avec les fonctions langagières (ex. intonation pour marquer un doute).	12	21.1	22	38.6	19	33.3	4	7	

When analysing in terms of positive and negative responses, there seem to exist three tendencies. Most instructors appear more likely to combine pronunciation exercises and explanations with grammar (82.5%, n = 47), or with lexis (86%, n = 49). Moreover, the number of answers in the "Jamais" column for the first and third items are really low, while they are quite high in the "Toujours" column, thus indicating grammar and lexis are the preferred ways to integrate pronunciation (see 6.4.6. for more details on lexis). On the other hand, it looks as if teachers are less likely to combine pronunciation exercises and explanations with syntax (56.1%, n = 32) and with language functions (59.6%, n = 34). For these two items, the number of answers in the "Jamais" column are rather high when compared to the "Toujours" column and to the results of the other items.

### 6.4.6. Section 6: Vocabulary Lists and Oral Tests

Because vocabulary was anticipated to be a field whereby teachers could tackle pronunciation easily (e.g., stress placement in words), two questions, that is, Q30 (see Figure 14) and Q31 (see Figure 15) questioned the respondents about the relation between vocabulary lists and pronunciation. It must first be noted that these two questions may have seen the answers

impacted by both the social desirability and self-deception biases (see 6.3.3.): absence of any information on vocabulary pronunciation in lists may not be socially accepted or desired among the teaching community.

# Figure 14

#### Pronunciation Information in Vocabulary Lists

Q30) Quand notez-vous une indication sur la prononciation dans les listes de vocabulaire fournies en classe d'anglais langue étrangère ?





Overall, as suggested by the data, vocabulary lists contain information for words that are presumed intrinsically complex (70.2%), and/or difficult for francophone learners (68.4%); some segments and stress in general can indeed be difficult for francophone learners (Dupoux et al., 1997; Frost, 2011; Walter, 2001) Besides, that combination (i.e., "mots jugés intrinsèquement complexes ..." and "mots jugés complexes pour les francophones ...") was selected by 47.% of the surveyed (n = 27). Also, the total absence of any information whatsoever (15.8%) appears to be more common than the presence of an pronunciation guidelines for every single word (7%) or than the presence of a QR code leading to audio files (14%).

If teachers indicate they do write information on the pronunciation of a word, they can provide it on either the segments (e.g., pronunciation of a vowel sound) or on the suprasegmental features (e.g., the place of the lexical stress). Question 31 asked what type of information teachers indeed provide.

### Figure 15







First, there is an odd inexplicable mismatch between Q30 and Q31 for 9 respondents in Q30 say they do not provide any pronunciation information in vocabulary lists, but 8 participants clicked on the same option in Q31. Figure 15 shows other more interesting results: nearly all participants (86%) state they add the pronunciation of segments to their vocabulary lists, while a bit more than half of the respondents (54.4%) say they also indicate the pronunciation of the lexical stress. The individual answers reveal that 50.9% of the teachers (n = 29) selected both items (i.e., segments and stress), 1.8% (n = 1) selected stress only, and that 12.3% (n = 7) selected the last option (i.e., "pas d'indications") and another 1.8% (n = 1) all three.

The final questions of Section 6, that is, Q32 (see Figure 16) and Q33 (see Figure 17) revolved around the place pronunciation holds in oral tests.

# Figure 16



■ 1 à 10%	■ 11 à 20%
= 21 à 30%	<b>31 à 40%</b>
■ 41 à 50%	■ 51 à 60%
■ 61 à 70%	■ 71 à 80%
<b>=</b> 81 à 90%	



Two options were most selected: "11 à 20%" (31.6%) and "21 à 30%" (31.6%). A bit more than one fifth of the instructors (21.1%) also selected "1 à 10%." The remaining items

were barely picked, or not picked at all (e.g., "51 à 60%" and "71 à 80%"). One interesting statistic is that 2 respondents (3.5%) dedicate close to half the points ("41 à 50%") in an oral test to pronunciation, while one other dedicates 61 to 70% and one last other goes as high as "81 à 90%."

No matter the share pronunciation has in oral tests, it may form an exclusion criterion (see Figure 17),<sup>57</sup> so looking solely at that percentages cannot give a comprehensive overview of the place of pronunciation in oral tests.

# Figure 17



For 5.3% of the participants, pronunciation is an exclusion criterion but it is not for 28.1% of the instructors. For a large majority of the panel (66.7%), pronunciation is an exclusion criterion in one case only: if pronunciation impedes intelligibility.

# 6.4.7. Section 7: Self-Assessment

Before analysing the ratings teachers give themselves regarding their pronunciation (Q34), mastery of theoretical concepts (Q35), and ability to teach pronunciation (Q36), it is crucial to say that the social desirability bias and self-deception (see 6.3.3.) are probably most present in these three questions: one can righteously expect that a teacher who rates their pronunciation 1 out 10 would be frowned upon (so both biases may come into play). On the other hand, it does not seem logic for an English teacher to pronounce things unintelligibly.

The data for the three questions are presented and combined in one graph (see Figure 18) with global remarks and means, but the results are not analysed further. This will however be done in a later section (see 6.5.).

<sup>&</sup>lt;sup>57</sup> An *exclusion criterion* is defined as a criterion that results in the failure of the task if not it is not met.

#### Figure 18

Self-Assessment of Own Pronunciation, Mastery of Theoretical Concepts, Ability to Teach Pronunciation Qs34 to 36) Sur une échelle de 1 à 10, j'évalue ma prononciation anglaise/ma maitrise de la théorie/ma capacité à l'enseigner à ...



The results for Q34 (pronunciation ratings) are not that surprising, as none of the instructors rate their pronunciation below 6 on the scale. Thus, the mean score for Q34 is quite high: 7.72. Still, the results for Q35 (mastery of theoretical concepts ratings) are more sparse. For instance, there is not a 10 point score, and 7 teachers grant themselves 5 or less than 5 points. The mean remains well above the second half of the scale though: 7.17. Lastly, the data for Q36 (ability to teach ratings) point to an overall self-satisfaction with regard to the ability to teach pronunciation. One respondent even gives themselves a 10. Once again, the mean remains similar: 7.49.

With regard to other surveys, highly self-rated pronunciation skills appear to be normal (Henderson et al., 2012, p. 12), but high self-rated methodology and theoretical skills do not (Baker, 2011b; Couper, 2017, 2021; Foote et al., 2012; Henderson et al., 2012).

## 6.5. Discussion and conclusion

The aim of this last subsection is to go back to both the research questions and to the hypotheses (see 6.2.). The research questions are answered in the order they were presented in earlier (see 6.2.). Before doing so, it is relevant to summarise the approach the instructors surveyed prefer when speaking of teaching pronunciation and see whether it is rather explicit (i.e., explaining rules), implicit (i.e., focusing on input), or both (i.e., having input and clear explanations).

#### 6.5.1. Explicit or Implicit Approaches

From the teachers' own choices in Q17 (see 6.4.5.), it appears the approach they favour is more explicit, but they remain implicit nonetheless. This is partially verified in Q16 (see 6.4.5.), since the teachers state they explain pronunciation features (i.e., an explicit activity), in spite of a stated focus on the practical side (see Q9 in 6.4.4.). Then, making use of metalanguages, such as keywords (see Q14 in 6.4.4.), and introducing IPA (see Q15 in 6.4.4.) are explicit activities.

On the other hand, the most often employed exercises are aloud readings and oral performances (see Q16 in 6.4.5.), which can be argued to be rather implicit: the pupils are exposed to oral input (i.e., their classmates reading and speaking) and one may expect that the teacher does not provide thorough pronunciation theory when a pupil is reading a piece of text. In essence, oral performances and aloud readings are implicit.

Still, the way these two activities are exploited renders them explicit: during them, the teachers provide a wide variety of feedback on various, but not all, mistakes (see Q21, Q22 and Q25 in 6.4.5.). Feedback, or guiding and questioning the pupil, is inherently explicit. Furthermore, when teachers present a new pronunciation feature, they claim they call on explicit techniques, such as contrasts, pictures, or exaggerated mouth movements (see Q18 in 6.4.5.). Even if, there is not any data on the amount of input the pupils are exposed to (e.g., the teacher speaking, listening comprehensions), teachers have to exercise listening skills in class and thus use a certain amount of oral input, so they must also work on pronunciation implicitly.

Therefore, it can be posited that teachers use a mixed approach in pronunciation teaching. On the one hand, they explicitly tackle pronunciation since they guide the pupils, show and explain how to pronounce some sounds thanks to a specific metalanguage. On the other, they use implicit activities where the pupils are passively exposed to input. On explicit or implicit approaches, it looks as if researchers promote both. The Articulatory Approach is based on the need of explicit knowledge of the relation between body and sounds (Billières et al., 2013a); phonological awareness (Venkatagiri & Levis, 2007, p. 273) along with PCA (Hu et al., 2013, p. 371) have proved to positively influence pronunciation. Explicit instruction was shown to lead to improvements, when compared to an implicit method, like repetitions (Gordon & Darcy, 2016, p. 81) At the same time, exposure to input (among others, Simon et al. 2014, p. 18, 2016, p. 740) and internalising voices (among others, LaScotte & Tarone, 2022, pp. 753–754; LaScotte & Tarone, 2019, pp. 104–105, 109; Moreno, 2016, p. 39) have been demonstrated to be beneficial.

This mixed approach can be explained thanks to the many meanings expressed in Q3 (see 6.4.3.). First, many respondents appear to believe theoretical knowledge on pronunciation is necessary for the teacher to teach pronunciation (Item 11), but a small majority does not believe theory is essential for the pupils (Item 21). Then, teachers also appear to believe that exposure to correct and sufficient English input is a must to learn pronunciation (Item 20). Moreover, the respondents' linguistics training is deemed satisfactory (hence an explicit method), and their methodology training as well, but less so (hence an implicit method) (Items 13, 14, 17, 18). Other justifications for the mixed approach include the reliance on (un)dedicated resources which may offer pronunciation knowledge (see Q19 in 6.4.5.) and the highly-self-rated teachers' theoretical and practical aptitude (see Qs35 to 36 in 6.4.7.). All this sheds light on why teachers deal with pronunciation explicitly and implicitly.

## 6.5.2. Research Questions

As far as the research questions are concerned, it was hypothesised about  $Q_{13}$  ("What goals do teachers pursue in pronunciation teaching?") that pronunciation goals revolved around intelligibility or having a BBC Pronunciation accent (H<sub>13</sub>). This hypothesis was partly proved (see Q2 in 6.4.2.): a majority of respondents (63.2%) claim they target intelligible pronunciation, but only 1.8% (n = 1) want their pupils to achieve native-like pronunciation. A last in-between goal aimed for by 35.1% of the participants is foreign accent deletion.

Then, the answer to  $Q_{14}$  ("What techniques do teachers use to teach pronunciation?") was assumed to be repetitions and oral presentations (H<sub>14</sub>). Again, this hypothesis was partly proved (see Q16 in 6.4.5.). The most frequent pronunciation activity is without a doubt aloud readings (employed by 31.6% of the participants), and oral performances are the second most frequent exercises (17.5%). Repetitions (e.g., choral repetitions, repetitions after feedback) are among the least frequent activities.

Concerning  $Q_{15}$  ("Do teachers use textbooks to teach pronunciation?"), teachers were presumed to rely on (dedicated) textbooks to teach pronunciation (H<sub>15</sub>). This hypothesis was not entirely verified (see Q19 in 6.4.5.). Combined, general textbooks and pronunciation dedicated textbooks are said to be used by 42.1% of the practitioners surveyed, but 33.3% do not call upon any textbook or any other resource. Besides, internet sites and posters with phonetic symbols are quite usual in classrooms (in the class of 31.6% and 29.8% of the respondents respectively). Because of the mixed results, it cannot be ascertained that H<sub>15</sub> is true. Turning to  $Q_{16}$  ("How do teachers use feedback regarding pronunciation?"), teachers were hypothesised to mainly use recasts in one context, that is, when a segment impairs intelligibility (H<sub>16</sub>). Several questions verified the hypothesis, which turned out to be not entirely correct (see Qs21 to 26 in 6.4.5.). On the type of feedback, the data show that recasts are most commonly provided (by 96.5% of the teachers) but they are not the only types of feedback teachers give, as clarification requests (49.1%), elicitation (47.4%), metalinguistic feedback (47.4%), and repetitions (64.9%) are fairly common as well.

Then, segments are the subject of feedback: 71.9% of the teachers concentrate on vowel sounds, 61.4% on consonant sounds, and 77.2% on diphthongs. Still intonation and lexical stress are paid attention to by many respondents (59.6% and 66.7% respectively).

Additionally, two types of exercises were distinguished to further detail the use of feedback: spontaneous speech acts and aloud readings. Overall, it does not appear that teachers focus on intelligibility mistakes in any of the given contexts, seeing the low percentages (especially during aloud readings [10.5%], less so in oral speech [31.6%]). What the participants rather focus on are the main pronunciation mistakes: 56.1% do so during speech acts and 61.4% during aloud readings. Therefore, teachers appear to use recasts (but not only) when a segment mistake (but not only) is considered to be a main pronunciation mistake.

As far as  $Q_{17}$  is concerned ("Do teachers use any kind of metalanguage to teach pronunciation?"), it was supposed that teachers may occasionally use IPA, (H<sub>17</sub>). This hypothesis was confirmed and even clarified (see Q14 and Q15 in 6.4.4. and Q16 and Q19 in 6.4.5). In total, 28.1% of the panel claims they frequently use IPA, which is less than the 50.9% who claims they use short English words to teach English pronunciation. This is also confirmed by 52.6% of the respondents who state they use IPA with their pupils, against 47.4% who state they do not use it. In either case, it appears pupils are at least introduced to IPA (by 63.2% of the teachers), despite a clear lack of time (according to 75.4%), despite the existence of online dictionaries with audio files (77.2% disagree that they render the teaching of IPA useless), and despite 52.6% assuming that their pupils have difficulties with the alphabet. It is then quite clear and apparent that teachers exploit IPA to some extent. The alphabet and its symbols are even physically present in the classroom of 29.8% of the surveyed in the form of posters. Yet, phonetic transcriptions (e.g., from symbols to written English and vice versa) do not appear to be frequent activities.

In other words, according to the questionnaire, teachers may use IPA, but they do not seem to use it in exercises, despite teacher training revolving around phonetic notation (see  $H_{22}$  below). It can then only be assumed what the purpose of IPA is: maybe it is a way to give precise feedback (see Q15 in 6.4.4.) and pronunciation information in vocabulary lists (see Q30 and Q31 in 6.4.6.).

The place non-native accents have in Belgian EFL classrooms was subject to  $Q_{18}$  ("Do teachers call upon non-native accents in class?") and it was supposed that instructors solely use NSs (H<sub>18</sub>). This hypothesis was entirely denied (see Q27 in 6.4.5.). Speakers with a native English accent seem to remain the norm, as 50.9% of the participants indicate they always use speakers with native accents in listening comprehensions and 29.8% report they never call upon non-native accents. Thus, this means that NNSs and non-native accents do have a place in the classroom: 1.8% of the practitioners claim they often use non-native accents and 39% claim they sometimes use them.

In regard to  $Q_{19}$  ("What norms do teachers refer to when teaching/evaluating pronunciation?"), it was hypothesised that instructors refer to BBC Pronunciation (H<sub>19</sub>); it only seemed logical since it was also assumed that teacher used the British norm as a goal. This hypothesis was partly proved (see Q1 in 6.4.2.): 47.7% do indeed say that they refer to BBC Pronunciation, but the exact same number of respondents say they do not refer to any norm when evaluating their pupils' pronunciation. Going deeper into details as to what these teachers may then refer to, one can assume they rely on intelligibility to test pupils' pronunciation (see H<sub>13</sub> above).

The answer to  $Q_{20}$  ("What features do teacher teach/focus on regarding pronunciation?") has already been partially given, so the hypothesis that teachers rather focus on segments and on lexical stress has also been partially tested (H<sub>20</sub>). Overall, H<sub>20</sub> was confirmed (see Qs10 to 13 in 6.4.4., Q16 and Q26 in 6.4.5., Q31 in 6.4.6.).

The instructors surveyed surely teach segments as at least 87.7% of them deal with all segments, but suprasegmentals also appear to be addressed in class since 70.2% claim to teach intonation and 73.7% claim to teach lexical stress, but in any case, a greater emphasis is put on segments. This emphasis naturally matches the degree of importance teachers grant the pronunciation features: the three most important items are vowel sounds (to 47.7% of the participants), diphthongs (to 36.8%) and lexical stress (to 33.3%). Besides, consonant sounds and intonation are not that far behind in terms of given importance. Logically, what teachers

focus on are what pupils are said to have most problems with, that is, lexical stress (35.1% of teachers say it is the most difficult feature for pupils) and diphthongs (26.3% say it is a problematic feature); also, a majority of the respondents (52.6%) rely on frequent errors to select pronunciation features to see in class, and their other selection criteria relate to errors or difficulties as well (e.g., the teachers' intuition of what is complicated).

Furthermore, the four activities that seem quite common in classrooms (i.e., aloud readings, oral performances, discrimination exercises, pronunciation feature explanations) are all (or at least can be) segment-oriented (e.g., discriminating between two close vowel sounds). They surely are suprasegmental-oriented as well, but the respondents pay less attention to that level. One may righteously expect that teachers provide feedback during these exercises, and what the respondents give most feedback on appear to be segments and lexical stress (intonation as well but it is less common).

A last sign pointing to a greater emphasis put on segments is the subject of extra information they incorporate within their vocabulary lists, and these pieces of information more often pertain to segments (86% of the cases) than to word stress (54.4%), which is a suprasegmental. Eventually, teachers do indeed focus on segments and on lexical stress, but they appear to address intonation quite often as well.

Apart from knowing what teachers do, it is relevant to inquire why they may (not) teach pronunciation (much) ( $Q_{21}$ , "Why do teachers [not] teach pronunciation [much]?"). The hypothesised reasons for the lack of pronunciation teaching were: lack of training in pronunciation theory ( $H_{21.1}$ ) and in pronunciation methodology ( $H_{21.2}$ ), teachers are not confident when teaching pronunciation ( $H_{21.3}$ ), and they believe exposure to input is enough ( $H_{21.4}$ ). A correlation was also put forward: the lower teachers rate their pronunciation abilities, the less likely they are to teach pronunciation thoroughly ( $H_{21.5}$ ). One of these hypotheses may have been verified (see Qs3 to 8 in 6.4.3., Qs34 to 36 in 6.4.7.). Before diving into  $Q_{21}$ , it must be mentioned that this research question originates from the assumed absence of pronunciation teaching. Yet, this assumption and the questionnaire data do not match up.

On the one hand, more teachers appear to be satisfied with what they learned on pronunciation theory (66.7%) than with what they learned on pronunciation methodology (49.1%). This lower percentage can be explained by the limited range of activities done in methodology training (see H<sub>22</sub> below). Still, 54.4% feel language teaching methodology classes prepared them well enough to teach pronunciation, and 80.7% feel the same towards

language/linguistic classes ( $H_{21.1}$  and  $H_{21.2}$ ). Oddly enough, the mean score for the pronunciation teaching ability is higher (M = 7.49) than the mean score for the mastery of theoretical concepts (M = 7.17). These two scores, the latter one specifically, point to an overall confidence when teaching pronunciation, and so do the satisfaction and preparation percentages given above ( $H_{21.3}$ ).

In addition, despite being considered important by a large majority (91.2%), input and its assumed positive impact on pronunciation does not appear to be a reason why teachers may not teach pronunciation ( $H_{21.4}$ ). This can be seen with the respondents' opinions on natives: 93% (totally) disagree that only a native is able to teach pronunciation. This result points to the fact that correct native-like input is not the exclusive criterion to teach pronunciation, so exposure to input is, according to the teachers, not enough to teach pronunciation. Linked to that matter, a large share of the teachers (47.4% and 94.7% respectively) think theory and practice are necessary ingredients to pronunciation teaching.

Finally, it was advanced that a low pronunciation ability rating equalled to a less thorough pronunciation approach (H<sub>21.5</sub>), but this hypothesis is almost impossible to verify for none of the respondents give themselves low pronunciation ratings, so a comparison between teachers is simply not doable. Nonetheless, the respondents have a rather high pronunciation rating mean (M = 7.72) and the approach to pronunciation depicted through the survey results appears to be rather complete (and in line with research guidelines [see 7.1.]): segments and some suprasegmentals are taught and practised explicitly and implicitly, and given feedback on, the pupils are exposed to a range of native and non-native accents, exercises can be perception based, IPA and keywords appear to be used, and various pronunciation resources seem to be exploited.

Concerning  $Q_{22}$  ("What kind of training in [teaching] pronunciation did/do teachers get?"), the hypotheses were the following: teachers received little training in pronunciation methodology (H<sub>22.1</sub>) and some training in phonology and/or phonetics (H<sub>22.2</sub>). The hypotheses were not verified (see Qs4 to 7 in 6.4.3).

The data on pronunciation methodology, despite first looking promising may actually be deceiving, as that training can be rather meagre (H<sub>22.1</sub>). A majority of participants claim their traineeship supervisor (80.7%) and their teachers (59.6%) commented on their pronunciation. Fewer teachers state they have learned techniques and ways to teach the pronunciation of a foreign language (15.8%), or of English specifically (38.6%). Then, none of the respondents say they have ever taken part in pronunciation training courses during in-service training, in spite of desire to do so (for 68.4%). Still, the respondents appear to be pleased with their methodology classes (see H<sub>21.2</sub> above).

According to the participants' statements, their training in phonology and/or phonetics seems rather complete (H<sub>22,2</sub>), and seems to have been IPA-oriented: phonetic transcriptions, learning IPA and another phonetic notation were quite common during their teacher training, as all three items were selected by at least 70.2% of the instructors. Prosody exercises were done by at least half of the respondents (52.6%). The activities a minority of the practitioners surveyed did are analysing the physical properties of pronunciation (24.6%), and learning the theory behind pronunciation (47.4%).

Finally,  $Q_{23}$  dealt with the integration of pronunciation into the classroom environment ("How is pronunciation integrated into the classroom?"). With the analysis of feedback (see  $Q_{16}$  above),  $H_{23.1}$  ("Pronunciation-dedicated moments mainly arise whenever communication breaks down") was slightly touched upon, but  $H_{23.2}$  was not ("The implicit or explicit integration of pronunciation into the classroom depends on the teachers' self-rated pronunciation level, in terms of pronunciation, ability to teach it, and mastery of theoretical concepts"). Both hypotheses were confirmed (see Qs28 to 29 in 6.4.5., Q30 in 6.4.6., Qs34 to 36 in 6.4.7.).

The first way to check  $H_{23,1}$  is to look at the activities most commonly employed in classroom and see whether feedback-related activities area among these (e.g., repetition after recasts). It is not the case as the teachers state aloud readings and oral performances are the most common pronunciation activities. These two exercises were looked at deeper (see  $H_{16}$  above) and it appears feedback (or pronunciation-dedicated moments) does occur when communication breaks down (more so with speech acts), but not only, as feedback is provided when pronunciation mistakes are considered important.

More on  $H_{23.1}$ , the survey shows that teachers emphasise communication in tests and in the way they organise pronunciation moments in their class. Firstly, 77.2% of the teachers say they integrate pronunciation modules into a larger sequence, depending on the pupils' needs. This link caters to what the pupils require to fulfil the communicative tasks of the sequence and is then also in line with the pursued goal of intelligibility (see  $H_{13}$  above): the pupils must be intelligible to communicate. Secondly, 84.2% dedicate from 1 to 30% of the total grade in oral tests to pronunciation, and some (66.7%) make it an exclusion criterion. In other words, pronunciation is tightly tied to communication and it possibly breaking down. As far as  $H_{23,2}$  is concerned, it is close to impossible to fully corroborate for none of the respondents give themselves low ratings, rendering any correlation between explicit teaching and scores unfeasible. In spite of that, the participants give themselves high pronunciation, theory, and teaching methodology scores, since the means are all above 7 (on a 10-point scale) and they tackle pronunciation more explicitly than implicitly (see 6.5.1.). Thus, it could be hypothesised that the higher the teachers' confidence is, the more explicit their pronunciation approach is.

### 7. Conclusion

This final section concludes this dissertation by first making a synthesis of the main information retrieved from both the literature review and the survey, and by then moving on to suggestions for future research.

### 7.1. Synthesis

The aim of this thesis was to answer the following question: Are francophone learners of English as a Foreign Language provided with the necessary tools to ensure viable pronunciation? This simple looking yet actually complex question had to be split into 23 other questions, which were themselves split into two groups, that is, the ones that would be answered theoretically thanks to a review of the available literature, and the ones that would be answered practically thanks to a survey. The following section goes back to these 23 questions in a theme-related order and it thus compares what researchers recommend doing and what practitioners say they do.

First and foremost, the central concept to the main research question must be defined: good pronunciation (Q<sub>7</sub>). This phrase is far too general and is replaced in the literature by three concepts. First, there is accentedness, defined as a difference with a speaking norm (Flege, 1984). Second, there is comprehensibility, defined as the difficulty to understand one's speech (Derwing et al., 1998). Finally, there is intelligibility, defined as actual understanding (Derwing et al., 1998), so it is considered to be the most important characteristic when assessing one's speech (Derwing et al., 2013).

Pronunciation can further be detailed in two levels (Couper, 2021): segmentals and suprasegmentals, the former referring to consonant and vowel sounds (Kissling, 2013), the latter to stress, rhythm and intonation (Field, 2005) ( $Q_8$ ). Both levels are argued to be critical to intelligibility (Bent et al., 2007) ( $Q_9$ ), suprasegmentals especially (Hahn, 2004), and that is why some researchers recommend focusing more on suprasegmentals (Levis & Grant, 2003), while others advise spending teaching time on both levels (Chun, 2002),

The 57 teachers surveyed appear to do the opposite of what Levis & Grant (2003) suggest because they focus more on segments than on suprasegments, but address both levels nonetheless ( $Q_{20}$ ). This choice is apparent in the way they use feedback, in the importance they award pronunciation features, and in how they exploit vocabulary lists. Besides, vocabulary lists and feedback are usual methods to incorporate a pronunciation component into the class,

which is actually most often integrated when the pupils need a specific feature  $(Q_{23})$  (e.g., pronunciation of a complex word in a list).

This emphasis on communication needs is just what researchers advocate. More precisely, there are two main goals or paradigms teachers may choose to focus on: intelligibility and nativeness (Levis, 2005), the former pertaining to being understood and the latter to sounding native. Researchers (among others, Bowen, 1972; Couper, 2021; Derwing & Munro, 2014; Diana, 2010; Galante & Thomson, 2017) encourage teachers to emphasise the intelligibility paradigm, or at least the learners' needs, over the nativeness paradigm ( $Q_1$ ). Yet, these goals may change depending on the teaching context (e.g., EFL, EIL, ESL), whereby sounding native can become completely justifiable ( $Q_{10}$ ). In addition, within the intelligibility paradigm, and in EIL contexts (Jenkins, 2002; Setter & Jenkins, 2005), it makes sense to expose learners to proficient non-native models and explore why these accented speakers remain intelligible (Jenkins, 2000; Low, 2021; Murphy, 2014) ( $Q_6$ ).

The 57 EFL teachers in francophone Belgium who answered the survey appear to follow all of these guidelines. They most aim for intelligible pronunciation or for accent deletion, so sounding native is not a frequent goal ( $Q_{13}$ ). Then, logically, quite a few instructors sometimes call upon non-native accents in class ( $Q_{18}$ ). Despite striving for intelligibility, many teachers use BBC Pronunciation to assess pronunciation, but many others say they do not refer to any kind of norm ( $Q_{19}$ ) and probably then evaluate pronunciation based on intelligibility.

Turning to pronunciation techniques, there does not appear to be a clear consensus on the ultimate best technique, but researchers agree on several that have proved their efficiency in improving intelligibility, comprehensibility, fluency, and/or accentedness (Q<sub>2</sub>). These are: phonetic notation (Fouz-González & Mompean, 2021), technology (Fouz-González, 2020; Thomson, 2012; Thomson, 2011), drama and imitation (Foote & McDonough, 2017; Galante, 2018; Galante & Thomson, 2017; LaScotte & Tarone, 2022; LaScotte & Tarone, 2019), perception exercises (Thomson, 2018), motor-based exercises (Messum & Young, 2021), the Silent Way (Richards & Rodgers, 2001a) and VTM (Renard, 2000a).

The teachers surveyed say they mostly use aloud readings and oral performances ( $Q_{14}$ ). At first glance, these are not the techniques researchers advocate, but other, less common activities, fall within research recommendations, and these are discrimination exercises (i.e., perception), imitations and phonetic transcriptions (i.e., phonetic notation). On that matter, IPA and short keywords (i.e., phonetic notation) are also claimed to be used in class (see below).

As far as phonetic notation is concerned ( $Q_5$ ), metalanguages offer possibilities in terms of autonomous learning (Marks, 2011; Mompean & Fouz-González, 2021; Mompean & Lintunen, 2015) and of feedback (Fouz-González & Mompean, 2021; Mompean & Fouz-González, 2021). Experiments also ascertained phonetic notation can improve perception (Fouz-González & Mompean, 2021). The data provided by the survey does not make it possible to know what teachers use a phonetic notation for, but they show that IPA and short keywords are used in class, in spite of several obstacles, such as complexity and time constraints ( $Q_{17}$ ). What is clear though, is that teachers do not appear to use phonetic notation much in order to perform transcription tasks (e.g., from symbols to words).

One last pronunciation teaching technique is feedback ( $Q_4$ ). Empirically tested, feedback (recasts in the study) proved to be a crucial component to pronunciation teaching (Saito & Lyster, 2012). The questioned teachers say they do make use of feedback, the moment, the type and the contents of which vary according to the activity. Overall, the teachers say they provide every type of feedback on the main pronunciation mistakes, and on segments, intonation or lexical stress ( $Q_{16}$ ).

Moreover, textbooks can constitute valuable assets to classrooms (Thornbury, 2013), especially when it comes to teaching pronunciation (Derwing et al., 2013), but researchers (Derwing et al., 2013; Levis, 2005; Thornbury, 2013) advise caution when dealing with them for they may reveal inconsistent with what research promotes (Q<sub>3</sub>): unclear explanations and unvaried exercises are rather common. However, the survey participants do not seem to rely too much on textbooks (Q<sub>15</sub>), contrary to what many other surveys claim (Baker, 2011a; Couper, 2017, 2021; Foote et al., 2012) (see below). Indeed, less than half the respondents say they use a textbook (whether dedicated to pronunciation or not), so what they use instead are internet sites, posters, or some do even not rely on any of these resources.

In general, teachers thus tackle pronunciation in class. It was first assumed that it would not be the case and several reasons were put forward as to why they would not teach pronunciation much. Because this assumption turned out to be false, several reasons why teachers do teach pronunciation (opposed to reasons why they do not teach it) were identified  $(Q_{21})$ : satisfactory complete linguistic training and satisfactory not-as-thorough methodology training  $(Q_{22})$ , enjoying pronunciation teaching, its efficiency, its importance, its relevancy and the supposed absence of a critical period. According to the surveyed, input is not so important that it turns any pronunciation teaching useless, which is what research also puts forward (i.e., input alone cannot guarantee efficient intelligible pronunciation acquisition) (Saito & Lyster, 2012) ( $Q_{12}$ ). Lastly, the practitioners' global positive relation to their training is visible in their high self-assessed scores.

Researchers surveyed the pronunciation approach of teachers before I did, and their conclusions can be summarised in one word: amateurism (Henderson et al., 2012). Teachers are not experts in pronunciation teaching but still teach it ( $Q_{11}$ ). After reviewing all the data offered by the survey, it can be argued that the 57 EFL teachers are not that amateur when it comes to pronunciation teaching.

In conclusion, there does not seem to be a clear consensus on the absolute best pronunciation acquisition method, but there appears to be one on ground rules (e.g., exercising perception, aiming for intelligibility). The same goes for the teachers themselves, who do not all have exactly similar attitudes and procedures with regard to pronunciation, but general tendencies which are mostly in line with these ground rules, can be unearthed through the data (e.g., reading texts aloud).

# 7.2. Prospects

Two other analyses were planned but had to be scrapped since they, combined with the rest of this dissertation, would go far beyond the scope of a master's thesis. Analysing textbooks and legal Belgian documents would indeed have been relevant to better understand the place pronunciation holds within EFL Belgian classes; it would also have made it possible to compare the teachers' stated practices with what it is available to them in textbooks and with what is legally expected from them. For instance, it would have been interesting from a research point of view to check whether aiming for intelligibility (i.e., the teachers' goal) stands in the legal texts, or to contrast the range of exercises teachers say they do and what level they focus, with what textbooks have to offer.

Suggestions for further research include: observing teachers' actual practices, asking pupils their views on pronunciation and comparing types of training. This last follow-up study came to mind when teachers had to indicate what they did during their methodology classes. In Liège, methodology classes are not specific to one language (Université de Liège, 2023), but in other universities (such as in Brussels), they are (Université Libre de Bruxelles, 2022), so the satisfaction with methodology classes may be greater if several are proposed, even maybe leading to a greater variety in pronunciation techniques used. This remains to be empirically verified. What is more, textbooks and legal texts are without a doubt and especially to teachers (Couper, 2021) important ways to incentivise teachers to focus more (or less) on pronunciation, and so do research and training. These four elements or pillars (i.e., research, training, legal texts, textbooks) all form a whole which may, if wanted, profoundly change the way teachers view pronunciation teaching. Research should indeed be complete and simple enough to understand for instructors (Couper, 2017; Derwing & Munro, 2005), training should follow researchers' recommendations (Chun, 2002; Gut et al., 2007), legal texts should consequently keep up to date with literature, and textbooks should eventually bridge the gap between the two training, research and teachers (Gut et al., 2007).

Everything thus appears to stem from research, which offers many, sometimes opposite, approaches. For instance, the Silent Way, although recommended by some, does away with input, while VTM insists on the teacher constantly speaking English. With all these suggestions, teachers may be at a loss as to what to do, and their practise is sometimes the contrary of what researchers promote (e.g., focus on segments). Other factors explaining why teachers may not really do as research recommends is difficulty. For instance, it may be difficult to have access to a language lab (Henderson et al., 2015) to enjoy the advantages of technology, or it is simply difficult to work on suprasegmentals since they are much more elusive than segments (Foote et al., 2016; Roberge, 2000b). Yet, out of the data, it is hardly ever the case that teachers do not do as research recommends: it mainly concerns the focus on segments and the range of activities. Eventually, sticking to a few guidelines (see 3.1.) may then reveal the most adequate and teacher-friendly approach; these could be focusing on both segmental and suprasegmental levels (e.g., constantly marking the placement of stress in vocabulary lists), working on features which impede the pupils' attempts at communicating (e.g., through lists of frequent errors) and working on the pupils' perception (e.g., through various listening comprehensions).

In conclusion, the answer to the main research question, that is, "Are francophone learners of English as a Foreign Language provided with the necessary tools to ensure viable pronunciation?" appears to be "Yes, they are." The results provided by the survey sometimes differ, but more often match what research advocates. It can thus be supposed that francophone learners of English as a Foreign Language are provided with the necessary efficient tools to ensure intelligible pronunciation.

#### References

- Academy of Motion Picture Arts and Sciences. (n.d.). *The 80th Academy Awards Memorable Moments*. Oscars.Org. Retrieved 30 March 2023, from https://www.oscars.org/oscars/ceremonies/2008/memorable-moments
- Badin, P., Tarabalka, Y., Elisei, F., & Bailly, G. (2010). Can you 'read' tongue movements?
  Evaluation of the contribution of tongue display to speech understanding. *Speech Communication*, 52(6), 493–503. https://doi.org/10.1016/j.specom.2010.03.002
- Baker, A. (2014). Exploring Teachers' Knowledge of Second Language Pronunciation Techniques: Teacher Cognitions, Observed Classroom Practices, and Student Perceptions. *TESOL Quarterly*, 48(1), 136–163. https://doi.org/10.1002/tesq.99
- Baker, A. A. (2011a). Discourse Prosody and Teachers' Stated Beliefs and Practices. TESOL Journal, 2(3), 263–292. https://doi.org/10.5054/tj.2011.259955
- Baker, A. A. (2011b). ESL teachers and pronunciation pedagogy: Exploring the development of teachers' cognitions and classroom practices. In J. Levis & K. LeVelle (Eds.), *Proceedings of the 2nd Pronunciation in Second Language Learning and Teaching Conference* (pp. 82–94). Iowa State University. https://ro.uow.edu.au/edupapers/368/
- Beebe, L. M., & Giles, H. (1984). Speech-accommodation theories: A discussion in terms of second-language acquisition. *International Journal of the Sociology of Language*, 1984(46), 5–32. https://doi.org/10.1515/ijsl.1984.46.5
- Bent, T., & Bradlow, A. R. (2003). The interlanguage speech intelligibility benefit. *The Journal* of the Acoustical Society of America, 114(3), 1600–1610. https://doi.org/10.1121/1.1603234
- Bent, T., Bradlow, A. R., & Smith, B. L. (2007). Chapter 19: Segmental errors in different word positions and their effects on intelligibility of non-native speech, All's well that begins

well. In O.-S. Bohn & M. J. Munro (Eds.), *Language experience in second language speech learning: In honor of James Emil Flege* (pp. 331–347). J. Benjamins Pub.

- Billières, M. (2000). Chapitre 1, Le corps en phonétique corrective. In R. Renard (Ed.), Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2, pp. 37–70). De Boeck Université.
- Billières, M. (2014a). Méthode verbo tonale: Diagnostic des erreurs sur l'axe clair / sombre. *Au son du fle*. https://www.verbotonale-phonetique.com/methode-verbo-tonale-diagnostic-erreurs-axe-clair-sombre/
- Billières, M. (2014b). Méthode verbo tonale: Principes généraux de correction. *Au son du fle*. https://www.verbotonale-phonetique.com/methode-verbo-tonale-principes-correction/
- Billières, M., Alazard, C., Astesano, C., & Nocaudie, O. (2013a, October 8). *Méthode articulatoire et méthode verbo-tonale*. Phonétique corrective en fle. https://mvtuoh.univ-tlse2.fr/seq04P0101.html
- Billières, M., Alazard, C., Astesano, C., & Nocaudie, O. (2013b, October 8). PHONETIQUE CORRECTIVE EN FLE, METHODE VERBO\_TONALE. Phonétique corrective en fle. https://mvt-uoh.univ-tlse2.fr/Intro\_seq1\_2\_3\_4.html
- Borrell, A. (2000). Chapitre 1, Les problèmes phonétiques liés à la distribution, Pour une correction verbo-tonale. In R. Renard (Ed.), *Apprentissage d'une langue* étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2, pp. 203–215). De Boeck Université.
- Borrell, A., & Salsignac, J. (2000). Chapitre 4, Importance de la prosodie en didicatique des langues (application au FLE). In R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale* (1re éd, Vol. 2, pp. 163–182). De Boeck Université.

- Bowen, J. D. (1972). Contextualizing Pronunciation Practice in the ESOL Classroom. *TESOL Quarterly*, 6(1), 83–94. https://doi.org/10.2307/3585862
- Bowler, B., & Cunningham, S. (2015). New Headway Pronunciation: Upper-Intermediate, Student's Practice Book (17th ed.). Oxford University Press.
- Brown, A. (1988). Functional Load and the Teaching of Pronunciation. *TESOL Quarterly*, 22(4), 593–606. https://doi.org/10.2307/3587258
- Busà, M. G. (2007). New Perspectives in Teaching Pronunciation. In A. Baldry, M. Pavesi, C.
  T. Torsello, & C. Taylor (Eds.), *From didactas to ecolingua: An ongoing research project on translation and corpus linguistics* (pp. 165–182). EUT.
- Carroll, J. B. (1962). The Prediction of Success in Intensive Foreign Language Training. In R. Glaser (Ed.), *Training Research and Education* (pp. 87–136). University of Pittsburgh Press. https://eric.ed.gov/?id=ED038051
- Cauldwell, R. (2011). Pronunciation in spontaneous speech. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, pp. xx–xxi). Cambridge university press.
- Chen, A. (2009). Perception of Paralinguistic Intonational Meaning in a Second Language. Language Learning, 59(2), 367–409. https://doi.org/10.1111/j.1467-9922.2009.00510.x
- Chun, D. M. (2002). Discourse intonation in L2: From theory and research to practice. Benjamins.
- Couper, G. (2003). The value of an explicit pronunciation syllabus in ESOL teaching. *Prospect*, *18*(3), 53–70.
- Couper, G. (2011). Investigations into pronunciation teaching. Speak Out!, 44, 9–13.
- Couper, G. (2014). Teaching concepts of pronunciation: Syllables, stress and drunk snails. *Speak Out!*, *50*, 34–38.

- Couper, G. (2017). Teacher Cognition of Pronunciation Teaching: Teachers' Concerns and Issues. *TESOL Quarterly*, *51*(4), 820–843. https://doi.org/10.1002/tesq.354
- Couper, G. (2021). Pronunciation Teaching Issues: Answering Teachers' Questions. *RELC Journal*, 52(1), 128–143. https://doi.org/10.1177/0033688220964041

Crystal, D. (2008). A dictionary of linguistics and phonetics (6th ed). Blackwell Pub.

- De Vriendt, S. (2000a). Chapitre 1, Le travail phonétique selon la méthode verbo-tonale dans l'enseignement secondaire. In R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale* (1re éd, pp. 245–256). De Boeck Université.
- De Vriendt, S. (2000b). Chapitre 2, Correction phonétique en langues germaniques pour apprenants francophones. In R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale* (1re éd, Vol. 2, pp. 257–270). De Boeck Université.
- Degrave, P., Rasier, L., & Hiligsmann, P. (2011). De invloed van een muzikale opleiding op de perceptie van de prosodie van een vreemde taal: Een verkennend onderzoek. In L. Rasier, V. Van Heuven, B. Defrancq, & Ph. Hiligsmann (Eds.), *Nederlands in het perspectief van uitspraakverwerving en contrastieve taalkunde* (pp. 83–100). Academia Press. https://dial.uclouvain.be/pr/boreal/object/boreal:86804
- Derwing, T. M. (2010). Utopian Goals for Pronunciation Teaching. In J. Levis & K. LeVelle (Eds.), *Proceedings of the 1st Pronunciation in Second Language Learning and Teaching Conference* (pp. 24–37). Iowa State University. https://www.researchgate.net/publication/267951652\_Utopian\_Goals\_for\_Pronunciati on\_Teaching

- Derwing, T. M., Diepenbroek, L. G., & Foote, J. A. (2013). How Well do General-Skills ESL Textbooks Address Pronunciation? *TESL Canada Journal*, 30(1), 22–44. https://doi.org/10.18806/tesl.v30i1.1124
- Derwing, T. M., & Munro, M. J. (2005). Second Language Accent and Pronunciation Teaching:
   A Research-Based Approach. *TESOL Quarterly*, 39(3), 379–397. https://doi.org/10.2307/3588486
- Derwing, T. M., & Munro, M. J. (2009). Putting accent in its place: Rethinking obstacles to communication. *Language Teaching*, 42(4), 476–490. https://doi.org/10.1017/S026144480800551X
- Derwing, T. M., Munro, M. J., & Carbonaro, M. (2000). Does Popular Speech Recognition Software Work with ESL Speech? *TESOL Quarterly*, *34*(3), 592–603. https://doi.org/10.2307/3587748
- Derwing, T. M., Munro, M. J., & Wiebe, G. (1998). Evidence in Favor of a Broad Framework for Pronunciation Instruction. *Language Learning*, 48(3), 393–410. https://doi.org/10.1111/0023-8333.00047
- Derwing, T. M., & Rossiter, M. J. (2002). ESL learners' perceptions of their pronunciation needs and strategies. *System*, *30*(2), 155–166. https://doi.org/10.1016/S0346-251X(02)00012-X
- Derwing, T. M., Rossiter, M. J., & Munro, M. J. (2002). Teaching Native Speakers to Listen to Foreign-accented Speech. *Journal of Multilingual and Multicultural Development*, 23(4), 245–259. https://doi.org/10.1080/01434630208666468
- Derwing, T., & Munro, M. J. (2014). Myth 1: Once you have been speaking a second language for years it's too late to change your pronunciation. In L. Grant (Ed.), *Pronunciation Myths: Applying Second Language Research to Classroom Teaching* (pp. 34–55). University of Michigan Press. https://doi.org/10.3998/mpub.4584330

- Detey, S., Fontan, L., & Pelligrini, T. (2016). Traitement de la prononciation en langue étrangère: Approches didactiques, méthodes automatiques et enjeux pour l'apprentissage. *Traitement Automatiques des Langues*, *57*(3), 15–39.
- Diana, A. (2010). La phonétique dans l'enseignement de l'anglais aux spécialistes d'autres disciplines: Enjeux et priorités. *Recherche et Pratiques Pédagogiques En Langues de Spécialité Cahiers de l APLIUT, Vol. XXIX N° 3,* 10–21. https://doi.org/10.4000/apliut.577
- Diepenbroek, L. G., & Derwing, T. M. (2014). To What Extent Do Popular ESL Textbooks Incorporate Oral Fluency and Pragmatic Development. *TESL Canada Journal*, 30(7), 1–20. https://doi.org/10.18806/tesl.v30i7.1149
- Dörnyei, Z., & Taguchi, T. (2010). *Questionnaires in second language research: Construction, administration, and processing* (Second Edition). Routledge, Taylor & Francis Group.
- Dupoux, E., Pallier, C., Sebastian, N., & Mehler, J. (1997). A Destressing "Deafness" in French? Journal of Memory and Language, 36(3), 406–421. https://doi.org/10.1006/jmla.1996.2500
- Dupoux, E., & Peperkamp, S. (2002). Fossil markers of language development: Phonological 'deafnesses' in adult speech processing. In J. Durand & B. Laks (Eds.), *Phonetics, Phonology, and Cognition* (pp. 168–190). Oxford University Press. http://www.lscp.net/persons/peperkamp/Dupoux\_Peperkamp\_(2002)\_Fossil\_markers\_ of\_language\_development.pdf
- Edwards, J. G. H., & Zampini, M. L. (2008). Introduction. In J. G. H. Edwards & M. L. Zampini (Eds.), *Phonology and second language acquisition* (pp. 1–11). J. Benjamins Publishing.

- Estebas-Vilaplana, E. (2017). The teaching and learning of L2 English intonation in a distance education environment: TL\_ToBI vs. the traditional models. *Linguistica*, *57*(1), 73–91. https://doi.org/10.4312/linguistica.57.1.73-91
- Ethnologue. (2022, February 21). What are the top 200 most spoken languages? Ethnologue. https://www.ethnologue.com/guides/ethnologue200
- General Data Protection Regulation, 02016R0679-20160504 (2016). https://eurlex.europa.eu/legal-content/EN/TXT/?uri=CELEX:02016R0679-20160504
- Fédération de l'Enseignement Secondaire Catholique. (2018). Langues modernes I, II, III, 2e et 3e degrés. Humanités générales et technologiques. Fédération de l'Enseignement Secondaire Catholique.
- Fédération Wallonie-Bruxelles. (2020, 2021). *Population scolaire par réseau*. Chiffres Clés. https://statistiques.cfwb.be/enseignement/fondamental-et-secondaire/populationscolaire-par-reseau/
- Field, J. (2005). Intelligibility and the Listener: The Role of Lexical Stress. *TESOL Quarterly*, *39*(3), 399–423. https://doi.org/10.2307/3588487
- Flege, J. E. (1984). The detection of French accent by American listeners. *The Journal of the Acoustical Society of America*, 76(3), 692–707. https://doi.org/10.1121/1.391256
- Foote, J. A., Holtby, A. K., & Derwing, T. M. (2012). Survey of the Teaching of Pronunciation in Adult ESL Programs in Canada, 2010. TESL Canada Journal, 29(1), 1–22. https://doi.org/10.18806/tesl.v29i1.1086
- Foote, J. A., & McDonough, K. (2017). Using shadowing with mobile technology to improve L2 pronunciation. *Journal of Second Language Pronunciation*, 3(1), 34–56. https://doi.org/10.1075/jslp.3.1.02foo

- Foote, J. A., Trofimovich, P., Collins, L., & Urzúa, F. S. (2016). Pronunciation teaching practices in communicative second language classes. *The Language Learning Journal*, 44(2), 181–196. https://doi.org/10.1080/09571736.2013.784345
- Fouz-González, J. (2020). Using apps for pronunciation training: An empirical evaluation of the English File Pronunciation App. *Language Learning & Technology*, 24(1), 62–85. https://doi.org/10125/44709
- Fouz-González, J., & Mompean, J. A. (2021). EXPLORING THE POTENTIAL OF PHONETIC SYMBOLS AND KEYWORDS AS LABELS FOR PERCEPTUAL TRAINING. Studies in Second Language Acquisition, 43(2), 297–328. https://doi.org/10.1017/S0272263120000455
- Fraser, H. (2010). Teaching suprasegmentals like the stars. Speak Out!, 43, 8–12.
- Frost, D. (2011). Stress and cues to relative prominence in English and French: A perceptual study. Journal of the International Phonetic Association, 41(1), 67–84. https://doi.org/10.1017/S0025100310000253
- Galante, A. (2018). Drama for L2 Speaking and Language Anxiety: Evidence from Brazilian EFL Learners. *RELC Journal*, 49(3), 273–289. https://doi.org/10.1177/0033688217746205
- Galante, A., & Piccardo, E. (2022). Teaching pronunciation: Toward intelligibility and comprehensibility. *ELT Journal*, *76*(3), 375–386. https://doi.org/10.1093/elt/ccab060
- Galante, A., & Thomson, R. I. (2017). The Effectiveness of Drama as an Instructional Approach for the Development of Second Language Oral Fluency, Comprehensibility, and Accentedness. *TESOL Quarterly*, 51(1), 115–142. https://doi.org/10.1002/tesq.290
- Gass, S., & Varonis, E. M. (1984). THE EFFECT OF FAMILIARITY ON THE COMPREHENSIBILITY OF NONNATIVE SPEECH. *Language Learning*, *34*(1), 65– 87. https://doi.org/10.1111/j.1467-1770.1984.tb00996.x

Gatbonton, E., Trofimovich, P., & Magid, M. (2005). Learners' Ethnic Group Affiliation and L2 Pronunciation Accuracy: A Sociolinguistic Investigation. *TESOL Quarterly*, 39(3), 489–511. https://doi.org/10.2307/3588491

- Gordon, J., & Darcy, I. (2016). The development of comprehensible speech in L2 learners: A classroom study on the effects of short-term pronunciation instruction. *Journal of Second Language Pronunciation*, 2(1), 56–92. https://doi.org/10.1075/jslp.2.1.03gor
- Gottfried, T. L. (2007). Chapter 13: Music and language learning, Effect of musical training on learning L2 speech. In O.-S. Bohn & M. J. Munro (Eds.), *Language experience in second language speech learning: In honor of James Emil Flege* (pp. 221–237). J. Benjamins Pub.
- Gussenhoven, C. (2002). Intonation and interpretation: Phonetics and Phonology. 47–57. https://www.isca-speech.org/archive\_open/sp2002/sp02\_047.pdf
- Gut, U. (2007). Learner corpora in second language prosody research and teaching. In J.
   Trouvain & U. Gut (Eds.), *Non-native prosody: Phonetic description and teaching practice* (pp. 145–167). Mouton de Gruyter.
- Gut, U. (2009). Non-native speech: A corpus-based analysis of phonological and phonetic properties of L2 English and German (T. Kohnen & J. Mukherjee, Eds.). Peter Lang.
- Gut, U., Trouvain, J., & Barry, W. J. (2007). Bridging research on phonetic descriptions with knowledge from teaching practice—The case of prosody in non-native speech. In J. Trouvain & U. Gut (Eds.), *Non-native prosody: Phonetic description and teaching practice* (pp. 3–21). Mouton de Gruyter.
- Hahn, L. D. (2004). Primary Stress and Intelligibility: Research to Motivate the Teaching of Suprasegmentals. *TESOL Quarterly*, 38(2), 201–223. https://doi.org/10.2307/3588378
- Hancock, M. (2007). English Pronunciation in Use: Intermediate (7th ed.). Cambridge University Press.

Henderson, A. (2015). Smoothie or Fruit Salad? Learners' Descriptions of Accents as Windows to Concept Formation. *Research in Language*, 13(1), 1–20. https://doi.org/10.1515/rela-2015-0009

- Henderson, A., Curnick, L., Frost, D., Kautzsch, A., Kirkova-Naskova, A., Levey, D., Tergujeff, E., & Waniek-Klimczak, E. (2015). The English Pronunciation Teaching in Europe Survey: Factors inside and outside the Classroom. In J. A. Mompean & J. Fouz-González (Eds.), *Investigating English Pronunciation* (pp. 260–291). Palgrave Macmillan UK. https://doi.org/10.1057/9781137509437\_12
- Henderson, A., Frost, D., Tergujeff, E., Kautzsch, A., Murphy, D., Kirkova-Naskova, A., Waniek-Klimczak, E., Levey, D., Cunnigham, U., & Curnick, L. (2012). The English Pronunciation Teaching in Europe Survey: Selected Results. *Research in Language*, *10*(1), 5–27. https://doi.org/10.2478/v10015-011-0047-4
- Howells, L., Lefebvre, C., Messum, P., & Young, R. (n.d.). *The Silent Way—Introducing the approach to teaching foreign languages*. Www.Silentway.Online. Retrieved 28 February 2023, from https://silentway.online/
- Hu, X., Ackermann, H., Martin, J. A., Erb, M., Winkler, S., & Reiterer, S. M. (2013). Language aptitude for pronunciation in advanced second language (L2) Learners: Behavioural predictors and neural substrates. *Brain and Language*, 127(3), 366–376. https://doi.org/10.1016/j.bandl.2012.11.006
- Intravaia, P. (2000). Chapitre 2, Pour une étiologie approfondie de l'erreur phonétique, Du crible phonologique au crible dialectique. In R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale* (1re éd, Vol. 2, pp. 217–242). De Boeck Université.

- Ioup, G. (2008). Chapter 2: Exploring the role of age in the acquisition of a second language phonology. In J. G. H. Edwards & M. L. Zampini (Eds.), *Phonology and second language acquisition* (pp. 41–62). J. Benjamins Publishing.
- Iverson, P., & Evans, B. G. (2007). Learning English vowels with different first-language vowel systems: Perception of formant targets, formant movement, and duration. *The Journal* of the Acoustical Society of America, 122(5), 2842–2854. https://doi.org/10.1121/1.2783198
- Iverson, P., Pinet, M., & Evans, B. G. (2012). Auditory training for experienced and inexperienced second-language learners: Native French speakers learning English vowels. *Applied Psycholinguistics*, 33(1), 145–160. https://doi.org/10.1017/S0142716411000300
- Jenkins, J. (2000). Pedagogic priorities 2: Negotiating intelligibility in the ELT classroom. In *The phonology of English as an international language: New models, new norms, new goals* (Nachdr., pp. 63–94). Oxford Univ. Press.
- Jenkins, J. (2002). A Sociolinguistically Based, Empirically Researched Pronunciation Syllabus for English as an International Language. *Applied Linguistics*, 23(1), 83–103. https://doi.org/10.1093/applin/23.1.83
- Jenkins, J. (2004). 5. RESEARCH IN TEACHING PRONUNCIATION AND INTONATION. *Annual Review of Applied Linguistics*, 24, 109–125. https://doi.org/10.1017/S0267190504000054
- Jenkins, J. (2011). Pronunciation for English as a Lingua Franca. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, pp. xxii–xxiii). Cambridge university press.

- Jilka, M. (2007). Different manifestations and perceptions of foreign accent in intonation. In J. Trouvain & U. Gut (Eds.), *Non-native prosody: Phonetic description and teaching practice* (pp. 77–96). Mouton de Gruyter.
- Jones, D. (2011a). *Cambridge English pronouncing dictionary* (P. Roach, J. Setter, & J. H. Esling, Eds.; 18th ed). Cambridge university press.
- Jones, D. (2011b). Introduction. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, pp. vi–xix). Cambridge university press.
- Jones, D. (2011c). Kilometre, kilometer. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, p. 277). Cambridge university press.
- Jones, D. (2011d). Liquorice. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, p. 291). Cambridge university press.
- Jones, D. (2011e). Research. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, p. 418). Cambridge university press.
- Kang, O. (2010). ESL Learners' Attitudes toward Pronunciation Instruction and Varieties of English. In J. Levis & K. LeVelle (Eds.), Proceedings of the 1st Pronunciation in Second Language Learning and Teaching Conference (pp. 105–118). Iowa State University.
- King, R. D. (1967). Functional Load and Sound Change. *Language*, 43(4), 831–852. https://doi.org/10.2307/411969
- Kissling, E. M. (2013). Teaching pronunciation: Is explicit phonetics instruction beneficial for FL learners? *The Modern Language Journal*, 97(3), 720–744. https://doi.org/10.1111/j.1540-4781.2013.12029.x
- Kosko, B. (2017, January 1). What scientific term or concept ought to be more widely known? Edge.Org. https://www.edge.org/response-detail/27130

Kramsch, C. (2002). Standard, norm and variability in language learning: A view from foreign language research. In S. M. Gass, K. Bardovi-Harlig, S. S. Magnan, & J. Walz (Eds.), *Pedagogical norms for second and foreign language learning and teaching: Studies in honour of Albert Valdman* (pp. 59–79). Benjamins.

- Krnic-Wambach, B., & Wambach, M. (2000). Chapitre 3, L'expression corporelle et musicale et le travail phonétique. In R. Renard (Ed.), *Apprentissage d'une langue* étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2, pp. 89–104). De Boeck Université.
- Krosnick, J. A., & Presser, S. (2009). *Question and Questionnaire Design* (J. D. Wright & P.
  V. Marsden, Eds.). Elsevier.
  https://web.stanford.edu/dept/communication/faculty/krosnick/docs/2009/2009\_handb
  ook\_krosnick.pdf
- Lane, J. (2023, February 9). *The 10 Most Spoken Languages In The World*. Babbel Magazine. https://www.babbel.com/en/magazine/the-10-most-spoken-languages-in-the-world
- LaScotte, D. K., & Tarone, E. (2022). Channeling Voices to Improve L2 English Intelligibility. *The Modern Language Journal*, *106*(4), 744–763. https://doi.org/10.1111/modl.12812
- LaScotte, D., Meyers, C., & Tarone, E. (2021). Voice and Mirroring in SLA: Top-Down Pedagogy for L2 Pronunciation Instruction. *RELC Journal*, 52(1), 144–154. https://doi.org/10.1177/0033688220953910
- LaScotte, D., & Tarone, E. (2019). Heteroglossia and Constructed Dialogue in SLA. *The Modern Language Journal*, *103*, 95–112. https://doi.org/10.1111/modl.12533
- Levis, J. (2007). COMPUTER TECHNOLOGY IN TEACHING AND RESEARCHING PRONUNCIATION. Annual Review of Applied Linguistics, 27, 184–202. https://doi.org/10.1017/S0267190508070098

- Levis, J. M. (1999). The intonation and meaning of normal yes/no questions. *World Englishes*, *18*(3), 373–380. https://doi.org/10.1111/1467-971X.00150
- Levis, J. M. (2005). Changing Contexts and Shifting Paradigms in Pronunciation Teaching. *TESOL Quarterly*, *39*(3), 369–377. https://doi.org/10.2307/3588485
- Levis, J. M., & Grant, L. (2003). Integrating Pronunciation Into ESL/EFL Classrooms. *TESOL Journal*, *12*(2), 13–19. https://doi.org/10.1002/j.1949-3533.2003.tb00125.x
- Levis, J. M., Sonsaat, S., Link, S., & Barriuso, T. A. (2016). Native and Nonnative Teachers of L2 Pronunciation: Effects on Learner Performance. *TESOL Quarterly*, 50(4), 894–931. https://doi.org/10.1002/tesq.272
- Levis, J., & Pickering, L. (2004). Teaching intonation in discourse using speech visualization technology. *System*, *32*(4), 505–524. https://doi.org/10.1016/j.system.2004.09.009
- Lewis, J. W. (2011). The Daniel Jones legacy. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, pp. xxiv-xxv). Cambridge university press.
- Lightbown, P. M. (2008). Easy as pie? Children learning languages. *Concordia Working Papers in Applied Linguistics*, *1*, 5–29.
- Lightbown, P., & Spada, N. M. (2013). *How languages are learned* (Fourth edition). Oxford University Press.
- Low, E. L. (2021). EIL Pronunciation Research and Practice: Issues, Challenges, and Future Directions. *RELC Journal*, 52(1), 22–34. https://doi.org/10.1177/0033688220987318
- Luo, D., Shimomura, N., Minematsu, N., Yamauchi, Y., & Hirose, K. (2008). Automatic pronunciation evaluation of language learners' utterances generated through shadowing. *Interspeech 2008*, 2807–2810. https://doi.org/10.21437/Interspeech.2008-476
- Macdonald, S. (2002). Pronunciation: Views and practices of reluctant teachers. *Prospect*, *17*(3), 3–18.

- Marks, J. (2011). Teaching and learning pronunciation. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, pp. xxvi–xxvii). Cambridge university press.
- Mehlhorn, G. (2007). Individual pronunciation coaching and prosody. In J. Trouvain & U. Gut (Eds.), *Non-native prosody: Phonetic description and teaching practice* (pp. 211–236). Mouton de Gruyter.
- Messum, P., & Young, R. (2021). Teaching Students to Pronounce English: A Motor Skill Approach in the Classroom. *RELC Journal*, 52(1), 169–178. https://doi.org/10.1177/0033688220964107
- Mompean, J. A., & Fouz-González, J. (2021). Phonetic Symbols in Contemporary Pronunciation Instruction. *RELC Journal*, 52(1), 155–168. https://doi.org/10.1177/0033688220943431
- Mompean, J. A., & Lintunen, P. (2015). Phonetic notation in foreign language teaching and learning: Potential advantages and learners' views. *Research in Language*, 13(3), 292–314. https://doi.org/10.1515/rela-2015-0026
- Moreno, L. (2016). *Channeling Charlie: Suprasegmental pronunciation in a second language learner's performance of others' voices*. University of Minnesota Digital Conservancy. https://conservancy.umn.edu/handle/11299/183052
- Munro, M. J., & Derwing, T. M. (1999). Foreign Accent, Comprehensibility, and Intelligibility in the Speech of Second Language Learners. *Language Learning*, 49(1), 285–310. https://doi.org/10.1111/0023-8333.49.s1.8
- Munro, M. J., & Derwing, T. M. (2001). MODELING PERCEPTIONS OF THE ACCENTEDNESS AND COMPREHENSIBILITY OF L2 SPEECH The Role of Speaking Rate. *Studies in Second Language Acquisition*, 23(4), 451–468. https://doi.org/10.1017/S0272263101004016

- Munro, M. J., & Derwing, T. M. (2006). The functional load principle in ESL pronunciation instruction: An exploratory study. System, 34(4), 520–531. https://doi.org/10.1016/j.system.2006.09.004
- Murphy, J. M. (2014). Intelligible, comprehensible, non-native models in ESL/EFL pronunciation teaching. *System*, 42, 258–269. https://doi.org/10.1016/j.system.2013.12.007
- Murphy, J. M., & Baker, A. A. (2015). History of ESL Pronunciation Teaching. In M. Reed & J. M. Levis (Eds.), *The Handbook of English Pronunciation* (1st ed., pp. 36–65). Wiley. https://doi.org/10.1002/9781118346952.ch3
- Oppenheim, A. N. (2001). *Questionnaire design, interviewing, and attitude measurement* (New Edition). Continuum.
- Pavelin, B. (2000). Chapitre 2, Statut et rôle du mouvement dans la communication orale en face àface. In R. Renard (Ed.), Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2). De Boeck Université.
- Pennington, M. C. (1988). Teaching pronunciation from the top down. *University of Hawai'i* Working Papers in ESL, 7(1), 203–227.
- Pennington, M. C. (2019). 'Top-Down' Pronunciation Teaching Revisited. *RELC Journal*, 50(3), 371–385. https://doi.org/10.1177/0033688219892096
- Pennington, M. C. (2021). Teaching Pronunciation: The State of the Art 2021. *RELC Journal*, 52(1), 3–21. https://doi.org/10.1177/00336882211002283
- Pronounce. (2012). In M. Waite (Ed.), *Paperback Oxford English dictionary* (7th ed). Oxford University Press.
- PronSci Pronunciation Science Ltd (Director). (2017, January 11). TPD W2 Stress #2. https://www.youtube.com/watch?v=EufYzScL2I0

- Pronunciation. (2012). In M. Waite (Ed.), *Paperback Oxford English dictionary* (7th ed, p. 576). Oxford University Press.
- Rasier, L. (2011). Uitspraakonderwijs Nederlands aan anderstaligen. In L. Rasier, V. Van Heuven, B. Defrancq, & Ph. Hiligsmann (Eds.), *Nederlands in het perspectief van uitspraakverwerving en contrastieve taalkunde* (pp. 119–141). Academia Press. https://dial.uclouvain.be/pr/boreal/object/boreal:86959
- Rasier, L., & Hiligsmann, P. (2009). Exploring the L1-L2 Relationship in the L2 Acquisition of Prosody. First and second languages: exploring the relationship in pedagogy-related contexts, Oxford, United Kingdom.
- Rasier, L., Hiligsmann, P., Caspers, J., & van Heuven, V. J. (2011). Accentual marking of information in Dutch and French as foreign languages: Production and perception. In M. Wrembel, M. Kul, & K. Dziubalska-Kołaczyk (Eds.), Achievements and perspectives in SLA of speech (pp. 227–238). Peter Lang.
- Renard, R. (Ed.). (2000a). Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2). De Boeck Université.
- Renard, R. (2000b). Avant-propos. In R. Renard (Ed.), Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2, pp. 5–7). De Boeck Université.
- Renard, R. (2000c). Chapitre 1, Une phonétique immergée. In R. Renard (Ed.), Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2, pp. 11–24). De Boeck Université.
- Renson, A. (2023). Le débat de société public régulé en anglais langue étrangère en Belgique Francophone. État des lieux sur les pratiques déclarées et expérimentation d'un enseignement explicite vs implicite du genre. [Unpublished doctoral thesis, Université de Liège]. https://orbi.uliege.be/handle/2268/300023
- Richards, J. C., & Rodgers, T. S. (2001a). *Approaches and Methods in Language Teaching* (2nd ed.). Cambridge University Press. https://doi.org/10.1017/CBO9780511667305
- Richards, J. C., & Rodgers, T. S. (2001b). The Silent Way. In Approaches and Methods in Language Teaching (2nd ed., pp. 81–89). Cambridge University Press. https://www.cambridge.org/core/books/abs/approaches-and-methods-in-languageteaching/silent-way/75639151C49CFB9FECC02814BDE9200B
- Rivenc, P. (2000). Chapitre 2, Place et rôle de la phonétique dans la méthodologie SGAV. In
  R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbotonale* (1re éd, Vol. 2, pp. 25–34). De Boeck Université.
- Roberge, C. (2000a). Chapitre 1, Rythme du discours, qui es-tu et où te caches-tu? In R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale* (1re éd, Vol. 2, pp. 107–130). De Boeck Université.
- Roberge, C. (2000b). Chapitre 2, Les intonations logique et affective, les grandes oubliées de l'éducation à la parole. In R. Renard (Ed.), *Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale* (1re éd, Vol. 2, pp. 131–154). De Boeck Université.
- Rogerson-Revell, P. M. (2021). Computer-Assisted Pronunciation Training (CAPT): Current Issues and Future Directions. *RELC Journal*, 52(1), 189–205. https://doi.org/10.1177/0033688220977406
- Rossiter, M. J., Derwing, T. M., Manimtim, L. G., & Thomson, R. I. (2010). Oral Fluency: The Neglected Component in the Communicative Language Classroom. *The Canadian Modern Language Review*, 66(4), 583–606. https://doi.org/10.3138/cmlr.66.4.583
- Saito, K., & Lyster, R. (2012). Effects of Form-Focused Instruction and Corrective Feedback on L2 Pronunciation Development of /1/ by Japanese Learners of English: FFI and L2

Pronunciation. *Language Learning*, 62(2), 595–633. https://doi.org/10.1111/j.1467-9922.2011.00639.x

- Sangster, C. (2011). The BBC, its Pronunciation Unit, and 'BBC English'. In P. Roach, J. Setter, & J. H. Esling (Eds.), *Cambridge English pronouncing dictionary* (18th ed, pp. xxviii–xxix). Cambridge university press.
- Sato, K., & Kleinsasser, R. C. (1999). Communicative Language Teaching (CLT): Practical Understandings. *The Modern Language Journal*, 83(4), 494–517. https://doi.org/10.1111/0026-7902.00037
- Schmidt, R. (2001). Attention. In P. Robinson (Ed.), *Cognition and Second Language Instruction* (1st ed., pp. 3–32). Cambridge University Press. https://doi.org/10.1017/CBO9781139524780.003
- Schmidt, R. W. (1990). The Role of Consciousness in Second Language Learning. *Applied Linguistics*, 11(2), 129–158. https://doi.org/10.1093/applin/11.2.129
- Setter, J., & Jenkins, J. (2005). State-of-the-Art Review Article. *Language Teaching*, 38(1), 1– 17. https://doi.org/10.1017/S026144480500251X
- silentway.online (Director). (2021, March 24). English the Silent Way: A film for language teachers. https://www.youtube.com/watch?v=TBaQ4Xg7qH4
- Simon, E., Lima, R., & De Cuypere, L. (2016). Acquiring non-native speech through early media exposure: Belgian children's productions of english vowels. *Poznan Studies in Contemporary Linguistics*, 52(4), 719–743. https://doi.org/10.1515/psicl-2016-0028
- Simon, E., Sjerps, M. J., & Fikkert, P. (2014). Phonological representations in children's native and non-native lexicon. *Bilingualism: Language and Cognition*, 17(1), 1–19. https://doi.org/10.1017/S1366728912000764
- Simons, G. (2021). Apreçu historique des méthodes d'enseignement en langues étrangères: De la méthode « grammaire-traduction » à la perspective « actionnelle ».

- Sluijter, A. M. C., & van Heuven, V. J. (1996). Acoustic Correlates of Linguistic Stress and Accent in Dutch and American English. 4. https://www.iscaspeech.org/archive\_v0/icslp\_1996/i96\_0630.html
- Spiardi, D. (n.d.). *Javier Bardem*. IMDb. Retrieved 21 February 2023, from http://www.imdb.com/name/nm0000849/bio
- Strange, W., & Shafer, V. L. (2008). Chapter 6: Speech perception in second language learners, The re-education of selective perception. In J. G. H. Edwards & M. L. Zampini (Eds.), *Phonology and second language acquisition* (pp. 153–191). J. Benjamins Publishing.
- Taylor, L. (2006). The changing landscape of English: Implications for language assessment. *ELT Journal*, 60(1), 51–60. https://doi.org/10.1093/elt/cci081
- Tergujeff, E. (2013). Learner Perspective on English Pronunciation Teaching in an EFL Context. *Research in Language*, *11*(1), 81–95. https://doi.org/10.2478/v10015-012-0010-z
- Thomson, R. (2012). English Accent Coach: Not quite a fairy godmother for pronunciation instruction, but a step in the right direction. *CONTACT Magazine*, *38*(1), 18–24.
- Thomson, R. I. (2011). Computer Assisted Pronunciation Training: Targeting Second Language Vowel Perception Improves Pronunciation. *CALICO Journal*, 28(3), 744– 765. https://doi.org/10.11139/cj.28.3.744-765
- Thomson, R. I. (2018). High Variability [Pronunciation] Training (HVPT): A proven technique about which every language teacher and learner ought to know. *Journal of Second Language Pronunciation*, 4(2), 208–231. https://doi.org/10.1075/jslp.17038.tho
- Thornbury, S. (2013). Resisting Coursebooks. In J. Gray (Ed.), Critical Perspectives on Language Teaching Materials (pp. 204–223). Palgrave Macmillan UK. https://doi.org/10.1057/9781137384263\_10

- Timmis, I. (2002). Native-speaker norms and International English: A classroom view. *ELT Journal*, 56(3), 240–249. https://doi.org/10.1093/elt/56.3.240
- Université catholique de Louvain. (2023). *Master [120] en langues et lettres modernes, orientation germaniques*. Uclouvain. https://uclouvain.be/prog-2023-germ2mprogramme
- Université de Liège. (2023). *Langues et lettres modernes, orientation germaniques, à finalité, programme détaillé.* Programmes ULiege. https://www.programmes.uliege.be/cocoon/20232024/programmes/P2UGER01\_C.ht ml
- Université Libre de Bruxelles. (2022). *Master en langues et lettres modernes, orientation Germaniques, Programme*. Ulb. https://www.ulb.be/fr/programme/2022-malgerm#programme
- Vaissière, J. (2002). Cross-linguistic prosodic transcription: French vs. English. In N. B. Volskaya, N. D. Svetozarova, & P. A. Skrelin (Eds.), *Problems and methods of experimental phonetics. In honour of the 70th anniversary of Pr. L. V. Bondarko* (St Petersburg State University Press, pp. 147–164). https://halshs.archivesouvertes.fr/halshs-00316156v2
- van Oostendorp, M. (2020). Phonology [Coursebook].
- VanPatten, B. (2002). Communicative classrooms, processing instruction, and pedagogical norms. In S. M. Gass, K. Bardovi-Harlig, S. S. Magnan, & J. Walz (Eds.), *Pedagogical* norms for second and foreign language learning and teaching: Studies in honour of Albert Valdman (pp. 105–118). Benjamins.
- Venkatagiri, H. S., & Levis, J. M. (2007). Phonological Awareness and Speech Comprehensibility: An Exploratory Study. *Language Awareness*, 16(4), 263–277. https://doi.org/10.2167/la417.0

- Wallonie-Bruxelles Enseignement. (2020). Programme d'études: Langues modernes.
   Enseignement secondaire ordinaire. Humanités générales et technologiques. 2e et 3e degrés. Service général de l'Enseignement.
- Walter, C. (2001). French speakers. In M. Swan & B. Smith, *Learner English: A teacher's guide to interference and other problems* (2nd ed., pp. 52–72). Cambridge University Press.

https://static1.squarespace.com/static/55a1bb03e4b072e2b123e170/t/5a122b078165f5 61f54e2bd5/1511140106233/swan-learnerenglish+145+to+161+Russian.pdf

- Wlomainck, P. (2000). Chapitre 3, Le travail du rythme et de l'intonation dans l'apprentissage d'une langue d'une langue étrangère. In R. Renard (Ed.), Apprentissage d'une langue étrangère/seconde, La phonétique verbo-tonale (1re éd, Vol. 2, pp. 155–161). De Boeck Université.
- Yenkimaleki, M., & van Heuven, V. J. (2021). Effects of attention to segmental vs. Suprasegmental features on the speech intelligibility and comprehensibility of the EFL learners targeting the perception or production-focused practice. *System*, 100, 1–14. https://doi.org/10.1016/j.system.2021.102557
- Zielinski, B. W. (2008). The listener: No longer the silent partner in reduced intelligibility. *System*, *36*(1), 69–84. https://doi.org/10.1016/j.system.2007.11.004

## Appendices

As mentioned in the introduction, the appendices were given to the supervisors and to the readers digitally on USB sticks, in both Word and PDF format.